NOTE TO USERS

Page(s) missing in number only; text follows. Page(s) were scanned as received.

103

This reproduction is the best copy available.



SENSORIMOTOR CONTROL OF LOCOMOTION BY INTACT AND INJURED SPINAL CORD

by

Sergiy Yakovenko

A thesis submitted to the Faculty of Graduate Studies and Research in partial fulfillment of the requirements for the degree of Doctor of Philosophy

Centre for Neuroscience University of Alberta Edmonton, Alberta, Canada Spring, 2004



Library and Archives Canada

Published Heritage Branch

Patrimoine de l'édition

395 Wellington Street Ottawa ON K1A 0N4 Canada 395, rue Wellington Ottawa ON K1A 0N4 Canada

Bibliothèque et

Direction du

Archives Canada

Your file Votre référence ISBN: 0-612-96340-3 Our file Notre référence ISBN: 0-612-96340-3

The author has granted a nonexclusive license allowing the Library and Archives Canada to reproduce, loan, distribute or sell copies of this thesis in microform, paper or electronic formats.

The author retains ownership of the copyright in this thesis. Neither the thesis nor substantial extracts from it may be printed or otherwise reproduced without the author's permission.

L'auteur a accordé une licence non exclusive permettant à la Bibliothèque et Archives Canada de reproduire, prêter, distribuer ou vendre des copies de cette thèse sous la forme de microfiche/film, de reproduction sur papier ou sur format électronique.

L'auteur conserve la propriété du droit d'auteur qui protège cette thèse. Ni la thèse ni des extraits substantiels de celle-ci ne doivent être imprimés ou aturement reproduits sans son autorisation.

In compliance with the Canadian Privacy Act some supporting forms may have been removed from this thesis.

While these forms may be included in the document page count, their removal does not represent any loss of content from the thesis. Conformément à la loi canadienne sur la protection de la vie privée, quelques formulaires secondaires ont été enlevés de cette thèse.

Bien que ces formulaires aient inclus dans la pagination, il n'y aura aucun contenu manquant.



This thesis is dedicated to my family -

To my father who always supported free-thinking and belief in my own abilities, To my mother for her affection and aspiration to share her profound wealth of heart, To my brother for saving my life.

Я хочу посветить этот тезис моей семье -

Моему отцу за поддержку свободного мышления и веру в мои возможности, Моей матери за любовь и стремление поделится всем своим душевным богатством,

Моему брату за спасение моей жизни.

Abstract

The control of locomotion is a complex problem of controlling the movement of a mechanical body with multiple degrees of freedom in a variable and often unpredictable environment. This thesis examines questions about the integration of the neuro-musculo-skeletal system for propulsion and stabilization. The five studies are included in this thesis cover the following topics: i.) muscle force generation during static and continuous muscle length changes; ii.) the role and contribution of stretch reflexes during locomotion; iii.) analysis of the spatiotemporal activity of lumbosacral motoneurons during normal locomotion; iv.) variability in the locomotor patterns during fictive locomotion and its implications for organization of the central pattern generator (CPG); v.) a test of the "propriospinal hypothesis" by intraspinal microstimulation (ISMS) of the thoracic spinal cord in rats with complete spinalization.

These studies show that the musculoskeletal system is capable of locomotor stabilization even in the absence of sensory feedback. Intrinsic muscle properties not only resist perturbations, but can also modulate gains in the sensory feedback pathways during locomotion. A finite control system based on multisensory rules can play a crucial role for stabilization of different modes of locomotion and resist to large perturbations. Study of the spatiotemporal activation of the lumbosacral motoneurons supports the hypothesis that activation of the lumbosacral MNs during walking is not wave-like, but it is a switch-like oscillation of activity. This finding led to a study of the patterns of MN activity during fictive locomotion, which is not influenced by motion-related sensory feedback controlling transitions between locomotor phases investigated in Chapter 3. The analysis of the fictive locomotor patterns revealed that the variation in the burst duration during spontaneous changes in cycle duration depends on the percent of the cycle occupied. These results can be explained by a conceptual model of the CPG with two half-centers receiving excitatory bias from descending inputs and proprioceptive feedback pathways. Finally, the results of the test of the "propriospinal hypothesis" with ISMS of thoracic segments in rats with complete spinal cord injury support the hypothesis.

Acknowledgements

I would like to express my deepest thanks to my supervisor and mentor, Dr. Arthur Prochazka, for the opportunity to work in his laboratory and for sharing his enthusiasm, insights and philosophy of the inner working of the cosmos.

During the course of my studies I benefited from collaborating with Dr. Vivian Mushahwar, Ms. Deborah M. Gillard, Dr. David McCrea, Dr. Michael Basso, Dr. David Bennett and Mr. Jan Kowalczewski on various research projects. I would like to thank Mr. Allen Denington and Mr. Michel Gauthier for their valuable advices and support throughout my graduate studies. I would also like to thank members of my examination committee, Dr. Karim Fouad, Dr. John Misiaszek, Dr. Richard Stein, Dr. Michael J. O'Donovan (external examiner), for the time and effort in providing me with feedback on my thesis. I would also like to extend my thanks to my fellow graduate students and the faculty and staff in Neuroscience for creating a supportive and enjoyable research environment.

On more personal note, I would like to thank Ms. Katrina Johnson, Mr. Gary and Drew Baldwin for their help, encouragement, friendship and for sharing their enthusiastic outlook on life. I am thankful to all my friends in Edmonton for their support and many happy memories.

Finally, I would like to thank the Canadian Institutes of Health and Research and Mr. Warren Hall for the financial support.

Table of Contents

Chapter 1	1
1.1 Introduction	2
1.2 Muscle	3
1.2.1 Basic morphology and properties of the contractile machinery	3
1.2.2 Hill-type muscle model	6
1.2.3 History-dependent muscle properties	8
1.3 Musculo-skeletal system – the plant	11
1.3.1 Passive dynamic properties of skeletal system	11
1.3.2 Stabilizing role of muscles	14
1.4 Sensory feedback	15
1.4.1 Proprioceptors involved in the regulation of locomotion	15
1.4.2 Models of spindles and tendon organs	17
1.4.3 Stretch reflexes	18
1.5 Control of locomotion	21
1.5.1 Simple locomotor reflexes	21
1.5.2 Locomotor central pattern generator (CPG)	23
1.5.3 Locomotor patterns generated in the presence of sensory feedback	26
1.6 Spinal cord trauma	29
1.7 Thesis objective and summary	32
1.8 References	34
	0 -
Chapter 2	53
2.1 Introduction	54
2.2 Methods	55
2.3 Results	62
2.4 Discussion	66
2.5 References	71
Chanter 3	74
3.1 Introduction	75
3.2 Methods	79
3.2.1. Structure of the locomotor model	70
3.2.2 Darametric testing	70 80
3.2.3 Stretch reflex model	82
3.2.4 Finite state rules	82
3.3 Results	85
3.3.1 Simulation of "deafferentation" experiment	85
3.3.2 Locomotor performance in the absence of sensory feedback	80
3.3.3 Effect of stretch reflexes on locomotor performance	80
3.3.4 Contribution of state dependent control to locomotor stability	07 00
3.3.5 Closed loop gain of stretch reflex during locomotion	90 02
2 / Dissussion	94
3.5 Deferences	90 100
J.J REFEICHCES	100

3.5 References

Chapter 4	103
4.1 Introduction	105
4.2 Methods	109
4.2.1 Digitization	109
4.2.2 Calculation of the center of MN activity	116
4.3 Results	117
4.3.1 Anatomical distribution of MN pools in the lumbosacral spinal cord	119
4.3.2 Analysis of the spatiotemporal pattern of MN activity	116
4.4 Discussion	126
4.5 References	134
hapter 5	139
5.1 Introduction	140
5.2 Methods	142
5.2.1 Surgical procedures	142
5.2.2 Data recording and analysis	143
5.2.3 Adaptive threshold crossing method	144
5.3 Results	146
5.4 Discussion	152
5.5 References	158
hapter 6	163
6.1 Introduction	164
6.2 Methods	165
6.3 Results	168
6.4 Discussion	170
6.5 References	173
hapter 7	176
7.1 Summary	177
7.2 Discussion and future directions	179
7.1.1 Intrinsic muscle properties	179
7.1.2 Contribution of stretch reflexes during locomotion	181
7.1.3 Patterns of spatiotemporal activation of the lumbosacral motoneurons	182
7.1.4 Adjustments of the locomotor phases with changes in cycle duration	185
7.1.5 The "propriospinal hypothesis"	186
7.2 References	189

List of Tables

Table 4.1. Rostral (L5-L6) and caudal (L7-S1) populations of extensor, BFp-FDL-gracilis-
SMp-ST and flexor MNs in the cat lumbar enlargement

112

List of Figures

Figure 1.1. Schematic of muscle model	4
Figure 1.2. Simplified diagram of the locomotor control system.	12
Figure 2.1. Experimental set-up.	57
Figure 2.2. Mean angle-torque data for constant voluntary activation of wrist flexors.	59
Figure 2.3. Mean angle-torque data for constant voluntary activation of wrist extensors.	61
Figure 2.4. Mean angle-torque data for constant electrical activation of wrist flexors 64	
Figure 2.5. Mean angle-torque data for constant electrical activation of wrist extensors	65
Figure 2.6. Effect of muscle force-velocity properties on the width of the loops	67
Figure 3.1. Structure of neuromuscular locomotor model	77
Figure 3.2. Muscle properties, activation profiles and control system schematic	79
Figure 3.3. Contribution of stretch reflexes during locomotion	84
Figure 3.4. Schematic explaining how Principal Component Analysis	86
Figure 3.5. Locomotor performance plots	88
Figure 3.6. PCA locomotor performance	91
Figure 3.7. Strength of stretch reflexes is modulated by muscle length	93
Figure 4.1. Anatomical models of the organization of the MN pools within the cat lumbar	
enlargement.	108
Figure 4.2. MN distribution along the rostrocaudal axis of the lumbar enlargement of the	
cat spinal cord.	111
Figure 4.3. EMG patterns of hindlimb muscles during forward level locomotion at	
moderate speeds ($\sim 0.6 \text{m/s}$)	118
Figure 4.4. Modelled spinal MN activity in the lumbosacral enlargement during locomotion	122
Figure 4.5. Variations in assumed activation patterns to assess parametric sensitivity	124
Figure 5.1. An adaptive optimal threshold method	145
Figure 5.2. The relationship between phase and cycle duration for the hindlimb nerve	
activity	147
Figure 5.3. The relationship between phase and cycle duration	148
Figure 5.4. The result of a one-way ANOVA of the slopes of the relationship between phase	
and cycle duration of hindlimb nerve activity	150
Figure 5.5. The result of the one-way ANOVA and multiple comparisons of the slopes of	
the phase modulation with Tukey's honestly significant difference criterion	151
Figure 5.6. Simple schematic of the locomotor pattern generator	154
Figure 6.1. Regeneration hypothesis and methods	166
Figure 6.2. Results	169

List of Abbreviations

adductor femoris magnus	AdFM
one-way analysis of variance	ANOVA
ankle flexors	AF
ankle extensors	AE
contractile element	CE
caudofemoralis	CF
central nervous system	CNS
central pattern generator	CPG
brain-derived neurotrophic factor	BDNF
biceps femoris anterior and posterior	BFa, BFp
extensor carpi radialis brevis	ECRB
extensor carpi radialis longus	ECRL
extensor carpi ulnaris	ECU
extensor digitorum longus	EDL
electromyographic / electromyography / electromyogram	EMG
functional electrical stimulation	FES
flexor carpi radialis	FCR
flexor carpi ulnaris	FCU
flexor digitorum longus	FDL
flexor hallucis longus	FHL
flexor digitorum longus and flexor hallucis longus	FDHL
gluteus maximus and medius	GlutMax,
GlutMed	
hip extensors	HE
hip flexors	HF
intraspinal microstimulation	ISMS
knee extensors	KE
bifunctional muscle with knee flexor and hip extensor function	KF-HE
L-3,4-dihydroxy-phenylalanine	L-DOPA
gastrocnemius lateralis and medialis	LG, MG
gastrocnemius and soleus	LGS
mesencephalic locomotor region	MLR
motoneuron	MN
maximum voluntary contraction	MVC
motor unit action potential	MUAP
biceps femoris posterior and semitendinosus together	PBST
principal component analysis	РСА
parallel elastic element	PE
peroneus longus	PerL
plantaris	Plan
peripheral nervous system	PNS
rectus femoris	RF
physiological range of motion	ROM
series elastic element	SE

SCI
SMa, SMp
SMAB
SRS
ST
SRTa, SRTm
TA
ТР
Tc
Te
Tf
VI, VL, VM

CHAPTER 1

General introduction

1

1.1 Introduction

At every moment of our existence the nervous system has to solve multiple control problems. Imagine the level of control involved in successful retrieval of an apple that seems just fingertips beyond your reach, while standing on a tall ladder leaning against an apple tree. It is not surprising then that control of movement involves multiple interconnected systems, organized in hierarchical order. While this system is a convenient adaptation, which reflects multiple steps in evolution, it presents a challenging subject to study.

There are two basic approaches to unraveling the details of the system. The reductionist approach is based on gathering facts about the smallest known components of the system of interest and combining this knowledge to arrive at a full understanding. Alternatively, the systems approach is based on studying the system as whole and deducing information about its components from global facts. Both of these approaches have their advantages and disadvantages; however, a symbiosis of the two is often the most successful. Study of the control of locomotion is a good example of such symbiosis, as will be demonstrated in this thesis.

This chapter contains a general overview of topics focused on the control of movement during locomotion.

1.2 Muscle

"Nature sets in motion by signs and watchwords, which are made with little momentum.... Just as in the army the soldiers are set in motion by one word as if by a given signal and continue to move until they receive another signal to stop, so the muscles move in order and harmony from established custom." – William Harvey (1578-1657)

1.2.1 Basic morphology and properties of the contractile machinery

The neural system delivers its action through biological actuators, muscles. There is a considerable diversity in muscle structure. Here we will overlook properties specific to cardiac and smooth muscles and concentrate on striated skeletal muscle. The word "striated" comes from light microscope observations of striped patterns of dark and light bands in muscle fibers formed by sarcomeres arranged in series. Muscle fibers are arranged in parallel and attached to a connective tissue sheet, called aponeurosis (Figure 1.1A). The force generated by muscle is transmitted through the aponeurosis to a tendon, which in turn is attached to the body skeleton. The points of muscle attachment, called origin and insertion, determine the line of muscle action. In some muscles, muscle fibers are arranged at an angle, called a pennation angle, to the line of muscle action. This organization increases the moment arm of muscle fibers and is common for muscles bearing high loads, such as gastrocnemius and vasti. However, this organization comes at the expense of the decreased range of motion. Muscles bearing lower loads, like the hip flexors iliopsoas and sartorius, have their fibers arranged in parallel with the line of action (He *et al.*, 1991).

Over 100 years ago, Blix (Blix, 1893) observed that the force generated by active muscles depends not only on the activation level but also on muscle length. He plotted length-force relationships of frog muscles from a series of isometric tetanic contractions performed by the muscle at different lengths.





Figure 1.1. Schematic of muscle model. A. Muscle attaches to skeleton through aponeurosis and tendon. Some muscle fibers are arranged at an angle to the line of muscle action. B. Hill-type muscle model, where CE is a contractile element, PE is an elastic element in parallel and SE is an elastic element in series.

Repeatedly reproduced (e.g. Abbot & Wilkie, 1953), the profile of the isometric length-force curve consisted of an ascending limb, followed by a plateau, a descending limb and a final ascending limb. With the exception of the final ascending limb, which is attributed to passive muscle stiffness (Huijing, 1985; Zajac, 1989), the initial ascending and descending limbs are attributed respectively to increases and decreases in the overlap of actin and myosin filaments within sarcomeres during active muscle contractions (Gordon *et al.*, 1966). The muscle length corresponding to the maximum value of active force is called the muscle's optimal length, at which the overlap between the muscle filaments is the greatest (Gordon *et al.*, 1966; Huxley, 1974).

In addition, Hill determined experimentally the dependence of active force generation on the velocity of shortening, by measuring the force exerted by a muscle at a given length, as the muscle actively shortened under a series of loads (Hill, 1938). Similarly, the dependence of active muscle force on the velocity of lengthening was characterized (Katz, 1939). During shortening (concentric) and lengthening (eccentric) contractions the muscle force-velocity relationship is similar to that of a viscous energy-dissipating element.

The force-length-velocity relationship of active muscle may be explained to a large degree by the sliding-filament theory (Huxley & Niedergerke, 1954; Huxley & Hanson, 1954; Huxley, 1957). It suggests that as sarcomeres change their length and the contractile filaments slide past each other, myosin cross-bridges can form connections with attachment sites on actin. The theory states that muscle force depends on the number of attached actin-myosin cross-bridges and the rate of cross-bridge attachment and detachment. If these parameters are independent, then the forcelength and force-velocity properties of a single sarcomere are independent of each other as well. Further studies demonstrated that the profile of the force-velocity curve varied only marginally with length (Bahler *et al.*, 1968; Edman & Andersson, 1968) and the profile of the force-length curve was independent of shortening velocity (Abbot & Wilkie, 1953). Thus, muscle can be conceptually represented by a "dashpot" in parallel with a force generator, which behaves like a "spring", at least on the initial ascending limb of the isometric force-length curve (Figure 1.1B). This description became the foundation for Hill-type muscle models.

1.2.2 Hill-type muscle model

Muscle modeling is a challenging field of systems physiology. The most broadly used muscle models are based on the classic paper by Nobel Prize laureate Archibald V. Hill in 1938 (Hill, 1938). The central principal of these models is that skeletal muscle behaves as a lumped sarcomere with the following independent properties: muscle activation, force-length and force-velocity relationships. Generally, this relationship can be described by the equation:

$$F_{ce} = F_{\max} \cdot f_v \cdot f_l \cdot a_t$$
 Eq. 1.1

where F_{co} is force exerted by the contractile element, F_{max} is the maximum isometric force, f_v and f_l are the normalized force-velocity and the force-length relations respectively and a_l is the normalized muscle activation.

In this phenomenological model, which deals with macroscopic properties, the contractile element (CE) is often connected in series and in parallel with two elastic elements. The series elastic element (SE) is usually stiff and represents properties of tendon, while the parallel elastic element (PE) represents passive properties of connective tissue (Figure 1.1B).

The main challenges of muscle modeling include optimization of model structure and realistic representation of built-in parameters. For example, until recently, most of the studies in biomechanics have used Hill-type models with force-length curves based on *isometric* (static) measurements to describe *continuous* active movements (e.g. Hoy et al., 1990; Pandy et al., 1990; Veltink et al., 1992). The isometric force-length curve described above predicts that a muscle stretched beyond its optimal length, therefore on the descending limb of the curve, generates less force with further stretch, i.e. muscle stiffness is negative (Stein et al., 1999). How realistic is this prediction? In control systems, negative stiffness in a force actuator would lead to unstable behavior. For example, if an actuator with negative stiffness were loaded, the stretch would decrease its force production and lead to further stretch, causing instability. From experiments in healthy human subjects, wrist extensors, but not flexors, exhibit isometric force-length curves with descending limbs and decreasing moment arms at muscle lengths beyond the optimal length (Lieber et al., 1994; Loren et al., 1996; Lieber & Friden, 1998). This would suggest that as the wrist is extended, the torque of muscles like extensor carpi radialis brevis (ECRB) would decrease. This prediction seemed at odds with observations of these muscles under functional electrical stimulation (FES) control (Prochazka et al., 1992). Chapter 2 of this thesis shows that in situations involving continuous movement, muscle stiffness remains positive throughout the physiological range of motion (ROM), even when the isometric force-length curve contains a pronounced descending limb. A similar result was found for rat medial gastrocnemius muscle (Meijer et al., 1997). Thus, isometric length-tension curves do *not* adequately represent muscle properties during continuous movements.

Once the level of model complexity is chosen, the next challenge is to find a realistic representation of model parameters, e.g. muscle or tendon length and stiffness, maximum velocity of concentric contraction, moment arms, etc. Some of these parameters can be estimated indirectly. For example, tendon stiffness can be determined from its cross-sectional area, the tendon slack length and Young's modulus (Hoy *et al.*, 1990). Physiological range of motion can be used to estimate muscle rest length and vice versa (Brown *et al.*, 1996). Another intriguing indirect method of parameter estimation is based on known or predicted optimization of natural movements, e.g. based on minimal jerk, which can be used to predict muscle response to a known perturbation (Bertram *et al.*, 1999; Raasch & Zajac, 1999; Neptune *et al.*, 2000). However, direct experimental measurement of muscle properties used in a model is likely to be the most reliable approach.

1.2.3 History-dependent muscle properties

History-dependent muscle properties, such as short-range stiffness (SRS), yielding (Joyce *et al.*, 1969), force enhancement (Edman *et al.*, 1978; Herzog & Leonard, 2002) and depression (Herzog *et al.*, 2000) are usually omitted in Hill-type muscle models. However, these properties can greatly influence muscle performance. Short-range stiffness is observed as an instantaneous increase in muscle stiffness when muscle is lengthened through its initial \sim 1% of rest length (Joyce *et al.*, 1969). The mechanism of SRS can be explained based on the cross-bridge theory. It is thought that at the onset of stretching a proportion of cross-bridges effectively remain attached and continue to deform elastically. When muscle is stretched beyond the initial \sim 1% of rest length, the number of attached cross-bridges reduces and muscle stiffness declines (Rack & Westbury, 1974). It has been proposed that SRS shares the same mechanism as yielding, a transient decrease in exerted force during ramp and hold experiments, and that together they play an important role in postural

stabilization (Malamud *et al.*, 1996). Finally, force enhancement is an increase in exerted isometric force immediately after stretching (Abbot & Aubert, 1951; Edman *et al.*, 1978, 1982). Since this property is most likely to manifest itself when the muscle is stretched beyond its optimal length (Edman *et al.*, 1978), this phenomenon together with SRS and yielding may explain the positive stiffness of muscle during continuous movement, even when the isometric force-length curve has a negative slope (Gillard *et al.*, 2000).

In light of new evidence, some of the assumptions of Hill-type muscle models have been criticized. The assumption of an independent relationship between muscle activation and the profile of the force-length curve has been criticized based on studies using the single frog muscle fiber preparation. It was shown that during submaximal contractions, lengthening increased force production beyond its isometric level (Edman & Reggiani, 1984; Stennett *et al.*, 1996). The mechanism of this enhanced activation at longer muscle lengths was linked to coupling between two processes, Ca^{2+} -troponin binding and actin-myosin binding (Fuchs, 1977; Rassier *et al.*, 1999). One way this effect can be taken into account is by defining a_t in Eq. 1.1, a parameter relating activation to force, as a function of length (Cheng *et al.*, 2000).

Another debatable assumption of Hill-type muscle models is that all sarcomeres in a muscle fiber change their length uniformly, which justifies describing muscle as a single lumped sarcomere. Edman and Reggiani were the first to demonstrate experimentally that the lengths of sarcomeres along a single fiber vary both when the muscle is active and at rest (Edman & Reggiani, 1984). Several observed muscle phenomena may be explained primarily by inhomogeneities in the length of sarcomeres in series, a state called sarcomere nonuniformity. For example, force enhancement, described above, may result from the redistribution of sarcomere lengths. After lengthening, some sarcomeres may be stretched beyond their optimal length and act as stiff transducers of load, while others would exert higher forces because they continue to operate on the ascending limb of the force-length curve (Edman & Reggiani, 1984). The phenomenon of force depression after active shortening was also attributed to sarcomere nonuniformities (Meijer *et al.*, 1997). Using variable amounts of shortening prior to isometric contractions, the authors demonstrated a population of force-length curves, which showed force depression dependent on the amount of preceding shortening at high lengths. It is remarkable that even in the presence of the descending limb in isometric force-length curves, the lines connecting points with the same previous shortening history had positive slopes. Recently, finite-element modeling, an analysis of stress distribution within materials, was successfully used to describe sarcomere length inhomogeneities in single fibers (reviewed in Huijing, 2000).

Muscle shows a great capacity for adaptation to changes in functional demands due to development, exercise or trauma (Belanger & McComas, 1989; Yang *et al.*, 1990; Gordon *et al.*, 1993; Navarrette & Vrbova, 1993). It is well known that lack of use leads to gradual deterioration of muscle and bone, e.g. after paralysis or prolonged exposure to zero gravity (Keller *et al.*, 1992; Gordon & Mao, 1994). Motor unit type can change due to electrical stimulation and exercise (Scott *et al.*, 1985; Gordon & Mao, 1994). It was shown that both exercise and change in muscle excursion lead to changes in the number of sarcomeres in series and a consequent adaptation of the force-length curve (Lynn & Morgan, 1994; Koh & Herzog, 1998; Lynn *et al.*, 1998). For example, downhill running exercise for several days increases the number of sarcomeres in myofilaments of vastus intermedius, a knee extensor, in rats (Lynn & Morgan, 1994; Lynn *et al.*, 1998). The same adaptation was reported after an increase in muscle range of motion following

surgical retinacular release of tibialis anterior, an ankle flexor (Koh & Herzog, 1998). Thus, it is not surprising that force-length curves of flexor and extensor muscles might be different (Gareis *et al.*, 1992; Gillard *et al.*, 2000). This suggests that force-length properties of a muscle are optimized to its functional demands.

To conclude, muscle modeling is challenging because of the complex morphology and nonlinearity of the musculoskeletal system. When assessing the results of muscle modeling, a balance should be achieved between oversimplification of muscle behavior on one hand, which may under-represent some important properties, and too much complexity on the other, which can lead to difficulties in implementation and hidden errors.

1.3 Musculo-skeletal system – the plant

"The power of locomotion is that which contracts and relaxes the muscles whereby the members and joints are moved, extended or flexed. This power reaches the limbs by way of the nerves and there are as many forms of power as there are of movement. Each muscle has its own peculiar purpose and it obeys the decree of the composite sense." – Avicenna (Ibn Sina) (980-1037)

1.3.1 Passive dynamic properties of skeletal system

The skeletal system is sometimes called in engineering terms the *plant*. It is controlled by the nervous system through non-linear force actuators, the muscles. The challenge of movement coordination is to manage redundant degrees of freedom of this system (Bernstein, 1967). It is conceivable that each individual degree of freedom, or independent direction of movement, *may* be individually controlled by the nervous system. However, there are many indications that structure and control of complex biological motor systems are organized hierarchically (Philippson, 1905; Ghez & Krakauer, 2000). The first system to consider is the skeletal system, which consists of mechanical segments connected to each other by joints. In the block diagram of Figure 1.2,



Figure 1.2. Simplified diagram of the locomotor control system. Loop 1 represents intrinsic muscle negative displacement feedback. Loop 2 contains delayed sensory feedback from tendon organ and spindle afferents. Loop 3 describes sensory pathways capable of modifying amplitude and / or timing of the locomotor CPG. Finally, loop 4 describes multisensory pathways, which may contribute to the control of phase transitions of the locomotor CPG.

the skeletal system is represented as *load*. It was recognized for a long time in biomechanics that purely mechanical systems are capable of "self-organizing" behaviour through interactions with the environment (Schoner & Kelso, 1988). For example, it was shown that simple, completely passive anthropomorphic models without actuators or sensory feedback can preserve stability while walking down shallow slopes (McGeer, 1990; Garcia *et al.*, 1998). Robots have been developed with passive dynamic properties and simple control algorithms that allowed them to hop, run or trot (Raibert, 1990). Thus, the skeletal system alone may contain intrinsic properties, which stabilize the system even in the absence of active control.

Do animals and humans use the passive dynamic properties of their skeletal system for motor control? Supporting evidence was adduced by Hoy and Zernicke (Hoy & Zernicke, 1985) using inverse dynamics methods to study active muscle moments and inertial moments during the swing phase of gait in cats. They demonstrated that the inertial moments due to angular accelerations of the limb segments contribute to knee flexion in early swing (F phase) and knee extension in late swing (E1 phase). Humans similarly show passive knee flexion during the swing phase (Mochon & McMahon, 1980). Another example of passive generation of locomotion is provided by the remarkable ability of relatively simple passive spring-mass models to describe behaviours like walking, running and jumping (Blickhan, 1989; Lee & Farley, 1998; Herr *et al.*, 2002; Seyfarth *et al.*, 2003). Taken together, these results suggest that the nervous system combines passive reactions of the skeletal system with active muscle forces in the control of movement. The challenge in neurophysiology is to disentangle the relative contributions of intrinsic musculoskeletal properties, active muscle forces and the neural activity that orchestrates the whole.

1.3.2 Stabilizing role of muscles

The inherent mechanical properties of muscles, and their arrangement across joints, act to resist postural perturbations and simplify control in the face of the multiple degrees of freedom of the skeletal system. Often muscles are attached to control several independent directions of joint excursion and to maximize automatic levering of the mechanical system when it is perturbed from its control position. For example, the arrangement of flexor and extensor muscles around the ankle joint is such that large yaw displacement (abduction-adduction), can be counteracted with large extrasaggital levering of these muscles to increase torques needed to stabilize quadrupeds and bipeds during locomotion (Lawrence *et al.*, 1993; Lawrence & Nichols, 1999).

Ankle flexors are prone to generate adduction torques (Lawrence & Nichols, 1999), which would tend to place the leg more medially during the swing phase of locomotion, a strategy linked to energy efficient walking (Donelan *et al.*, 2001). Finally, it was shown that during postural perturbations in cats, the muscles of the leg exert stabilizing *passive* lateral forces before active muscle contractions mediated by the nervous system occur (Macpherson, 1988).

The spring-like property of muscle actuators has long been recognized as equivalent to negative displacement feedback. This has led to the development of motor control theories that incorporate muscle stiffness as an integral component, e.g. the equilibrium point (λ) theory (Feldman, 1966). Spring-like force-length and viscous force-velocity behaviour serves to resist muscle lengthening or shortening and tends to return the system to its control position. In this sense mechanical properties play an important stabilizing role during coordinated behaviours such as locomotion (Gerritsen *et al.*, 1998; Prochazka & Yakovenko, 2002). Note that the negative feedback loop 1 in Figure 1.2, which corresponds to the intrinsic muscle properties, belongs to the feedforward and feedback pathways. Because of that, it can be predicted that muscle shortening, 14

which decreases the gain of loop 1, automatically attenuates the loop gains of all feedback pathways (Prochazka *et al.*, 1997a; Yakovenko *et al.*, 2004). Alternatively, muscle lengthening produces the opposite effect; it increases the gains of loops 1-3 in Figure 1.2. The intrinsic muscle properties not only contribute to automatic load compensation, but they may also play an important role of modulating loop gains of sensory feedback.

1.4 Sensory feedback

Spinal motoneurons are "the final common pathway." - Sir Charles Sherrington (1857-1952)

1.4.1 Proprioceptors involved in the regulation of locomotion

In the early days of neuroscience, it was recognized that sensation of movement, muscle sense, originates in muscles and is caused by mechanoreceptors located in the muscle belly, ligaments and joints. These receptors were unified in a single class called proprioceptors (Sherrington, 1906). Since then, experimental studies have shown that other receptors respond to movement and also contribute to kinaesthesia. Thus, they may be included in the proprioceptor group. For example, skin receptors, initially classified as exteroreceptors together with auditory, visual and olfactory systems, can generate powerful illusions of movement in response to stretch (Moberg, 1983; Collins & Prochazka, 1996). In order to direct the discussion, we will concentrate on an overview of the muscle spindle and tendon organ receptors only.

Muscle spindles are spindle-shaped encapsulated sensory organs located in the muscle belly. The human body contains 25,000 - 30,000 muscle spindles, most of which are located in the extremities (Hulliger, 1984). Each spindle contains two types of sensory endings embedded in specialized intrafusal muscle fibers, which in their turn attach to extrafusal muscle fibers. Though intrafusal fibers may sometimes share innervation by ß-skeletofusimotor axons with extrafusal fibers, spindle fibers receive input from specific small-diameter γ -motoneurons (Emonet-Denand *et al.*, 1975). The two types of sensory endings belong to primary group Ia fibers (conduction velocity 72–120 m/s in cats) and secondary group II fibers (20–72 m/s) (Matthews, 1972).

Spindle afferents respond to changes in muscle length even in the absence of fusimotor drive. Group Ia afferents are more sensitive to dynamic changes, i.e. they are more sensitive to changes in velocity and acceleration, while group II afferents respond to static length changes and have a low dynamic sensitivity. Similar to the surrounding muscle, intrafusal fibers possess non-linear properties resulting in non-linear aspects of spindle afferent response, e.g. Ia sensitivity depends on the amplitude, offset and velocity of stretch (Hulliger, 1984; Poppele & Quick, 1985; Prochazka, 1996). When fusimotor drive is present, the sensitivity of spindle afferents to variations in length changes. Generally, dynamic fusimotor drive increases the sensitivity (gain) of response of Ia sensory endings. Static fusimotor drive increases the firing bias of both group Ia and II endings of sensory afferents and reduces Ia stretch sensitivity (reviewed in Prochazka, 1996).

Overall, there are about 80% more tendon organs than spindles. Tendon organs are encapsulated structures, which can be found at the junction of muscle and tendinous structures and are about 1 mm long and 0.1 mm in diameter (Barker, 1974). Within the capsule, free sensory endings are intertwined with the collagen fibers of tendon in braided fashion, so that tendon stretches deform the nerve endings. Each tendon organ , which gives rise to a group Ib afferent axon outside of the capsule, interacts with 10-20 motor units and each motor unit affects 1-6 tendon organs (Proske, 1981; Jami, 1992). Unlike spindles, tendon organs are relatively sensitive to changes of length or velocity, but respond to force generated actively by the muscle (Houk & Henneman, 1967; Stephens *et al.*, 1975). Tendon organs may possess discontinuities in their force response characteristics and signal whole muscle force non-linearly (Jami et al. 1985). For example, because of the parallel arrangement of muscle fibers, contraction of fibers not "sampled" by a tendon organ can in some cases unload it, reducing the firing of the Ib afferent (Houk and Henneman 1967; Stuart et al. 1972).

1.4.2 Models of spindles and tendon organs

Mathematical models of spindles and tendon organs have been developed based on inputoutput characteristics collected in acute experiments. In the case of muscle spindles, the receptorbearing muscles were detached from their surrounding structures, their tendons were cut and attached to length servos and the muscles were subjected to sinusoidal or "ramp and hold" stretches, while the activity of the sensory afferents was recorded from teased dorsal root filaments (Matthews & Stein, 1969; Poppele & Quick, 1985). The response characteristics of Ia afferents were usually fitted with functions of muscle length and velocity. Tendon organs have been studied by stimulating single motor units, groups of motor units, or the whole muscle and recording the Ib responses in dorsal root filaments. Frequency analyses of their response to feedback-controlled forces applied to the muscle have also been performed (Houk & Simon, 1967; Anderson, 1974; Stephens et al., 1975). These models are in the form of transfer functions between the force signal and the firing frequency of Ib afferents. The tendon organ transfer function is that of a high-pass filter with several zeros and poles. The Bode plots of Ib afferents resemble those of muscle spindle group II afferents more than Ia afferents, in that their high-pass characteristics are less marked. At low levels of active force, the firing of single motor units and the recruitment of new motor units cause irregularity and steps in Ib firing rates (Jami et al., 1985).

Spindle and tendon organ models have been successfully tested in freely moving normal cats (Prochazka *et al.*, 1976; Loeb & Duysens, 1979; Prochazka & Gorassini, 1998b, a). Accompanied by measurements of muscle length, velocity and EMG activity, ensemble activity of spindle group Ia and II afferents and tendon organ group Ib afferents were chronically recorded in dorsal root ganglia. Spindle models relating muscle length to afferent firing produced passable predictions of Ia and II responses to fast muscle stretches that occur during locomotion. Predictions of the afferent firing rates were further improved by the addition of EMG-dependent offsets, which represented biasing of Ia firing rate due to alpha-gamma linkage (Emonet-Denand *et al.*, 1975). Though muscle force was not directly measured in these studies, tendon models produced good predictions of Ib firing rates from EMG profiles as well.

The development of representative mathematical models of sensory feedback and their use in combination with musculo-skeletal models may prove to be crucial for our understanding of the role of sensory feedback during complex behaviours such as locomotion, control of balance and voluntary movement of the upper extremities.

1.4.3 Stretch reflexes

Muscle stretch gives rise to reflex responses through sensory feedback to the spinal cord and brain from muscle receptors. Since Sherrington's time, it has been realised that so-called reflex responses to muscle stretch are not invariant, and may be modified by descending and segmental CNS mechanisms (Loeb, 2000).

The contribution of segmental Ia-mediated stretch reflexes and their electrically-elicited counterparts, H-reflexes, to ongoing activity has been extensively studied (Dietz, 1996; Prochazka, 1996; Brooke *et al.*, 1997; Dietz, 1998; Duysens *et al.*, 2000). It was found that human H-reflexes

are bigger during standing than during locomotion and their amplitude is phase-dependent within the locomotor step cycle (Garrett *et al.*, 1981; Garrett & Luckwill, 1983; Capaday & Stein, 1986). Hreflexes were also found to be further reduced during running and in difficult beam-walking (Capaday & Stein, 1987; Llewellyn *et al.*, 1990). It remains unclear whether the origin of this modulation is controlled centrally (Schneider *et al.*, 2000) or through sensory feedback pathways (Misiaszek *et al.*, 1998). Similarly, stretch reflexes elicited during locomotion by rapid stretching of muscles also show phase-dependent modulation (Orlovsky & Shik, 1965; Akazawa *et al.*, 1982; Hiebert *et al.*, 1996; Sinkjaer, 1997). However, stretch-evoked responses show a higher variability than H-reflexes, which is probably due to the fact that a variety of pathways contribute to stretch responses (Sinkjaer, 1997; Sinkjaer *et al.*, 1999; Christensen *et al.*, 2001).

Stretch reflexes have been subdivided into groups based on their latencies (Pearson *et al.*, 1999; Gritsenko *et al.*, 2001). Short-latency responses are produced by the firing of spindle and tendon organ afferents in response to muscle stretch. Spindle group Ia afferents monosynaptically excite homonymous alpha motoneurons and cause the receptor-bearing muscle to resist the imposed stretch. While Ib afferents reflexly inhibit homonymous motoneurons in static postures, this switches to homonymous excitation during locomotion, at least in cats (Conway *et al.*, 1987; Pearson & Collins, 1993; Gossard *et al.*, 1994; Guertin *et al.*, 1995; Pratt, 1995). The evidence for switching of Ib reflex sign in humans remains ambiguous (Dietz, 1998; Stephens & Yang, 1999; Capaday, 2000; Misiaszek *et al.*, 2000; Pang & Yang, 2000). Spindle group II afferents may also contribute to stretch reflexes through di- and oligo-synaptic pathways (Matthews, 1991; Sinkjaer *et al.*, 2000; Grey *et al.*, 2001).

Key concepts of the stretch reflex system can be presented in the form of the control system diagram shown in Figure 1.2 (loops $1 \notin 2$). The contribution of spindle group Ia and II afferents to stretch reflexes is represented as negative feedback (loop 2), which augments the intrinsic stiffness of active muscles (loop 1). The diagram shows that intrinsic stiffness can be viewed as negative displacement feedback (Partridge, 1966). As mentioned above, Ib feedback switches from inhibition to excitation of homonymous motoneurons during locomotion (Gossard et al., 1994; McCrea et al., 1995). Ib homonymous excitation represents positive force feedback. Positive feedback is generally associated with instability in engineering systems. However, a theoretical study (Prochazka et al., 1997a, b) showed that the force-length properties of muscles tend to stabilize what would otherwise be an unstable system with positive force feedback. In control systems, positive force feedback with embedded negative displacement loop (positive muscle stiffness) behaves similarly to a negative displacement loop. For example, load can increase muscle force through the positive feedback loop to homonymous motoneurons. The muscle can then exert higher forces at the same muscle length, which may lead to muscle shortening or negative displacement. To summarize, during locomotion, stretch reflexes (loop 2) augment the automatic load compensation provided by intrinsic muscle stiffness (loop 1).

How much do the stretch reflexes contribute to the ongoing activity of muscles during locomotion? Several research groups have attempted to answer this challenging question. Selective nerve block of fusimotor drive was shown to reduce EMG activity of ankle extensors by about 50% in a locomotor cat preparation, indicating that spindle input is a major contributor to motoneuronal excitation (Severin, 1970). However, some alpha motoneurons may have been blocked along with gamma motoneurons in these experiments. In another set of acute experiments

in cats, the contribution of stretch reflexes was estimated by comparing force responses to sinusoidal stretches before and after deafferentation (Bennett *et al.*, 1996). The difference in the force generated by ankle extensors was about 25%. In humans, it was estimated that stretch reflexes induced by abrupt lengthening of soleus, an ankle extensor, contributed 30-60% of the total EMG activity during stance (Yang *et al.*, 1991) and reached about 50% in the late swing phase (Sinkjaer *et al.*, 1996). However, it is difficult to separate the force attributable to intrinsic muscle stiffness alone and the additional force attributable to stretch reflex augmentation of that stiffness in the closed-loop system, where muscle is allowed to change length during load compensation. On one hand, the above estimates indicate that stretch reflexes contribute significantly to locomotor control. On the other hand, the intrinsic musculoskeletal capacity for stabilization should not be underestimated. Interestingly, it was demonstrated recently that large-fiber sensory loss and absent tendon jerks in mammals, including humans, did not abolish weight-support and even locomotion, provided lateral support was present (Allum *et al.*, 1998; Pearson *et al.*, 2003).

1.5 Control of locomotion

"If then the soul in pigeons, frogs, vipers, and tortoises, is not confined to the brain, but can continue for a long time to actuate their bodies independent of that organ \dots why should we deny that, in man and such animals that resemble him most, the parts may continue to be actuated by the soul or sentient principle for some few minutes after their communication with the brain has been cut off?" — Robert Whytt (1714-1766)

1.5.1 Simple locomotor reflexes

The study of locomotion dates back to La Mettrie's observations of rhythm generation in animals with cervical spinalization (Mettrie, 1745). A century later, Spencer (Spencer, 1855) and Sechenov (Sechenov, 1863) proposed that movements are composed of chains of reflexes. Further experimental evidence for this hypothesis was provided by studies of flexion and extension reflexes

21

in spinalized or decerebrated dogs and cats (Freusberg, 1874; Sherrington, 1910). At about the same time, Muybridge was the first to publish cinematic recordings of animal movement and introduced an objective kinematic method of analysing locomotion (Muybridge, 1887).

Later, Philippson (Philippson, 1905) proposed not only a comprehensive subdivision of the step cycle based on kinematics, but also he suggested the locomotor mechanism based on interactions with environment and reflex responses. He stated that extensor thrust was evoked by ground contact. This reaction was reinforced by the flexor crossed-extensor reflex of the contralateral limb, until exteroceptive stimuli caused by the increased pressure on the foot caused flexion of the ipsilateral limb and extension of the contralateral limb. The cycle then repeated for the contralateral limb. In spite of this appealing description, Sherrington and Brown (Sherrington, 1910; Brown, 1911) recognized that spinalized dogs were capable of "movements of progression" or "reflexive stepping" without ground contact and even after cutaneous deafferentation.

The remarkable study by Sherrington (Sherrington, 1910) remains a reference work of biomechanical actions of cat hindlimb muscles and their participation in flexion and extension reflexes and air-stepping even to this day. In this study, Sherrington suggested that the locomotor pattern is generated by the interplay of "primary" and "secondary" reflexes. While the "primary" reflex is continuous and can be evoked by electrical stimulation of the rostral stump of the transected cervical spinal cord or by mechanical stimulation of various peripheral points outside the limb, the "secondary" reflex is antagonistic to the "primary" reflex and is evoked by the "primary" reflex through a proprioceptive response. Thus, the "secondary" reflex interrupts the "primary" reflex until the proprioceptive signal is abolished and then the cycle repeats. Essentially, Sherrington reconciled the purely peripheral scheme of pattern generation with the contradictory
results of the deafferentation studies by introducing a central factor that could be affected by descending or peripheral inputs.

A study by Brown, Sherrington's student, provided fundamental evidence of a central origin of locomotor activity (Brown, 1911). He demonstrated that decerebrate and low-spinal animals can generate alternating flexion and extension activity in completely deafferented hindlimb muscles. Though recognizing an important modulatory role of sensory feedback, Brown proposed that the spinal cord contains a pattern generating apparatus with at least two antagonistically organized half-centers or a "pair of antagonistic opposites". He called this mechanism "the intrinsic factor." It was later renamed "the central pattern generator" (CPG) (Grillner & Zangger, 1975).

1.5.2 Locomotor central pattern generator (CPG)

A number of rhythmic behaviours are produced by stereotypic periodic patterns of muscle activity. After the pioneering work of Sherrington and Brown, many studies established that isolated neuronal networks in the CNS can generate commands not only for walking, but also for swimming, flying, scratching, paw-shaking, breathing, chewing and even micturition and ejaculation (Grillner, 1975; Prochazka, 1996; Kiehn *et al.*, 1998; Orlovsky *et al.*, 1999; Carro-Juarez *et al.*, 2003).

Because of the complexity of the nervous system, it is challenging to study structural details and function of the locomotor CPG in vertebrates. Simpler invertebrate neural systems have proved to be a valuable source of information on endogenous characteristics of single CPG neurons and their network interactions. Using intracellular recording techniques, single-cell oscillators have been identified in *Aplesia*, *Helix*, leech and crustacean to name a few. These oscillators produce stable periodic bursting and can be perturbed by synaptic inputs or neuromodulators, which can modify membrane properties and synaptic strength (Marder & Bucher, 2001). Mathematical models based on realistic membrane properties contributed considerably to our understanding of these "simple" systems (Kopell, 1988). For example, the role of reciprocal inhibitory connections in a dynamically coupled network of two endogenous oscillators regulating the heartbeat in the leech has been recently examined using bifurcation analysis (Cymbalyuk *et al.*, 2002). This study demonstrated how the half-center configuration of the CPG controls bursting characteristics such as period, phase and duty-cycles and incorporates characteristics of single-neuron oscillators to ensure its stable operation.

One of the intriguing properties of many endogenously bursting neurons and their follower neurons is the mechanism of the plateau potential. This enables neurons to switch between two stable membrane potentials. Short depolarizing currents can trigger a sustained membrane depolarization or plateau, whose duration outlasts the stimulus and depends on persistent, slow inward currents. In the depolarized state, the "plateau" can be rapidly terminated by a short hyperpolarizing current (Marder & Calabrese, 1996). This membrane property was found in invertebrates (Pin *et al.*, 1990; Ramirez & Pearson, 1991) as well as vertebrates (Conway *et al.*, 1988; Hounsgaard *et al.*, 1988; Hounsgaard & Kiehn, 1989; Bennett *et al.*, 1998; Gorassini *et al.*, 1999). If these bistable potentials correspond to inactive hyperpolarized and active depolarized states, then the neuron may exhibit bursting activity. In fact, plateau potentials were found to be critical for heartbeat generation in leech (Angstadt & Calabrese, 1991; Opdyke & Calabrese, 1994). Because of susceptibility to neuromodulation, e.g. by serotonin (5-hydroxytryptamine, 5HT), noradrenergic agonists (Katz & Harris-Warrick, 1990; Lee & Heckman, 1999), plateau potentials were implicated as a mechanism of reorganization of the neural networks to produce rhythmic behaviors, e.g. escape behavior in mollusks and pyloric activity in crustaceans (Katz & Harris-Warrick, 1990). Thus, plateau potentials may be responsible for endogenous bursting and shaping of the CNS output.

The location, structure and properties of neural networks implicated in locomotor pattern generation have been investigated in many different species (Kiehn et al., 1998). Generally, there are two alternative opinions about localization of the locomotor pattern generator in vertebrates. It may be confined to the rostral segments of the lumbosacral enlargement (L4-L5 spinal segments in cats) (Cazalets et al., 1995) or it may be distributed throughout the lumbosacral enlargement (Deliagina et al., 1983; Ho & O'Donovan, 1993; Kjaerulff & Kiehn, 1996; Arshavsky et al., 1997; Cowley & Schmidt, 1997; Kremer & Lev-Tov, 1997). Because it is technically challenging to monitor the activity of large neural networks, all of the studies on localization of CPGs have been carried out in reduced preparations including isolated embryonic spinal cords of rat (Lev-Tov & O'Donovan, 1995), spinalized cat (Andersson et al., 1978; Grillner & Zangger, 1979; Deliagina et al., 1983), lamprey (Zhang et al., 1996; Grillner et al., 1998) and mudpuppy (Cheng et al., 1998). Rhythmic activity of motoneuron pools which often had the specific temporal features observed in locomotion, was evoked by administration of pharmacological agents such as dihydroxyphenylalanine (DOPA), N-methyl D-aspartate (NMDA), serotonin, or by electrical stimulation of descending or peripheral sensory pathways. However, these preparations lacked either descending inputs or sensory inputs from actively moving limbs or both. It was shown that descending pathways originating in cerebral, cerebellar and midbrain areas exert a significant influence on the amplitude and timing of muscle contractions during normal locomotion (Armstrong & Marple-Horvat, 1996; Rho et al., 1997). Structures within the brainstem can neuromodulate spinal cord circuitry using descending noradrenergic and serotonergic pathways (Barbeau & Rossignol, 1990, 1991). Moreover, sensory feedback from moving limbs delivers phase-dependent modulation of motoneuron activity and initiates phase transitions (Grillner, 1975; Grillner & Rossignol, 1978; Prochazka, 1996).

One popular version of the "distributed" hypothesis describes the hindlimb CPG as a series of autonomous "unit burst generators" distributed along the neuraxis and dynamically coupled by propriospinal connections (Grillner & Wallen, 1985). Taking this representation literally, one might have assumed that the potential for rhythmogenesis is equally distributed throughout the lumbosacral enlargement. However, there is evidence that the more rostral segments of the enlargement may produce rhythmic activity more robustly than the sacral segments in response to administration of excitatory amino acids and adrenergic agonists (Kiehn & Kjaerulff, 1998; Marcoux & Rossignol, 2000). In contrast, a recent study of rhythmogenesis in response to intraspinal microstimulation showed the opposite result, namely that the caudal lumbosacral spinal cord produces rhythmic activity more robustly than its more rostral segments (Guevremont *et al.*, 2003). Unfortunately, there is no clear answer to this controversy at this time. It is possible that the rostral neural networks are more susceptible to neuromodulators, while the caudal neural networks are more sensitive to electrical stimulation.

1.5.3 Locomotor patterns generated in the presence of sensory feedback

For a hundred years, two opposing hypotheses of the role of sensory reflexes in stereotypical movements were discussed. Sensory reflexes were believed to be responsible for generation of locomotor activity according to the first hypothesis (Philippson, 1905), while according to the second hypothesis, reflexes modulated the centrally generated motor patterns (Sherrington, 1910; Brown, 1911). On one hand, deafferentation studies in the cat provided evidence that many subtle features of muscle activity during overground locomotion are present in the centrally generated patterns. For example, it was suggested that a small burst of activity in bifunctional knee flexors at the end of the swing phase was evoked by sensory feedback during overground locomotion (Engberg & Lundberg, 1969). However, it was later shown that this doublebursting persisted in animals with transected hindlimb dorsal roots and therefore had to be generated centrally (Grillner & Zangger, 1975). On the other hand, the locomotor rhythm is generally more variable after sensory feedback is abolished by deafferentation (Grillner & Zangger, 1975; Wetzel et al., 1976; Goldberger & Murray, 1980; Giuliani & Smith, 1987). Moreover, deafferented animals are less capable of compensating for changes in load and roughness of terrain (Allum et al., 1998). The locomotor deficits are even more devastating in humans with sensory neuropathy than in deafferented quadrupeds (Lajoie et al., 1996). Evidently, humans with largefibre sensory loss develop a control strategy based on passive stability (increased base of support, prolonged double stance) and multisensory information on body position from vision, vestibular systems and remaining proprioceptors, including those in the neck muscles. If vision is obstructed, these people find it difficult to walk without stumbling and falling even on a flat surface.

Several studies have indicated that humans may have a spinal locomotor CPG, but the evidence is inconclusive. People with functionally complete spinal cord injury (SCI) have been capable of producing weak locomotor-like activity inadequate for weight-bearing (Calancie *et al.*, 1994; Dimitrijevic *et al.*, 1998). However, because sensory pathways were intact, the activity could have originated from reciprocating stretch reflexes. Also, steady vibration of the legs of people with SCI induces air-stepping (Gurfinkel *et al.*, 1998). However, the origin of the rhythmic activity can be

either central or peripheral in this study also, i.e. muscle contraction may lead to pendular motion of the suspended leg and cause phasic sensory drive to muscles. Indirect evidence for the existence of the human locomotor CPG comes from studies of stepping in human infants. Human infants readily produce locomotor patterns in the presence of motion-related proprioceptive feedback, even though their cortical and cerebellar control was incomplete (Lamb & Yang, 2000).

Much of what we know about the motor control strategies for locomotion comes from studying motor activity during artificial interventions in reduced preparations. Half a century ago a Moscow group (Shik *et al.*, 1966; Shik & Orlovsky, 1976) discovered that cats decerebrated rostral to the superior colliculus could locomote when a region called the midbrain locomotor region (MLR) was electrically stimulated. In this preparation sensory information can entrain and reset the centrally generated activity (Guertin *et al.*, 1995). This was confirmed in chronic spinal cats and acute spinal cats administered with clonidine (Rossignol, 1996). For example, chronic spinal kittens could rapidly adapt the rhythm of each leg to the speed of the individual belts of a split-belt treadmill (Forssberg *et al.*, 1980). In addition, state dependent changes in neural pathways were studied at rest and during either MLR- or drug-induced locomotion in decerebrate cat (Gossard *et al.*, 1994; McCrea *et al.*, 1995). Using these preparations, details about projections of sensory pathways and their contribution to motoneuron activation during locomotion were established (reviewed in Jankowska, 2001; McCrea, 2001).

Sensory feedback is not only involved in the amplitude regulation of muscle activity (see 1.4.3), but it can also modify timing of the locomotor rhythm. A number of studies have shown that proprioceptive signals entrain, enhance and reset the ongoing locomotor activity (Duysens & Pearson, 1980; Andersson & Grillner, 1983). Figure 1.2 contains Brown's classical model of flexor

and extensor half-centers controlling each limb (Brown, 1911). In this diagram, the sensory pathways affecting the CPG, e.g. from load-sensitive afferents (Duysens & Pearson, 1980; Conway *et al.*, 1987; Pearson & Collins, 1993) are shown by loop 3, which embeds intrinsic muscle properties (loops 1), stretch reflexes (loop 2) and other pathways, e.g. beta drive to muscle spindles (Emonet-Denand *et al.*, 1975). Sensory pathways projecting to the locomotor CPG may provide its autonomous adaptation to the dynamics of the mechanical system and are important for maintaining locomotor stability (Ogihara & Yamazaki, 2001).

Multisensory signals including those related to extensor muscle force and hip position (Fig. 1.2, loop 4) have been hypothesized to initiate the phase transition from stance to swing in a number of species, e.g. locusts, crayfish and cats (Prochazka, 1996). The process can be expressed as the following simple sensory rule that "fires" when its conditions are met: IF the leg is extended AND extensor force is low, THEN initiate swing. This type of controller is called a finite state system, which was initially devised to control industrial production processes. It has also proved to be effective in the control of above-knee prostheses (Prochazka, 1993), functional electrical stimulation in human gait (Popovic *et al.*, 1989) and control of movement in biomimetic robots (Nelson & Quinn, 1999). The finite state control of locomotion may provide automatic adaptation to large perturbations of the body.

To summarize, locomotor patterns are generated by complex interactions between multiple control systems. This organization provides stable and automatic load compensation as well as higher-level control based on state regulation.

1.6 Spinal cord trauma

"Everything may die, nothing may be regenerated. It is for the science of the future to change, if possible, this harsh decree." — Santiago Ramon y Cajal (1852-1934)

Spinal cord injury generally presents as devastating damage to the nervous system produced by mechanical crush of the spinal cord by ruptured vertebral bones, which normally enclose and protect it. During the early stage, about 18 hours after injury, cell bodies and axons disrupted by the trauma undergo passive, or necrotic, death characterized by swelling and evidence of nuclear dissolution. The initial injury triggers cascades of pathological changes such as inflammation, local ischemia and release of free radicals. Glial cells including microglia and astrocytes proliferate to form a scar and increase production of axon growth inhibitory molecules (McKeon *et al.*, 1991; Fitch & Silver, 1997). The secondary damage spreads several spinal segments in the rostral and caudal directions during the next several weeks. Neurons die not only passively, but also actively, a process called apoptosis. Evidence of oligodendrocyte death and consequent demyelination of spared axons has been found below and above the lesion (Bresnahan *et al.*, 1976; Liu *et al.*, 1997).

Another key aspect of spinal cord injury is neuronal reorganization. Almost a century ago, Ramon y Cajal not only recognized that the glial scar represents a physical barrier for axon regeneration, but he also observed that injury to the optic nerve, spinal cord or cortex induces a sprouting response (Ramon y Cajal, 1928). This sprouting response is likely mediated by neurotrophic factors, such as brain-derived neurotrophic factor (BDNF) secreted from activated microglia close to the edge of the injury (Batchelor *et al.*, 2002). The new connections may even be functional (Bareyre *et al.*, 2004). Other structural and physiological changes to spinal segments

30

below the lesion include: loss of the terminals of the descending pathways (Yakovleff *et al.*, 1995), reinnervation of vacant synapses by afferent pathways, decrease in the threshold of the postsynaptic neurons, and loss of the presynaptic inhibition (Basso, 2000).

Thus, a comprehensive spinal cord repair strategy should minimize the effect of the secondary injury, remove the physical barrier formed by the cyst and glial scar, compensate for demyelination of spared axons, promote axonal long-distance regeneration and ensure proper wiring of the regenerating axons with their targets.

A number of therapies have been developed to reduce and repair trauma to the spinal cord. Primary and secondary injury may be reduced by the administration of anti-inflammatory drugs and agents reducing apoptosis, such as selective blockers of glutamate receptors (McDonald *et al.*, 1998). A number of repair and rehabilitation strategies are the subject of current research including cell transplantation (Kawaja & Gage, 1991; Tuszynski *et al.*, 1994a; Tuszynski *et al.*, 1994b; Lazarov-Spiegler *et al.*, 1996; Nakahara *et al.*, 1996; Guest *et al.*, 1997; Li *et al.*, 1997; Diener & Bregman, 1998; Ramon-Cueto *et al.*, 1998; Tuszynski *et al.*, 1998; McDonald *et al.*, 1999; Ramon-Cueto *et al.*, 2000), gene therapy (Kawaja & Gage, 1991; Tuszynski *et al.*, 1994a; Tuszynski *et al.*, 1999b; Menei *et al.*, 1998), modulation of intracellular signalling (Liu *et al.*, 1999; Cai *et al.*, 2001), manipulation of growth inhibitory environment (Schnell & Schwab, 1990; Fouad *et al.*, 1997; De Leon *et al.*, 1998; Fouad *et al.*, 2000) and restoration of lost functions using functional electrical stimulation (Brindley *et al.*, 1986; Stein *et al.*, 1992; Mushahwar *et al.*, 2000).

1.7 Thesis objective and summary

The objective of this thesis is to study the control of locomotion by multiple systems including the musculoskeletal system, sensory pathways and the lumbosacral pattern generator.

Chapter 2 deals with a controversy around intrinsic muscle properties important for modeling of muscle behavior. The study reported in this chapter tests the following two hypotheses: 1) static isometric length-tension curves of active wrist muscles apply to continuous movements; 2) hysteresis in length-tension curves of active wrist muscles is maximal at maximal muscle lengths.

Chapter 3 contains a study of stretch reflexes during locomotion. The goal of this study was to test the following two hypotheses: 1) stretch reflexes are too weak and too delayed to contribute significantly to weight-bearing; 2) the important contributions of sensory input involve state-dependent processing.

Chapter 4 describes a study of locomotor pattern generation during normal locomotion. This study introduces a novel method of analysis of the lumbosacral motoneuron activity. It reports details of the spatiotemporal motoneuron activity pattern during normal level locomotion and the implications for the locomotor control system.

Chapter 5 contains a study of fictive locomotion. In this study, the analysis of the locomotor activity revealed a novel flexibility in the pattern generation, which may reflect general properties of the locomotor pattern generating networks.

Chapter 6 describes a study of motor function recovery after spinal cord injury. It tests the following hypothesis: the observed motor improvements after long distance axon regeneration may be explained by excitation of propriospinal neurons that relay excitation to the locomotor CPG.

Finally, Chapter 7 contains a summary, general discussion and future directions of the studies described in this thesis.

1.8 References

- ABBOT, B. C. & AUBERT, X. M. (1951). The force exerted by active striated muscle during and after change in length. J Physiol 117, 77-86.
- ABBOT, B. C. & WILKIE, D. R. (1953). The relation between velocity of shortening and the tensionlength curve of skeletal muscle. *J Physiol* 120, 214-223.
- AKAZAWA, K., ALDRIDGE, J. W., STEEVES, J. D. & STEIN, R. B. (1982). Modulation of stretch reflexes during locomotion in the mesencephalic cat. *J Physiol* 329, 553-567.
- ALLUM, J. H., BLOEM, B. R., CARPENTER, M. G., HULLIGER, M. & HADDERS-ALGRA, M. (1998). Proprioceptive control of posture: a review of new concepts. *Gait Posture* 8, 214-242.
- ANDERSON, J. H. (1974). Dynamic characteristics of Golgi tendon organs. Brain Res 67, 531-537.
- ANDERSSON, O., FORSSBERG, H., GRILLNER, S. & LINDQUIST, M. (1978). Phasic gain control of the transmission in cutaneous reflex pathways to motoneurones during 'fictive' locomotion. *Brain Res* 149, 503-507.
- ANDERSSON, O. & GRILLNER, S. (1983). Peripheral control of the cat's step cycle. II. Entrainment of the central pattern generators for locomotion by sinusoidal hip movements during "fictive locomotion." *Acta Physiol Scand* **118**, 229-239.
- ANGSTADT, J. D. & CALABRESE, R. L. (1991). Calcium currents and graded synaptic transmission between heart interneurons of the leech. *J Neurosci* 11, 746-759.
- ARMSTRONG, D. M. & MARPLE-HORVAT, D. E. (1996). Role of the cerebellum and motor cortex in the regulation of visually controlled locomotion. *Can J Physiol Pharmacol* 74, 443-455.
- ARSHAVSKY, Y. I., DELIAGINA, T. G. & ORLOVSKY, G. N. (1997). Pattern generation. Curr Opin Neurobiol 7, 781-789.
- BAHLER, A. S., FALES, J. T. & ZIERLER, K. L. (1968). The dynamic properties of mammalian skeletal muscle. *J Gen Physiol* 51, 369-384.
- BARBEAU, H. & ROSSIGNOL, S. (1987). Recovery of locomotion after chronic spinalization in the adult cat. *Brain Res* 412, 84-95.

- BARBEAU, H. & ROSSIGNOL, S. (1990). The effects of serotonergic drugs on the locomotor pattern and on cutaneous reflexes of the adult chronic spinal cat. *Brain Res* **514**, 55-67.
- BARBEAU, H. & ROSSIGNOL, S. (1991). Initiation and modulation of the locomotor pattern in the adult chronic spinal cat by noradrenergic, serotonergic and dopaminergic drugs. *Brain Res* 546, 250-260.
- BAREYRE, F. M., KERSCHENSTEINER, M., RAINETEAU, O., METTENLEITER, T. C., WEINMANN, O. & SCHWAB, M. E. (2004). The injured spinal cord spontaneously forms a new intraspinal circuit in adult rats. *Nat Neurosci* 7, 269-277.
- BARKER, D. (1974). The morphology of muscle receptors. In *Handbook of Sensory Physiology*. ed. HUNT, C. C. Springer, Berlin.
- BASSO, D. M. (2000). Neuroanatomical substrates of functional recovery after experimental spinal cord injury: implications of basic science research for human spinal cord injury. *Phys Ther* 80, 808-817.
- BATCHELOR, P. E., PORRITT, M. J., MARTINELLO, P., PARISH, C. L., LIBERATORE, G. T., DONNAN, G. A. & HOWELLS, D. W. (2002). Macrophages and Microglia Produce Local Trophic Gradients That Stimulate Axonal Sprouting Toward but Not beyond the Wound Edge. *Mol Cell Neurosci* 21, 436-453.
- BELANGER, A. Y. & MCCOMAS, A. J. (1989). Contractile properties of human skeletal muscle in childhood and adolescence. *Eur J Appl Physiol Occup Physiol* 58, 563-567.
- BENNETT, D. J., DE SERRES, S. J. & STEIN, R. B. (1996). Gain of the triceps surae stretch reflex in decerebrate and spinal cats during postural and locomotor activities. *J Physiol* 496 (Pt 3), 837-850.
- BENNETT, D. J., HULTBORN, H., FEDIRCHUK, B. & GORASSINI, M. (1998). Synaptic Activation of Plateaus in Hindlimb Motoneurons of Decerebrate Cats. *J Neurophysiol* 80, 2023-2037.
- BERNSTEIN, N. (1967). The Coordination and Regulation of Movements. Pergamon, Oxford.
- BERTRAM, J. E., RUINA, A., CANNON, C. E., CHANG, Y. H. & COLEMAN, M. J. (1999). A point-mass model of gibbon locomotion. *J Exp Biol* 202, 2609-2617.
- BLICKHAN, R. (1989). The spring-mass model for running and hopping. J Biomech 22, 1217-1227.
- BLIX, M. (1893). Die Lange und Spannung des Muskels. Skand. Arch. Physiol. 3, 295-318.
- BRESNAHAN, J. C., KING, J. S., MARTIN, G. F. & YASHON, D. (1976). A neuroanatomical analysis of spinal cord injury in the rhesus monkey (Macaca mulatta). *J Neurol Sci* 28, 521-542.

- BRINDLEY, G. S., POLKEY, C. E., RUSHTON, D. N. & CARDOZO, L. (1986). Sacral anterior root stimulators for bladder control in paraplegia: the first 50 cases. J Neurol Neurosurg Psychiatry 49, 1104-1114.
- BROOKE, J. D., CHENG, J., COLLINS, D. F., MCILROY, W. E., MISIASZEK, J. E. & STAINES, W. R. (1997). Sensori-sensory afferent conditioning with leg movement: gain control in spinal reflex and ascending paths. *Prog Neurobiol* 51, 393-421.
- BROWN, I. E., LIINAMAA, T. L. & LOEB, G. E. (1996). Relationships between range of motion, lo, and passive force in five strap-like muscles of the feline hind limb. *J Morphol* 230, 69-77.
- BROWN, T. G. (1911). The intrinsic factor in the act of progression in the mammal, vol. 84. Proc. Roy. Soc., London.
- CAI, D., QIU, J., CAO, Z., MCATEE, M., BREGMAN, B. S. & FILBIN, M. T. (2001). Neuronal cyclic AMP controls the developmental loss in ability of axons to regenerate. *J Neurosci* 21, 4731-4739.
- CALANCIE, B., NEEDHAM-SHROPSHIRE, B., JACOBS, P., WILLER, K., ZYCH, G. & GREEN, B. A. (1994). Involuntary stepping after chronic spinal cord injury. Evidence for a central rhythm generator for locomotion in man. *Brain* 117 (Pt 5), 1143-1159.
- CAPADAY, C. (2000). Control of a 'simple' stretch reflex in humans. Trends Neurosci 23, 528-530.
- CAPADAY, C. & STEIN, R. B. (1986). Amplitude modulation of the soleus H-reflex in the human during walking and standing. *J Neurosci* 6, 1308-1313.
- CAPADAY, C. & STEIN, R. B. (1987). Difference in the amplitude of the human soleus H reflex during walking and running. *J Physiol* 392, 513-522.
- CARRO-JUAREZ, M., CRUZ, S. L. & RODRIGUEZ-MANZO, G. (2003). Evidence for the involvement of a spinal pattern generator in the control of the genital motor pattern of ejaculation. *Brain Res* 975, 222-228.
- CAZALETS, J. R., BORDE, M. & CLARAC, F. (1995). Localization and organization of the central pattern generator for hindlimb locomotion in newborn rat. *J Neurosci* 15, 4943-4951.
- CHENG, E. J., BROWN, I. E. & LOEB, G. E. (2000). Virtual muscle: a computational approach to understanding the effects of muscle properties on motor control. *J Neurosci Methods* 101, 117-130.
- CHENG, J., STEIN, R. B., JOVANOVIC, K., YOSHIDA, K., BENNETT, D. J. & HAN, Y. (1998). Identification, localization, and modulation of neural networks for walking in the mudpuppy (Necturus maculatus) spinal cord. *J Neurosci* 18, 4295-4304.

- CHRISTENSEN, L. O., ANDERSEN, J. B., SINKJAER, T. & NIELSEN, J. (2001). Transcranial magnetic stimulation and stretch reflexes in the tibialis anterior muscle during human walking. *J Physiol* **531**, 545-557.
- COLLINS, D. F. & PROCHAZKA, A. (1996). Movement illusions evoked by ensemble cutaneous input from the dorsum of the human hand. *J Physiol* 496 (Pt 3), 857-871.
- CONWAY, B. A., HULTBORN, H. & KIEHN, O. (1987). Proprioceptive input resets central locomotor rhythm in the spinal cat. *Exp Brain Res* 68, 643-656.
- CONWAY, B. A., HULTBORN, H., KIEHN, O. & MINTZ, I. (1988). Plateau potentials in alphamotoneurones induced by intravenous injection of L-dopa and clonidine in the spinal cat. *J Physiol* 405, 369-384.
- COWLEY, K. C. & SCHMIDT, B. J. (1997). Regional distribution of the locomotor pattern-generating network in the neonatal rat spinal cord. *J Neurophysiol* 77, 247-259.
- CYMBALYUK, G. S., GAUDRY, Q., MASINO, M. A. & CALABRESE, R. L. (2002). Bursting in leech heart interneurons: cell-autonomous and network-based mechanisms. *J Neurosci* 22, 10580-10592.
- DE LEON, R. D., HODGSON, J. A., ROY, R. R. & EDGERTON, V. R. (1998). Full weight-bearing hindlimb standing following stand training in the adult spinal cat. *J Neurophysiol* 80, 83-91.
- DELIAGINA, T. G., ORLOVSKY, G. N. & PAVLOVA, G. A. (1983). The capacity for generation of rhythmic oscillations is distributed in the lumbosacral spinal cord of the cat. *Exp Brain Res* 53, 81-90.
- DIENER, P. S. & BREGMAN, B. S. (1998). Fetal spinal cord transplants support the development of target reaching and coordinated postural adjustments after neonatal cervical spinal cord injury. *J Neurosci* 18, 763-778.
- DIETZ, V. (1996). Interaction between central programs and afferent input in the control of posture and locomotion. *J Biomech* 29, 841-844.
- DIETZ, V. (1998). Evidence for a load receptor contribution to the control of posture and locomotion. *Neurosci Biobebav Rev* 22, 495-499.
- DIMITRIJEVIC, M. R., GERASIMENKO, Y. & PINTER, M. M. (1998). Evidence for a Spinal Central Pattern Generator in Humans. *Ann NY Acad Sci* 860, 360-376.
- DONELAN, J. M., KRAM, R. & KUO, A. D. (2001). Mechanical and metabolic determinants of the preferred step width in human walking. *Proc R Soc Lond B Biol Sci* 268, 1985-1992.

- DUYSENS, J., CLARAC, F. & CRUSE, H. (2000). Load-regulating mechanisms in gait and posture: comparative aspects. *Physiol Rev* 80, 83-133.
- DUYSENS, J. & PEARSON, K. G. (1980). Inhibition of flexor burst generation by loading ankle extensor muscles in walking cats. *Brain Res* 187, 321-332.
- EDGERTON, V. R., DE LEON, R. D., TILLAKARATNE, N., RECKTENWALD, M. R., HODGSON, J. A. & ROY, R. R. (1997). Use-dependent plasticity in spinal stepping and standing. *Adv Neurol* 72, 233-247.
- EDMAN, K. A. & ANDERSSON, K. E. (1968). The variation in active tension with sarcomere length in vertebrate skeletal muscle and its relation to fibre width. *Experientia* 24, 134-136.
- EDMAN, K. A., ELZINGA, G. & NOBLE, M. I. (1978). Enhancement of mechanical performance by stretch during tetanic contractions of vertebrate skeletal muscle fibres. *J Physiol* 281, 139-155.
- EDMAN, K. A., ELZINGA, G. & NOBLE, M. I. (1982). Residual force enhancement after stretch of contracting frog single muscle fibers. *J Gen Physiol* 80, 769-784.
- EDMAN, K. A. & REGGIANI, C. (1984). Redistribution of sarcomere length during isometric contraction of frog muscle fibres and its relation to tension creep. *J Physiol* **351**, 169-198.
- EMONET-DENAND, F., JAMI, L. & LAPORTE, Y. (1975). Skeleto-fusimotor axons in the hind-limb muscles of the cat. *J Physiol* 249, 153-166.
- ENGBERG, I. & LUNDBERG, A. (1969). An electromyographic analysis of muscular activity in the hindlimb of the cat during unrestrained locomotion. Acta Physiol Scand 75, 614-630.
- FELDMAN, A. G. (1966). Functional tuning of the nervous system with control of movement and maintenance of steady posture. II. Controllable parameters of the muscle. *Biophysics* 11, 565-578.
- FITCH, M. T. & SILVER, J. (1997). Activated macrophages and the blood-brain barrier: inflammation after CNS injury leads to increases in putative inhibitory molecules. *Exp Neurol* 148, 587-603.
- FORSSBERG, H., GRILLNER, S., HALBERTSMA, J. & ROSSIGNOL, S. (1980). The locomotion of the low spinal cat. II. Interlimb coordination. *Acta Physiol Scand* **108**, 283-295.
- FOUAD, K., DIETZ, V. & SCHWAB, M. E. (2001). Improving axonal growth and functional recovery after experimental spinal cord injury by neutralizing myelin associated inhibitors. *Brain Res Brain Res Rev* 36, 204-212.

- FOUAD, K., METZ, G. A., MERKLER, D., DIETZ, V. & SCHWAB, M. E. (2000). Treadmill training in incomplete spinal cord injured rats. *Behav Brain Res* **115**, 107-113.
- FREUSBERG, A. (1874). Reflexbewegungen beim Hunde. Pflugers Arch. Physiol. 9, 358-391.
- FUCHS, F. (1977). Cooperative interactions between calcium-binding sites on glycerinated muscle fibers. The influence of cross-bridge attachment. *Biochim Biophys Acta* 462, 314-322.
- GARCIA, M., CHATTERJEE, A., RUINA, A. & COLEMAN, M. (1998). The simplest walking model: stability, complexity, and scaling. *J Biomech Eng* 120, 281-288.
- GAREIS, H., SOLOMONOW, M., BARATTA, R., BEST, R. & D'AMBROSIA, R. (1992). The isometric length-force models of nine different skeletal muscles. *J Biomech* 25, 903-916.
- GARRETT, M. & LUCKWILL, R. G. (1983). Role of reflex responses of knee musculature during the swing phase of walking in man. *Eur J Appl Physiol Occup Physiol* **52**, 36-41.
- GARRETT, M., LUCKWILL, R. G. & MCALEER, J. J. A. (1981). The sensitivity of the monosynaptic reflex arc in the leg extensor muscles of the walking man. In *Biomechanics*, vol. VII. ed. MORECKI, A. University Park Press, Baltimore.
- GERRITSEN, K. G., VAN DEN BOGERT, A. J., HULLIGER, M. & ZERNICKE, R. F. (1998). Intrinsic muscle properties facilitate locomotor control a computer simulation study. *Motor Control* 2, 206-220.
- GHEZ, C. & KRAKAUER, J. (2000). The organization of movement. In *Principles of Neural Science*. ed. KANDEL, E. R., SCHWARTZ, J. H. & JESSELL, T. M., pp. 653. McGraw-Hill, New York.
- GILLARD, D. M., YAKOVENKO, S., CAMERON, T. & PROCHAZKA, A. (2000). Isometric muscle length-tension curves do not predict angle-torque curves of human wrist in continuous active movements. *J Biomech* 33, 1341-1348.
- GIULIANI, C. A. & SMITH, J. L. (1987). Stepping behaviors in chronic spinal cats with one hindlimb deafferented. *J Neurosci* 7, 2537-2546.
- GOLDBERGER, M. E. & MURRAY, M. (1980). Locomotor recovery after deafferentation of one side of the cat's trunk. *Exp Neurol* 67, 103-117.
- GORASSINI, M., BENNETT, D. J., KIEHN, O., EKEN, T. & HULTBORN, H. (1999). Activation patterns of hindlimb motor units in the awake rat and their relation to motoneuron intrinsic properties. *J Neurophysiol* 82, 709-717.
- GORDON, A. M., HUXLEY, A. F. & JULIAN, F. J. (1966). The variation in isometric tension with sarcomere length in vertebrate muscle fibres. *J Physiol* **184**, 170-192.

- GORDON, T. & MAO, J. (1994). Muscle atrophy and procedures for training after spinal cord injury. *Phys Ther* 74, 50-60.
- GORDON, T., YANG, J. F., AYER, K., STEIN, R. B. & TYREMAN, N. (1993). Recovery potential of muscle after partial denervation: a comparison between rats and humans. *Brain Res Bull* 30, 477-482.
- GOSSARD, J. P., BROWNSTONE, R. M., BARAJON, I. & HULTBORN, H. (1994). Transmission in a locomotorrelated group Ib pathway from hindlimb extensor muscles in the cat. *Exp Brain Res* 98, 213-228.
- GREY, M. J., LADOUCEUR, M., ANDERSEN, J. B., NIELSEN, J. B. & SINKJAER, T. (2001). Group II muscle afferents probably contribute to the medium latency soleus stretch reflex during walking in humans. J Physiol 534, 925-933.
- GRILLNER, S. (1975). Locomotion in vertebrates: central mechanisms and reflex interaction. *Physiol Rev* 55, 247-304.
- GRILLNER, S., PARKER, D. & EL MANIKA, A. (1998). Vertebrate locomotion a lamprey perspective. In Annals of the New York Academy of Sciences, vol. 860, pp. 1-18. New York Academy of Sciences, New York.
- GRILLNER, S. & ROSSIGNOL, S. (1978). On the initiation of the swing phase of locomotion in chronic spinal cats. *Brain Res* 146, 269-277.
- GRILLINER, S. & WALLEN, P. (1985). Central pattern generators for locomotion, with special reference to vertebrates. *Annu Rev Neurosci* 8, 233-261.
- GRILLNER, S. & ZANGGER, P. (1975). How detailed is the central pattern generation for locomotion? Brain Res 88, 367-371.
- GRILLNER, S. & ZANGGER, P. (1979). On the central generation of locomotion in the low spinal cat. *Exp Brain Res* 34, 241-261.
- GRITSENKO, V., MUSHAHWAR, V. & PROCHAZKA, A. (2001). Adaptive changes in locomotor control after partial denervation of triceps surae muscles in the cat. *J Physiol (Lond)* 533, 299-311.
- GUERTIN, P., ANGEL, M. J., PERREAULT, M. C. & MCCREA, D. A. (1995). Ankle extensor group I afferents excite extensors throughout the hindlimb during fictive locomotion in the cat. *J Physiol* 487 (Pt 1), 197-209.

- GUEST, J. D., RAO, A., OLSON, L., BUNGE, M. B. & BUNGE, R. P. (1997). The ability of human Schwann cell grafts to promote regeneration in the transected nude rat spinal cord. *Exp Neurol* **148**, 502-522.
- GUEVREMONT, L., RENZI, C. G., KOWALCZEWSKI, J., SAIGAL, R. & MUSHAHWAR, V. K. (2003). Tonic intraspinal microstimulation of the lumbosacral cord generates rhythmic walking in spinal cats. In *Society for Neuroscience Meeting*, Orlando.
- GURFINKEL, V. S., LEVIK, Y. S., KAZENNIKOV, O. V. & SELIONOV, V. A. (1998). Locomotor-like movements evoked by leg muscle vibration in humans. *Eur J Neurosci* 10, 1608-1612.
- HE, J., LEVINE, W. S. & LOEB, G. E. (1991). Feedback gains for correcting small perturbations to standing posture. *Automatic Control, IEEE Transactions on* 36, 322-332.
- HERR, H. M., HUANG, G. T. & MCMAHON, T. A. (2002). A model of scale effects in mammalian quadrupedal running. *J Exp Biol* 205, 959-967.
- HERZOG, W. & LEONARD, T. R. (2002). Force enhancement following stretching of skeletal muscle: a new mechanism. *J Exp Biol* 205, 1275-1283.
- HERZOG, W., LEONARD, T. R. & WU, J. Z. (2000). The relationship between force depression following shortening and mechanical work in skeletal muscle. *J Biomech* 33, 659-668.
- HIEBERT, G. W., WHELAN, P. J., PROCHAZKA, A. & PEARSON, K. G. (1996). Contribution of hind limb flexor muscle afferents to the timing of phase transitions in the cat step cycle. J Neurophysiol 75, 1126-1137.
- HILL, A. V. (1938). The heat of shortening and the dynamic constants of muscle. *Proc. R. Soc. Lond. B* 126, 136-195.
- HO, S. & O'DONOVAN, M. J. (1993). Regionalization and intersegmental coordination of rhythmgenerating networks in the spinal cord of the chick embryo. *J Neurosci* 13, 1354-1371.
- HOUK, J. & HENNEMAN, E. (1967). Responses of Golgi tendon organs to active contractions of the soleus muscle of the cat. *J Neurophysiol* 30, 466-481.
- HOUK, J. & SIMON, W. (1967). Responses of Golgi tendon organs to forces applied to muscle tendon. *J Neurophysiol* **30**, 1466-1481.
- HOUNSGAARD, J., HULTBORN, H., JESPERSEN, B. & KIEHN, O. (1988). Bistability of alpha-motoneurones in the decerebrate cat and in the acute spinal cat after intravenous 5-hydroxytryptophan. J Physiol 405, 345-367.

- HOUNSGAARD, J. & KIEHN, O. (1989). Serotonin-induced bistability of turtle motoneurones caused by a nifedipine-sensitive calcium plateau potential. *J Physiol* 414, 265-282.
- HOY, M. G., ZAJAC, F. E. & GORDON, M. E. (1990). A musculoskeletal model of the human lower extremity: the effect of muscle, tendon, and moment arm on the moment-angle relationship of musculotendon actuators at the hip, knee, and ankle. *J Biomech* 23, 157-169.
- HOY, M. G. & ZERNICKE, R. F. (1985). Modulation of limb dynamics in the swing phase of locomotion. *J Biomech* 18, 49-60.
- HUIJING, P. A. (1985). Architecture of the human gastrocnemius muscle and some functional consequences. Acta Anat (Basel) 123, 101-107.
- HUIJING, P. A. (2000). Modeling of homogeneous muscle: Is it realistic to consider skeletal muscle as a lumped sarcomere of fiber? In *Biomechanics and neural control of posture and movement*. ed. WINTERS, J. M. & CRAGO, P. E., pp. 92-98. Springer-Verlag, New York.
- HULLIGER, M. (1984). The mammalian muscle spindle and its central control. *Rev Physiol Biochem Pharmacol* 101, 1-110.
- HUXLEY, A. F. (1957). Muscle structure and theories of contraction. *Prog. Biophys. Mol. Biol.* 7, 255–318.
- HUXLEY, A. F. (1974). Muscular contraction. J Physiol 243, 1-43.
- HUXLEY, A. F. & NIEDERGERKE, R. (1954). Structural changes in muscle during contraction. *Nature* 173, 971-973.
- HUXLEY, H. E. & HANSON, J. (1954). Changes in the cross-striations of muscle during contraction and stretch and their structural interpretation. *Nature* **173**, 973-976.
- JAMI, L. (1992). Golgi tendon organs in mammalian skeletal muscle: functional properties and central actions. *Physiol Rev* 72, 623-666.
- JAMI, L., PETIT, J., PROSKE, U. & ZYTNICKI, D. (1985). Responses of tendon organs to unfused contractions of single motor units. *J Neurophysiol* 53, 32-42.
- JANKOWSKA, E. (2001). Spinal interneuronal systems: identification, multifunctional character and reconfigurations in mammals. *J Physiol* 533, 31-40.
- JOYCE, G. C., RACK, P. M. & WESTBURY, D. R. (1969). The mechanical properties of cat soleus muscle during controlled lengthening and shortening movements. *J Physiol* 204, 461-474.

- KATZ, B. (1939). The relation between force and speed in muscular contraction. J. Physiol. 96, 45-64.
- KATZ, P. S. & HARRIS-WARRICK, R. M. (1990). Neuromodulation of the crab pyloric central pattern generator by serotonergic/cholinergic proprioceptive afferents. *J Neurosci* 10, 1495-1512.
- KAWAJA, M. D. & GAGE, F. H. (1991). Reactive astrocytes are substrates for the growth of adult CNS axons in the presence of elevated levels of nerve growth factor. *Neuron* 7, 1019-1030.
- KELLER, T. S., STRAUSS, A. M. & SZPALSKI, M. (1992). Prevention of bone loss and muscle atrophy during manned space flight. *Microgravity Q* 2, 89-102.
- KIEHN, O., HARRIS-WARRICK, R. M., JORDAN, L. M., HULTBORN, H. & KUDO, N. (1998). Neuronal Mechanisms for Generating Locomotor Activity, vol. 860. New York Academy of Sciences, New York.
- KIEHN, O. & KJAERULFF, O. (1998). Distribution of central pattern generators for rhythmic motor outputs in the spinal cord of limbed vertebrates. In *Annals of the New York Academy of Sciences*, vol. 860. ed. KIEHN, O., HARRIS-WARRICK, R. M., JORDAN, L. M., HULTBORN, H. & KUDO, N., pp. 110–129. New York Academy of Sciences, New York.
- KJAERULFF, O. & KIEHN, O. (1996). Distribution of networks generating and coordinating locomotor activity in the neonatal rat spinal cord in vitro: a lesion study. *J Neurosci* 16, 5777-5794.
- KOH, T. J. & HERZOG, W. (1998). Excursion is important in regulating sarcomere number in the growing rabbit tibialis anterior. *J Physiol* **508** (Pt 1), 267-280.
- KOPELL, N. (1988). Toward a theory of modeling central pattern generators. In *Neural Control of Rhythmic Movements in Vertebrates*, vol. 10. ed. COHEN, A. H., ROSSIGNOL, S. & GRILLNER, S., pp. 369-413. Wiley, New York.
- KREMER, E. & LEV-TOV, A. (1997). Localization of the spinal network associated with generation of hindlimb locomotion in the neonatal rat and organization of its transverse coupling system. *J Neurophysiol* 77, 1155-1170.
- LAJOIE, Y., TEASDALE, N., COLE, J. D., BURNETT, M., BARD, C., FLEURY, M., FORGET, R., PAILLARD, J. & LAMARRE, Y. (1996). Gait of a deafferented subject without large myelinated sensory fibers below the neck. *Neurology* 47, 109-115.
- LAMB, T. & YANG, J. F. (2000). Could Different Directions of Infant Stepping Be Controlled by the Same Locomotor Central Pattern Generator? *J Neurophysiol* 83, 2814-2824.

- LAWRENCE, J. H., 3RD & NICHOLS, T. R. (1999). A three-dimensional biomechanical analysis of the cat ankle joint complex: II. Effect of ankle joint orientation on evoked isometric joint torque. *Journal of Applied Biomechanics* 15, 106-119.
- LAWRENCE, J. H., 3RD, NICHOLS, T. R. & ENGLISH, A. W. (1993). Cat hindlimb muscles exert substantial torques outside the sagittal plane. *J Neurophysiol* 69, 282-285.
- LAZAROV-SPIEGLER, O., SOLOMON, A. S., ZEEV-BRANN, A. B., HIRSCHBERG, D. L., LAVIE, V. & SCHWARTZ, M. (1996). Transplantation of activated macrophages overcomes central nervous system regrowth failure. *Faseb J* 10, 1296-1302.
- LEE, C. R. & FARLEY, C. T. (1998). Determinants of the center of mass trajectory in human walking and running. *J Exp Biol* 201 (Pt 21), 2935-2944.
- LEE, R. H. & HECKMAN, C. J. (1999). Enhancement of Bistability in Spinal Motoneurons In Vivo by the Noradrenergic alpha 1 Agonist Methoxamine. *J Neurophysiol* **81**, 2164-2174.
- LEV-TOV, A. & O'DONOVAN, M. J. (1995). Calcium imaging of motoneuron activity in the en-bloc spinal cord preparation of the neonatal rat. *J Neurophysiol* 74, 1324-1334.
- LI, Y., FIELD, P. M. & RAISMAN, G. (1997). Repair of adult rat corticospinal tract by transplants of olfactory ensheathing cells. *Science* 277, 2000-2002.
- LIEBER, R. L. & FRIDEN, J. (1998). Musculoskeletal balance of the human wrist elucidated using intraoperative laser diffraction. *J Electromyogr Kinesiol* 8, 93-100.
- LIEBER, R. L., LOREN, G. J. & FRIDEN, J. (1994). In vivo measurement of human wrist extensor muscle sarcomere length changes. *J Neurophysiol* 71, 874-881.
- LIU, X. Z., XU, X. M., HU, R., DU, C., ZHANG, S. X., MCDONALD, J. W., DONG, H. X., WU, Y. J., FAN, G. S., JACQUIN, M. F., HSU, C. Y. & CHOI, D. W. (1997). Neuronal and Glial Apoptosis after Traumatic Spinal Cord Injury. J. Neurosci. 17, 5395-5406.
- LIU, Y., KIM, D., HIMES, B. T., CHOW, S. Y., SCHALLERT, T., MURRAY, M., TESSLER, A. & FISCHER, I. (1999). Transplants of fibroblasts genetically modified to express BDNF promote regeneration of adult rat rubrospinal axons and recovery of forelimb function. J Neurosci 19, 4370-4387.
- LLEWELLYN, M., YANG, J. F. & PROCHAZKA, A. (1990). Human H-reflexes are smaller in difficult beam walking than in normal treadmill walking. *Exp Brain Res* 83, 22-28.
- LOEB, G. E. (2000). Comentary: What is a reflex? In *Biomechanics and neural control of posture* and movement. ed. WINTERS, J. M. & CRAGO, P. E., pp. 251-252. Springer-Verlag, New York.

- LOEB, G. E. & DUYSENS, J. (1979). Activity patterns in individual hindlimb primary and secondary muscle spindle afferents during normal movements in unrestrained cats. *J Neurophysiol* 42, 420-440.
- LOREN, G. J., SHOEMAKER, S. D., BURKHOLDER, T. J., JACOBSON, M. D., FRIDEN, J. & LIEBER, R. L. (1996). Human wrist motors: biomechanical design and application to tendon transfers. *J Biomech* 29, 331-342.
- LYNN, R. & MORGAN, D. L. (1994). Decline running produces more sarcomeres in rat vastus intermedius muscle fibers than does incline running. *J Appl Physiol* 77, 1439-1444.
- LYNN, R., TALBOT, J. A. & MORGAN, D. L. (1998). Differences in rat skeletal muscles after incline and decline running. *J Appl Physiol* 85, 98-104.
- MACPHERSON, J. M. (1988). Strategies that simplify the control of quadrupedal stance. I. Forces at the ground. *J Neurophysiol* 60, 204-217.
- MALAMUD, J. G., GODT, R. E. & NICHOLS, T. R. (1996). Relationship between short-range stiffness and yielding in type-identified, chemically skinned muscle fibers from the cat triceps surae muscles. *J Neurophysiol* 76, 2280-2289.
- MARCOUX, J. & ROSSIGNOL, S. (2000). Initiating or blocking locomotion in spinal cats by applying noradrenergic drugs to restricted lumbar spinal segments [In Process Citation]. J Neurosci 20, 8577-8585.
- MARDER, E. & BUCHER, D. (2001). Central pattern generators and the control of rhythmic movements. *Curr Biol* 11, R986-996.
- MARDER, E. & CALABRESE, R. L. (1996). Principles of rhythmic motor pattern generation. *Physiol Rev* 76, 687-717.
- MATTHEWS, P. B. (1991). The human stretch reflex and the motor cortex. Trends Neurosci 14, 87-91.
- MATTHEWS, P. B. & STEIN, R. B. (1969). The regularity of primary and secondary muscle spindle afferent discharges. *J Physiol* 202, 59-82.
- MATTHEWS, P. B. C. (1972). Mammalian Muscle Receptors and Their Central Actions. Arnold, London.
- MCCREA, D. A. (2001). Spinal circuitry of sensorimotor control of locomotion. J Physiol (Lond) 533, 41-50.

- MCCREA, D. A., SHEFCHYK, S. J., STEPHENS, M. J. & PEARSON, K. G. (1995). Disynaptic group I excitation of synergist ankle extensor motoneurones during fictive locomotion in the cat. *J Physiol* **487 (Pt 2)**, 527-539.
- MCDONALD, J. W., ALTHOMSONS, S. P., HYRC, K. L., CHOI, D. W. & GOLDBERG, M. P. (1998). Oligodendrocytes from forebrain are highly vulnerable to AMPA/kainate receptor-mediated excitotoxicity. *Nat Med* 4, 291-297.
- MCDONALD, J. W., LIU, X. Z., QU, Y., LIU, S., MICKEY, S. K., TURETSKY, D., GOTTLIEB, D. I. & CHOI, D. W. (1999). Transplanted embryonic stem cells survive, differentiate and promote recovery in injured rat spinal cord. *Nat Med* 5, 1410-1412.
- MCGEER, T. (1990). Passive bipedal running. Proc R Soc Lond B Biol Sci 240, 107-134.
- MCKEON, R. J., SCHREIBER, R. C., RUDGE, J. S. & SILVER, J. (1991). Reduction of neurite outgrowth in a model of glial scarring following CNS injury is correlated with the expression of inhibitory molecules on reactive astrocytes. *J Neurosci* 11, 3398-3411.
- MEIJER, K., GROOTEBOER, H. J., KOOPMAN, B. F. J. M. & HUIJING, P. A. (1997). Fully isometric lengthforce curves of rat muscle differ from those during and after concentric contractions. *Journal of Applied Biomechanics* 13, 164-181.
- MENEI, P., MONTERO-MENEI, C., WHITTEMORE, S. R., BUNGE, R. P. & BUNGE, M. B. (1998). Schwann cells genetically modified to secrete human BDNF promote enhanced axonal regrowth across transected adult rat spinal cord. *Eur J Neurosci* 10, 607-621.
- METTRIE, D. L., J.O. (1745). L'Homme-Machine. English translation: Man a machine; and Man a plant (1994). Maya Hackett, Indianapolis.
- MISIASZEK, J. E., CHENG, J., BROOKE, J. D. & STAINES, W. R. (1998). Movement-induced modulation of soleus H reflexes with altered length of biarticular muscles. *Brain Res* **795**, 25-36.
- MISIASZEK, J. E., STEPHENS, M. J., YANG, J. F. & PEARSON, K. G. (2000). Early corrective reactions of the leg to perturbations at the torso during walking in humans. *Exp Brain Res* 131, 511-523.
- MOBERG, E. (1983). The role of cutaneous afferents in position sense, kinaesthesia, and motor function of the hand. *Brain* 106 (Pt 1), 1-19.

MOCHON, S. & MCMAHON, T. A. (1980). Ballistic walking. J Biomech 13, 49-57.

MUSHAHWAR, V. K., COLLINS, D. F. & PROCHAZKA, A. (2000). Spinal cord microstimulation generates functional limb movements in chronically implanted cats. *Exp Neurol* 163, 422-429.

MUYBRIDGE, E. (1887). Animal Locomotion, Philadelphia.

- NAKAHARA, Y., GAGE, F. H. & TUSZYNSKI, M. H. (1996). Grafts of fibroblasts genetically modified to secrete NGF, BDNF, NT-3, or basic FGF elicit differential responses in the adult spinal cord. *Cell Transplant* 5, 191-204.
- NAVARRETTE, R. & VRBOVA, G. (1993). Activity-dependent interactions between motoneurones and muscles: their role in the development of the motor unit. *Prog Neurobiol* 41, 93-124.
- NELSON, G. M. & QUINN, R. D. (1999). Posture control of a cockroach-like robot. *IEEE Control Systems* 19, 9-14.
- NEPTUNE, R. R., KAUTZ, S. A. & ZAJAC, F. E. (2000). Muscle contributions to specific biomechanical functions do not change in forward versus backward pedaling. *J Biomech* 33, 155-164.
- OGIHARA, N. & YAMAZAKI, N. (2001). Generation of human bipedal locomotion by a bio-mimetic neuro-musculo-skeletal model. *Biol Cybern* 84, 1-11.
- OPDYKE, C. A. & CALABRESE, R. L. (1994). A persistent sodium current contributes to oscillatory activity in heart interneurons of the medicinal leech. J Comp Physiol [A] 175, 781-789.
- ORLOVSKY, G. N., DELIAGINA, T. G. & GRILLNER, S. (1999). Neuronal control of locomotion. Oxford University Press, Oxford.
- ORLOVSKY, G. N. & SHIK, M. L. (1965). Standard elements of cyclic movement. Biofizika 5, 847-854.
- PANDY, M. G., ZAJAC, F. E., SIM, E. & LEVINE, W. S. (1990). An optimal control model for maximumheight human jumping. *J Biomech* 23, 1185-1198.
- PANG, M. Y. & YANG, J. F. (2000). The initiation of the swing phase in human infant stepping: importance of hip position and leg loading. *J Physiol* **528** Pt 2, 389-404.
- PARTRIDGE, L. D. (1966). Signal-handling characteristics of load-moving skeletal muscle. *Am J Physiol* **210**, 1178-1191.
- PEARSON, K. G. & COLLINS, D. F. (1993). Reversal of the influence of group Ib afferents from plantaris on activity in medial gastrocnemius muscle during locomotor activity. *J Neurophysiol* 70, 1009-1017.
- PEARSON, K. G., FOUAD, K. & MISIASZEK, J. E. (1999). Adaptive changes in motor activity associated with functional recovery following muscle denervation in walking cats. *J Neurophysiol* 82, 370-381.

- PEARSON, K. G., MISIASZEK, J. E. & HULLIGER, M. (2003). Chemical ablation of sensory afferents in the walking system of the cat abolishes the capacity for functional recovery after peripheral nerve lesions. *Exp Brain Res* **150**, 50-60.
- PHILIPPSON, M. (1905). L'autonomie et la centralisation dans le systeme nerveux des animaux. Trav.Lab.Physiol.Inst.Solvay (Bruxelles) 7, 1–208.
- PIN, T., CREST, M., EHILE, E., JACQUET, G. & GOLA, M. (1990). Plateau-generating nerve cells in Helix: morphologica and electrophysiological characteristics. *Journal of Experimental Biology* 152, 189-209.
- POPOVIC, D., TOMOVIC, R. & SCHWIRTLICH, L. (1989). Hybrid assistive system-the motor neuroprosthesis. *IEEE Transactions on Biomedical Engineering* 36, 729-737.
- POPPELE, R. E. & QUICK, D. C. (1985). Effect of intrafusal muscle mechanics on mammalian muscle spindle sensitivity. *J Neurosci* 5, 1881-1885.
- PRATT, C. A. (1995). Evidence of positive force feedback among hindlimb extensors in the intact standing cat. *J Neurophysiol* 73, 2578-2583.
- PROCHAZKA, A. (1993). Comparison of natural and artificial control of movement. *IEEE Trans. Rehab.* Eng. 1, 7-17.
- PROCHAZKA, A. (1996). Proprioceptive feedback and movement regulation. In Handbook of Physiology. Section 12. Exercise: Regulation and Integration of Multiple Systems. ed. ROWELL, L. & SHEPERD, J. T. American Physiological Society, New York.
- PROCHAZKA, A., ELEK, J. & JAVIDAN, M. (1992). Attenuation of pathological tremors by functional electrical stimulation. I: Method. Ann Biomed Eng 20, 205-224.
- PROCHAZKA, A., GILLARD, D. & BENNETT, D. J. (1997a). Implications of Positive Feedback in the Control of Movement. *J Neurophysiol* 77, 3237-3251.
- PROCHAZKA, A., GILLARD, D. & BENNETT, D. J. (1997b). Positive Force Feedback Control of Muscles. J Neurophysiol 77, 3226-3236.
- PROCHAZKA, A. & GORASSINI, M. (1998a). Ensemble firing of muscle afferents recorded during normal locomotion in cats. J Physiol (Lond) 507, 293-304.
- PROCHAZKA, A. & GORASSINI, M. (1998b). Models of ensemble firing of muscle spindle afferents recorded during normal locomotion in cats. *J Physiol (Lond)* 507, 277-291.
- PROCHAZKA, A., WESTERMAN, R. A. & ZICCONE, S. P. (1976). Discharges of single hindlimb afferents in the freely moving cat. *J Neurophysiol* 39, 1090-1104.

- PROCHAZKA, A. & YAKOVENKO, S. (2002). Locomotor control: from spring-like reactions of muscles to neural prediction. In *The somatosensory system. Deciphering the brain's own body image.* ed. NELSON, R. J., pp. 141-181. CRC Press.
- PROSKE, U. (1981). The Golgi tendon organ. Properties of the receptor and reflex action of impulses arising from tendon organs. In *MTP International Review of Physiology, Neurophysiology IV.* ed. PORTER, R. MTP University Park Press, Baltimore.
- RAASCH, C. C. & ZAJAC, F. E. (1999). Locomotor Strategy for Pedaling: Muscle Groups and Biomechanical Functions. J Neurophysiol 82, 515-525.
- RACK, P. M. & WESTBURY, D. R. (1974). The short range stiffness of active mammalian muscle and its effect on mechanical properties. *J Physiol* 240, 331-350.
- RAIBERT, M. H. (1990). Trotting, pacing and bounding by a quadruped robot. *J Biomech* 23 Suppl 1, 79-98.
- RAINETEAU, O., FOUAD, K., BAREYRE, F. M. & SCHWAB, M. E. (2002). Reorganization of descending motor tracts in the rat spinal cord. *Eur J Neurosci* 16, 1761-1771.
- RAMIREZ, J. M. & PEARSON, K. G. (1991). Octopamine induces bursting and plateau potentials in insect neurones. *Brain Res* 549, 332-337.
- RAMON Y CAJAL, S. (1928). Degeneration and regeneration of the nervous system. Oxford University Press, London.
- RAMON-CUETO, A., CORDERO, M. I., SANTOS-BENITO, F. F. & AVILA, J. (2000). Functional recovery of paraplegic rats and motor axon regeneration in their spinal cords by olfactory ensheathing glia. *Neuron* 25, 425-435.
- RAMON-CUETO, A., PLANT, G. W., AVILA, J. & BUNGE, M. B. (1998). Long-distance axonal regeneration in the transected adult rat spinal cord is promoted by olfactory ensheathing glia transplants. J Neurosci 18, 3803-3815.
- RASSIER, D. E., MACINTOSH, B. R. & HERZOG, W. (1999). Length dependence of active force production in skeletal muscle. *J Appl Physiol* 86, 1445-1457.
- RHO, M. J., CABANA, T. & DREW, T. (1997). Organization of the projections from the pericruciate cortex to the pontomedullary reticular formation of the cat: a quantitative retrograde tracing study. *J Comp Neurol* **388**, 228-249.

ROSSIGNOL, S. (1996). Visuomotor regulation of locomotion. Can J Physiol Pharmacol 74, 418-425.

- SCHNEIDER, C., LAVOIE, B. A. & CAPADAY, C. (2000). On the origin of the soleus H-reflex modulation pattern during human walking and its task-dependent differences. *J Neurophysiol* 83, 2881-2890.
- SCHNELL, L. & SCHWAB, M. E. (1990). Axonal regeneration in the rat spinal cord produced by an antibody against myelin-associated neurite growth inhibitors. *Nature* 343, 269-272.
- SCHONER, G. & KELSO, J. A. (1988). A synergetic theory of environmentally-specified and learned patterns of movement coordination. II. Component oscillator dynamics. *Biol Cybern* 58, 81-89.
- SCOTT, O. M., VRBOVA, G., HYDE, S. A. & DUBOWITZ, V. (1985). Effects of chronic low frequency electrical stimulation on normal human tibialis anterior muscle. *J Neurol Neurosurg Psychiatry* 48, 774-781.
- SECHENOV, I. M. (1863). Reflexes of the Brain, (Рефлексы головного мозга). In *I.M. Sechenov*, *Selected Works*. ed. SUBKOV, A. A., pp. 264-322. State Publishing House, Moscow.
- SEVERIN, F. V. (1970). [Role of the gamma-motor system in activation of extensor alpha-motor neurons during controlled locomotion]. *Biofizika* 15, 1096-1102.
- SEYFARTH, A., GEYER, H. & HERR, H. (2003). Swing-leg retraction: a simple control model for stable running. *J Exp Biol* 206, 2547-2555.
- SHERRINGTON, C. S. (1906). On the proprio-ceptive system, especially in its reflex aspects. *Brain* 29, 467-482.
- SHERRINGTON, C. S. (1910). Flexion-reflex of the limb, crossed extension-reflex, and reflex stepping and standing. *Journal of Physiology (London)* 40, 28-121.
- SHIK, M. L. & ORLOVSKY, G. N. (1976). Neurophysiology of locomotor automatism. *Physiol Rev* 56, 465-501.
- SHIK, M. L., SEVERIN, F. V. & ORLOVSKII, G. N. (1966). [Control of walking and running by means of electric stimulation of the midbrain]. *Biofizika* 11, 659-666.
- SINKJAER, T. (1997). Muscle, reflex and central components in the control of the ankle joint in healthy and spastic man. Acta Neurol Scand Suppl 170, 1-28.
- SINKJAER, T., ANDERSEN, J. B., LADOUCEUR, M., CHRISTENSEN, L. O. & NIELSEN, J. B. (2000). Major role for sensory feedback in soleus EMG activity in the stance phase of walking in man. *J Physiol* 523 Pt 3, 817-827.

- SINKJAER, T., ANDERSEN, J. B. & LARSEN, B. (1996). Soleus stretch reflex modulation during gait in humans. J Neurophysiol 76, 1112-1120.
- SINKJAER, T., ANDERSEN, J. B., NIELSEN, J. F. & HANSEN, H. J. (1999). Soleus long-latency stretch reflexes during walking in healthy and spastic humans. *Clin Neurophysiol* **110**, 951-959.
- SPENCER, H. (1855). Principles of Psychology. Appleton, New York.
- STEIN, R. B., MOMOSE, K. & BOBET, J. (1999). Biomechanics of human quadriceps muscles during electrical stimulation. *J Biomech* 32, 347-357.
- STEIN, R. B., PECKHAM, P. H. & POPOVIC, D. B. (1992). Neural Prostheses. In *Replacing Motor Function After Disease or Disability*. Oxford University Press, New York.
- STENNETT, R., OGINO, K., MORGAN, J. P. & BURKHOFF, D. (1996). Length-dependent activation in intact ferret hearts: study of steady-state Ca(2+)-stress-strain interrelations. Am J Physiol 270, H1940-1950.
- STEPHENS, J. A., REINKING, R. M. & STUART, D. G. (1975). Tendon organs of cat medial gastrocnemius: responses to active and passive forces as a function of muscle length. *J Neurophysiol* 38, 1217-1231.
- STEPHENS, M. J. & YANG, J. F. (1999). Loading during the stance phase of walking in humans increases the extensor EMG amplitude but does not change the duration of the step cycle. *Exp Brain Res* 124, 363-370.
- TUSZYNSKI, M. H., PETERSON, D. A., RAY, J., BAIRD, A., NAKAHARA, Y. & GAGE, F. H. (1994a). Fibroblasts genetically modified to produce nerve growth factor induce robust neuritic ingrowth after grafting to the spinal cord. *Exp Neurol* **126**, 1-14.
- TUSZYNSKI, M. H., SENUT, M. C., RAY, J. & ROBERTS, J. (1994b). Somatic gene transfer to the adult primate central nervous system: in vitro and in vivo characterization of cells genetically modified to secrete nerve growth factor. *Neurobiol Dis* 1, 67-78.
- TUSZYNSKI, M. H., WEIDNER, N., MCCORMACK, M., MILLER, I., POWELL, H. & CONNER, J. (1998). Grafts of genetically modified Schwann cells to the spinal cord: survival, axon growth, and myelination. *Cell Transplant* 7, 187-196.
- VELTINK, P. H., CHIZECK, H. J., CRAGO, P. E. & EL-BIALY, A. (1992). Nonlinear joint angle control for artificially stimulated muscle. *IEEE Trans Biomed Eng* **39**, 368-380.
- WETZEL, M. C., ATWATER, A. E., WAIT, J. V. & STUART, D. G. (1976). Kinematics of locomotion by cats with a single hindlimb deafferented. *J Neurophysiol* 39, 667-678.

- YAKOVENKO, S., GRITSENKO, V. & PROCHAZKA, A. (2004). Contribution of stretch reflexes to locomotor control: a modeling study. *Biological Cybernetics* **90**, 146 155.
- YAKOVLEFF, A., CABELGUEN, J. M., ORSAL, D., GIMENEZ Y RIBOTTA, M., RAJAOFETRA, N., DRIAN, M. J., BUSSEL, B. & PRIVAT, A. (1995). Fictive motor activities in adult chronic spinal rats transplanted with embryonic brainstem neurons. *Exp Brain Res* 106, 69-78.
- YANG, J. F., STEIN, R. B. & JAMES, K. B. (1991). Contribution of peripheral afferents to the activation of the soleus muscle during walking in humans. *Exp Brain Res* 87, 679-687.
- YANG, J. F., STEIN, R. B., JHAMANDAS, J. & GORDON, T. (1990). Motor unit numbers and contractile properties after spinal cord injury. *Ann Neurol* 28, 496-502.
- ZAJAC, F. E. (1989). Muscle and tendon: properties, models, scaling, and application to biomechanics and motor control. *Crit Rev Biomed Eng* 17, 359-411.
- ZHANG, W., POMBAL, M. A., EL MANIRA, A. & GRILLNER, S. (1996). Rostrocaudal distribution of 5-HT innervation in the lamprey spinal cord and differential effects of 5-HT on fictive locomotion. *J Comp Neurol* 374, 278-290.

CHAPTER 2

Isometric muscle length-tension curves do not predict angle-torque curves of buman wrist in continuous active movements.

Adapted from an original publication:

Gillard DM, Yakovenko S, Cameron T, and Prochazka A. Journal of Biomechanics 33: 1341-1348, 2000.

The author contributed to the development of ideas, data collection and analysis, writing and revising the manuscript.

2.1 Introduction

The force generated by active muscles depends not only on activation level but also on length and velocity. The classical length-tension curve of active muscle, first described by Blix (Blix, 1894), has an ascending limb, a plateau, a descending limb and a final ascending limb at maximal physiological lengths. The initial ascending and descending limbs are attributed to increases and decreases in overlap of actin and myosin filaments as sarcomeres lengthen (Gordon *et al.*, 1966). The final ascending limb is attributed to passive stiffness (Huijing, 1985; Zajac, 1989).

Isometric length-tension curves in different cat hindlimb muscles range from the classical curve described above to monotonic curves (Gareis *et al.*, 1992). From experiments in healthy human subjects and cadavers, Lieber et al. (Lieber *et al.*, 1994), Loren et al. (Loren *et al.*, 1996) and Lieber and Frieden (Lieber & Friden, 1998) predicted that over the physiological range of motion (ROM), wrist flexors would have monotonic isometric length-tension curves while wrist extensors would show classical curves with prominent descending limbs. The data indicated that extensor carpi radialis brevis (ECRB) might operate on the descending limb over much of its functional range. As this seemed at odds with the mechanical properties of the hand during functional electrical stimulation (Prochazka *et al.*, 1992), we wondered whether the isometric data were really as predicted by Loren et al. (Loren *et al.*, 1996) and more generally, whether isometric length tension curves generalized to continuous active movements.

Though one often hears the opinion that because isometric length-tension curves represent discrete static measurements they probably do not generalize to continuous movement, few publications address the issue in detail and the evidence is fragmentary (Geffen, 1964; Baratta *et al.*, 1993; Vance *et al.*, 1994). In biomechanics, muscle is often modeled as the product of an

isometric length-tension curve and a Hill-type force-velocity relationship (e.g. Veltink *et al.*, 1992). Such models predict that for constant-velocity stretches muscle force should decline as stretch proceeds beyond the peak of the isometric length-tension curve. A decline in force with increasing length may be interpreted as negative stiffness (Stein *et al.*, 1999), a property that is difficult to deal with in feedback control systems.

Actively contracting muscle has a high short-range stiffness when stretched through an initial $\sim 1\%$ of rest length (Joyce *et al.*, 1969). Actin-myosin cross-bridges are thought to remain attached and deform elastically within this range and become detached when stretch continues, resulting in a lower stiffness (Rack & Westbury, 1974). Consequently, when an active muscle is cyclically stretched and shortened, short-range stiffness may produce hysteresis (a divergence of length-tension curves for shortening and lengthening not attributable to force-velocity characteristics). Another nonlinear property of muscle that could produce hysteresis is the so-called force-enhancement effect during stretch and force deficit during shortening (Julian & Morgan, 1979a; Edman *et al.*, 1982; Sugi & Tsuchiya, 1988; Morgan, 1990). This and a related phenomenon, "creep", have been attributed to the build-up of sarcomere non-uniformities at long muscle lengths (Gordon *et al.*, 1966; Julian & Morgan, 1979a; Edman & Reggiani, 1984; Morgan, 1990).

In this study we tested two hypotheses: 1) static isometric length-tension curves of active wrist muscles apply to continuous movements; 2) hysteresis in length-tension curves of active wrist muscles is maximal at maximal muscle lengths.

2.2 Methods

Angle-torque curves of wrist flexors or extensors were measured in 4 subjects, 2 males and 2 females, aged 21-52 in steady contractions produced either by electrical stimulation or constant

55

voluntary contraction. No subjects had any history of wrist trauma or musculoskeletal disease. All subjects gave written consent in accordance with the requirements of the University of Alberta Human Ethics committee and the Declaration of Helsinki.

The subject was seated in a comfortable chair with his/her elbow flexed to 90° and shoulder abducted by 30-40°. The elbow was held in place with a rigid molding of thermoplastic material (Sansplint) and the forearm was stabilized just distal to the elbow and just proximal to the wrist with two pairs of padded restraints (Fig. 2.1). The hand was formed into a fist and secured in a form-fitting Sansplint cast. The cast was attached to a rigid, circular arc which acted as a guide for a flexible steel cable connecting one end of the arc to a force gauge and a rotary servo motor. This allowed force to be applied at a constant distance from the wrist pivot point. Other cables suspended from the ceiling supported the cast in the horizontal plane. A pointer centered at the mid-point of the arc was positioned over a protractor for calibrating angle measurements obtained from an angular displacement transducer attached to the servo motor. The servo (stiffness 19.3 N/mm) was controlled using Simulink and Real-Time Workshop Toolbox (The Mathworks, Inc, Natick, MA) and a digital microprocessor interface (dSPACE Inc., Northville, MI).

Two fatigue tests were performed in each subject prior to the main experiment. With the wrist in neutral position, 10% MVC was maintained for 120s by 1) electrical stimulation or 2) voluntary effort. In stimulation trials, torque typically declined presumably because of fatigue in fast-fatiguable muscle fibres, reaching a plateau after about 20s. In the main experiments, 20s of stimulation was applied between trials to bring the muscles to this plateau state. The same fatigue tests were performed immediately after the experiments to test for stationarity.



Figure 2.1. Experimental set-up. Subject's elbow and forearm were stabilized with an elbow cast and padded wrist restraints. The hand, bound into a fist, was secured inside a form-fitting cast attached to a circular arc suspended from the ceiling. The arc guided a cable connected to a rotary servo motor, allowing force to be applied at a constant distance from the wrist joint, i.e. force was proportional to torque. The torque required to move the wrist through different angles was measured while muscles were activated at a constant level, either voluntarily with EMG feedback, or by trains of electrical stimuli applied to the muscle nerves proximal to the elbow.

The servo moved the hand at a speed of 10° s⁻¹. To avoid fatigue in full range of motion (ROM: +50°) trials with active muscle contraction, data were obtained separately for 1) muscle shortening over the whole 50° range, 2) first 25° of lengthening; 3) second 25° of lengthening. All trials were repeated 3 times. Sets of six +10° movements centered around -40°, -20°, 0°, +20°, +40° (flexion negative, extension positive) were recorded in random order. Next, torque in isometric contractions from rest at each of the above angles was measured. Finally, passive angle-torque characteristics with muscles relaxed were recorded in +50° and +10° movements.

In voluntary contraction experiments, EMG recordings were obtained from pairs of selfadhesive electrodes (Jason ElectroTrace, ET001, Huntington Beach, CA) 1cm apart over ECRB 6cm distal to the lateral epicondyle, and flexor carpi radialis (FCR) 6cm distal to the medial epicondyle. A reference electrode was placed over the ulnar olecranon process. EMG signals were high-passfiltered (30Hz), amplified, full-wave-rectified, low-pass-filtered (1Hz) and displayed on an oscilloscope. The subject flexed or extended isometrically from the neutral wrist position to match the force trace to a target level corresponding to 10%MVC, using wrist muscles only. EMG crosstalk from finger muscles was estimated in separate strong contractions to be <10% of EMG from wrist muscles. Antagonist EMGs were monitored and played through a loudspeaker to the subject, whose job was to minimize co-contraction. Muscle activation commenced 2s before the onset of recording to give the subject time to stabilize the EMG.


Figure 2.2. Mean angle-torque data for constant voluntary activation of wrist flexors. (A) Torque was measured in three contraction conditions: (1) isometric (dotted lines); (2) $\pm 10^{\circ}$ triangular displacements centered at five different angles (small loops); (3) $\pm 50^{\circ}$ triangular displacements over full range of motion (full ROM) of wrist joint (large loop). Narrow loops at bottom of figure show angle-torque profiles with muscles completely relaxed. Note that all torque values in this and subsequent figures were normalized to mean torque of $\pm 10^{\circ}$ angle-torque loops at 0° wrist angle. (B) Active torque derived by subtracting passive torque values from corresponding total torque values in (A). Vertical bars: standard errors of means. Note large width of all ROM loops, even though the movements were applied very slowly to minimize velocity-dependent torque components. This suggests hysteresis rather than viscous properties. Note the positive slopes of the $\pm 10^{\circ}$ ROM loops too, even in the range 0-30°, over which the isometric angle-torque curve descended slightly.

The same subjects participated in the experiments involving electrical stimulation. Selfadhesive electrodes (ConMed Versa-stim, 45x45 mm) were placed over the median nerve 8-10 cm proximal to the humeral medial epicondyle to stimulate flexors and over the radial nerve 8-10 cm proximal to the lateral epicondyle to stimulate extensors. Stimulation at motor points proximal to the origins of the muscles activated, avoided variations in stimulus efficacy as muscles changed length and shape. An indifferent electrode (45x90 mm) was placed on the skin just proximal to the wrist crease on the anterior surface of the forearm. Stimulation comprised trains of currentcontrolled biphasic pulses (33/s, 100ms primary pulse width). Subjects were instructed to remain relaxed and allow the stimulation to control their hand. Practice was required not to intervene. We checked for intervention periodically by turning stimulation off unexpectedly (Prochazka *et al.*, 1997).

Since electrical stimulation does not recruit motor units in physiological order (slow, fatigue-resistant to fast-fatiguable), stimulation was turned on between each set of cyclical movement trials for 20s to cause fatigue of fast-fatiguable muscle fibers. Stimulation was then adjusted to achieve 10% MVC and commenced 2s before recording started to allow for equilibration. The same number and types of movement trials were performed as in voluntary contraction experiments.

Signals from the force transducer, servo input, displacement sensor and EMGs were digitized at 100/s and averaged using a Cambridge Electronic Design (CED) 1401 laboratory interface and personal computer using SIGAVG (version 5.42), Matlab (version 4.2c) and SigmaPlot (version 4.0) software. For each subject torques were normalized to the mean torque of the $+10^{\circ}$ angle-torque loops at 0° wrist angle.

EXTENSOR ANGLE-TORQUE CURVES Constant voluntary EMG activation of extensors



Figure 2.3. Mean angle-torque data for constant (10% maximal), voluntary activation of wrist extensors. Trials and data presentation identical to those in Figure 2.2. Compared to the flexor data, total isometric torque declined more noticeably as the wrist was flexed beyond neutral (i.e. as the extensors lengthened). There was also a decline in torque over this angular range in full ROM trials. These declines were particularly marked in Figure 2.2B, after the passive angle-torque data had been subtracted. In spite of these descending limbs, the slopes of smaller $\pm 10^{\circ}$ loops remained positive.

2.3 Results

Full ROM trials involving voluntary control of flexor EMG produced monotonic angletorque curves for active lengthening characterized by an initial steep slope, followed by a less steep slope (Fig. 2.2A). At the onset of shortening there was an initial steep drop in torque followed by a less steep decline. $\pm 10^{\circ}$ trials also showed monotonic angle-torque curves, but notice the greater mean slopes, i.e. larger stiffnesses than would be predicted either from the isometric or full ROM curves. The isometric data show a classical length-tension profile: an ascending limb, a descending limb and a small increase at the longest muscle length. All angle-torque movement trials with voluntary flexor contractions showed prominent hysteresis, torque being larger during lengthening of the active muscles than during shortening. The passive angle-torque curves also exhibited some hysteresis, but far less than the active curves. Passive hysteresis is attributable to connective tissue properties (Given *et al.*, 1995).

In our first subject full ROM cycles were applied in one continuous sweep, but fatigue near the end of each trial, evidenced by mismatches between start and end coordinates of angle-torque curves, led us to split full ROM trials into three portions in the other subjects (see Methods). Good correspondence between start and end coordinates in most trials indicated that fatigue was thereby minimized. Figure 2.2B was obtained by subtracting passive angle-torque curves in Figure 2.2A from corresponding total torque curves. This standard procedure provides an estimate of the active angle-torque profile and emphasizes the descending limb in isometric and full ROM trials.

In contrast to the flexor results, in full ROM trials of wrist extensors (Fig. 2.3), the total angle-torque curve during muscle lengthening showed a clear transition from an ascending limb to a descending limb with a final small increase in torque at longest muscle length. Thus, although the

full ROM plot had a zone of negative stiffness during muscle lengthening, stiffness was always positive in the $\pm 10^{\circ}$ trials. A prominent hysteresis was again evident in both the full ROM and $\pm 10^{\circ}$ trials. The active angle-torque plots (Fig. 2.3B) showed similar properties to those in Figure 2.2B, though with more prominent descending limbs in isometric and full ROM curves.

Electrical stimulation of median nerve proximal to the elbow elicited reliable contractions of the wrist flexors in all subjects. Unfortunately it was much more difficult to find equivalent motor points over radial nerve proximal to the elbow. Seven people were tested but only one had an accessible and selective wrist extensor motor point. We felt it essential that the motor point be proximal to any part of the muscles being stimulated, so only this one subject was tested for this part of the experiment.

The flexor stimulation angle-torque curves (Fig. 2.4) showed the same general characteristics as those in voluntary trials (Fig. 2.2). This shows that whether the muscles are activated at a constant level by electrical stimulation or by voluntary effort, the same general conclusions may be drawn.

Since only one subject was tested with extensor stimulation, the same experiment was done on two separate occasions to obtain enough data for averaging. Again the results were comparable to the voluntary trials (Fig. 2.3), though the full ROM did not exhibit the descending limb of the angle-torque curve as clearly and maximal torque occurred at a more flexed angle than in isometric trials. As in Figure 2.3, the $\pm 10^{\circ}$ trials showed little evidence of negative stiffness.



Figure 2.4. Mean angle-torque data for constant (10% maximal), electrical activation of wrist flexors. Stimulation was applied to the muscle nerve above the elbow to minimize the effect of changes in muscle length and shape on the position of the motor point. Trials and data presentation as in Figure 2.2. The properties of the angle-torque curves were basically the same as those in Figure 2.2, which indicates that the muscles were activated similarly in both situations.



Figure 2.5. Mean angle-torque data for constant (10% maximal), electrical activation of wrist extensors Trials and data presentation as in Figure 2.3. The properties of the angle-torque curves were similar to those in Figure 2.3, indicating that the muscles were activated similarly in both situations. In this case the data could only be obtained in one subject, because only this subject had an accessible motor point for extensor stimulation proximal to the elbow.

In order to explore how much of the separation between lengthening and shortening curves in movement trials was due to static hysteresis as opposed to viscosity, we performed a series of $\pm 10^{\circ}$ trials at different velocities in one subject. Figure 2.6 shows the maximal difference in torque (δ T) within the angle-torque loops of averaged cycles plotted against angular velocity. δ T is the sum of the components of force due to viscosity and hysteresis, so there is a steady increase in δ T with velocity. The regression line, which intercepts the y-axis above zero provides a measure of the separation of the angle-torque loops that can be attributed to static hysteresis alone.

2.4 Discussion

Our study was originally motivated by Loren et al. (Loren *et al.*, 1996) who predicted that isometric angle-torque curves of active wrist extensors would differ from those of flexors. The study developed into a test of the more general hypothesis that isometric length-tension curves of active wrist muscles are also valid in continuous movements. In agreement with Lieber et al. (Lieber *et al.*, 1994; Lieber & Friden, 1998) we found that in isometric and full ROM trials, active wrist flexor torque increased monotonically with extension whereas active extensor torque followed the classical isometric length-tension curve with an ascending limb, a plateau region, a descending limb and a final ascending limb.

The striking difference was that when $\pm 10^{\circ}$ cyclical movements were imposed, stiffness during extensor activation remained positive at all wrist angles, including the range in which isometric and full ROM trials showed descending limbs. Stein et al. (Stein *et al.*, 1999) reported similar observations in electrically-stimulated quadriceps muscles in humans but not in voluntary contractions.



Figure 2.6. Effect of muscle force-velocity properties on the width of the $\pm 10^{\circ}$ ROM loops. The maximal difference in torque (δT) within the loops is plotted against angular velocity. δT is the sum of the components of force due to the force-velocity relationship and hysteresis. The force-velocity contribution to δT is reflected in the linear increase in δT with velocity. The regression line intercepts the *y*-axis above zero. This provides a measure of the separation of static hysteresis due to muscle short-range stiffness properties.

We measured angular displacement and torque about the wrist rather than muscle length and force. If angle were proportional to muscle displacement and torque were proportional to force, the angle-torque profiles would be identical to the muscle length-tension curves. However, wrist torque is the sum of torques of several muscles, each of which has a slightly different moment arm. The larger the moment arm, the greater the muscle length change for a given change in joint angle. This means that each muscle might traverse a different region of its length-tension curve. Furthermore, torque is only proportional to muscle force if the moment arms remain constant. Horii et al. (Horii et al., 1993) and Loren et al. (Loren et al., 1996) estimated the moment arms of the main wrist muscles: FCR, flexor carpi ulnaris (FCU), extensor carpi ulnaris (ECU), ECRB and extensor carpi radialis longus (ECRL) from the slopes of the relationship between wrist angle and muscle length in cadavers. According to Loren et al, the moment arm of FCU (generating 50-60% of flexor torque) decreases by 35% as FCU lengthens from full wrist flexion to extension, but Horii et al. concluded that it increases by $\sim 15\%$. The corresponding values for FCR moment arms were a 12% decrease (Loren et al., 1996) and a 20% increase (Horii et al., 1993). From Loren et al., the reduction in flexor stiffness we observed as the wrist extended (Fig. 2.2) could therefore be due to declining flexor moment arms (not just muscle length-tension properties), but from Horii et al. we could conclude that active muscle force declined with increasing stretch.

Regarding extensors, according to Loren et al. ECRB (~60% of extensor torque) moment arm increases slightly (~10%) as the wrist is moved from full extension, but then declines by ~50% to full flexion. Horii et al. estimated a linear decline of ~13%. For ECRL (20-25% of extensor torque), Loren et al. estimated a large decline in moment arm, while Horii et al. found only a small (~6%) decline restricted to the last 30° of flexion. ECU (20-25% of extensor torque) moment arm showed a bell-shaped profile peaking in the mid-range according to Loren et al., while Horii et al. maintained that it changed little from extension to mid-point, then declined 75% to full flexion.

Though these results vary in detail, the common factor is a general decline in extensor moment arm with increasing flexion, particularly beyond the mid-point of wrist angle. This could partly explain the descending limbs of isometric and full ROM angle-torque profiles in our data (Fig. 2.3). But the important point remains that when $\pm 10^{\circ}$ movements were imposed within the angular range of the descending limb of the isometric and full ROM curves, mean wrist stiffness was positive rather than negative, in spite of possible declines in moment arms. The prediction from the isometric angle-torque relationship of a region of negative stiffness is therefore clearly invalid for small dynamic movements. Given that in activities of daily life, the wrist normally operates over the range 10° flexion to 35° extension (Brumfield & Champoux, 1984), isometric curves, with their descending limbs, do not represent normal muscle behavior.

The marked difference in torque during lengthening and shortening of active muscles, which resulted in the loops seen in all the continuous angle-torque plots, took us by surprise, because we had chosen very low angular velocities to minimize viscous forces. When we examined this further (Fig. 2.6), it became clear that the loops resulted more from static hysteresis than viscosity. In most of the $+10^{\circ}$ loops the hysteresis was attributable to abrupt changes in torque at the onset of muscle lengthening and shortening which suggests short-range stiffness effects (Rack & Westbury, 1974). However the maintained separation of ascending and descending portions of the loops throughout the cycles is not explained by short-range stiffness. Lombardi and Piazzesi (Lombardi & Piazzesi, 1990) adduced evidence that cross-bridge cycling rates in frog muscle fibers are higher during lengthening than during shortening contractions and other work suggests that the average force produced for each attached cross-bridge is higher in lengthening than in isometric

69

contractions (Herzog, 1998). These mechanisms may explain the maintained separation of ascending and descending limbs of angle-torque curves. According to Nardone et al. (Nardone *et al.*, 1989), fast and slow motor units of human calf muscles are recruited selectively in voluntary lengthening and shortening contractions respectively. This is another possible cause of hysteresis, though we would then have expected to see larger differences between voluntary and electrical stimulation data.

Stein et al. (Stein et al., 1999) applied velocities of 24°/s to the knee and so the component of viscous stiffness was probably quite large (see our Fig. 2.6). Nonetheless, some static hysteresis was inferred. Because this reached a maximum at maximal muscle length, sarcomere nonuniformities were suspected, the notion being that some sarcomeres "pop" to a length beyond which cross-bridges can form and in-series sarcomeres shorten to compensate (Julian & Morgan, 1979b). When muscle is allowed to shorten after a large lengthening contraction, its ability to sustain force is therefore reduced (Gordon et al., 1966; Edman & Reggiani, 1984; Morgan, 1990). In our study, the fact that the $\pm 10^{\circ}$ loops were actually thinner at long than at intermediate lengths detracts from this explanation. Furthermore, our hysteresis loops were symmetrical: increases in torque at the onset of lengthening were similar to decreases in torque at the onset of shortening. Sarcomere nonuniformities would operate chiefly at one end of the loop: at the onset of shortening after a large lengthening contraction. Stein et al. (Stein et al., 1999) reported that the hysteresis was not present in voluntary activation. This was not the case in our study: if anything, hysteresis was more marked in voluntary trials than in stimulation trials. Gielen et al. (Gielen et al., 1984) measured the elastic and viscous properties of reflexly active wrist muscles in humans. Subjects were instructed "not to intervene" when force or displacement applied to the hand during voluntary contractions against a steady load was altered. There was a prominent static hysteresis which the authors attributed to short-range stiffness rather than reflex action. Our results support this interpretation.

In summary, although we confirmed the claim of Lieber et al. (Lieber *et al.*, 1994) and Loren et al. (Loren *et al.*, 1996) that active wrist extensors but not flexors showed descending limbs in their isometric angle-torque curves, in small continuous movements, stiffness remained positive throughout. We conclude that isometric length-tension curves of active wrist muscles are not representative of continuous movements. Hysteresis in angle-torque curves was attributed to shortrange stiffness and stretch-potentiation rather than to sarcomere nonuniformities.

2.5 References

- BARATTA, R. V., SOLOMONOW, M., BEST, R. & D'AMBROSIA, R. (1993). Isotonic length/force models of nine different skeletal muscles. *Med Biol Eng Comput* 31, 449-458.
- BLIX, M. (1894). Die Lange und Spannung des Muskels. Skand. Arch. Physiol. 5, 149-206.
- BRUMFIELD, R. H. & CHAMPOUX, J. A. (1984). A biomechanical study of normal functional wrist motion. *Clin Orthop*, 23-25.
- EDMAN, K. A., ELZINGA, G. & NOBLE, M. I. (1982). Residual force enhancement after stretch of contracting frog single muscle fibers. *J Gen Physiol* 80, 769-784.
- EDMAN, K. A. & REGGIANI, C. (1984). Redistribution of sarcomere length during isometric contraction of frog muscle fibres and its relation to tension creep. *J Physiol* **351**, 169-198.
- GAREIS, H., SOLOMONOW, M., BARATTA, R., BEST, R. & D'AMBROSIA, R. (1992). The isometric length-force models of nine different skeletal muscles. *J Biomech* 25, 903-916.
- GEFFEN, L. B. (1964). Optimum length for contraction of rat circulated limb muscles. Arch Int Physiol Biochim 72, 825-834.
- GIELEN, C. C., HOUK, J. C., MARCUS, S. L. & MILLER, L. E. (1984). Viscoelastic properties of the wrist motor servo in man. Ann Biomed Eng 12, 599-620.

- GIVEN, J. D., DEWALD, J. P. & RYMER, W. Z. (1995). Joint dependent passive stiffness in paretic and contralateral limbs of spastic patients with hemiparetic stroke. *J Neurol Neurosurg Psychiatry* 59, 271-279.
- GORDON, A. M., HUXLEY, A. F. & JULIAN, F. J. (1966). The variation in isometric tension with sarcomere length in vertebrate muscle fibres. *J Physiol* 184, 170-192.
- HERZOG, W. (1998). History dependence of force production in skeletal muscle: a proposal for mechanisms. *J Electromyogr Kinesiol* **8**, 111-117.
- HORII, E., AN, K. N. & LINSCHEID, R. L. (1993). Excursion of prime wrist tendons. J Hand Surg [Am] 18, 83-90.
- HUJJING, P. A. (1985). Architecture of the human gastrocnemius muscle and some functional consequences. Acta Anat (Basel) 123, 101-107.
- JOYCE, G. C., RACK, P. M. & WESTBURY, D. R. (1969). The mechanical properties of cat soleus muscle during controlled lengthening and shortening movements. *J Physiol* 204, 461-474.
- JULIAN, F. J. & MORGAN, D. L. (1979a). The effect on tension of non-uniform distribution of length changes applied to frog muscle fibres. *J Physiol* 293, 379-392.
- JULIAN, F. J. & MORGAN, D. L. (1979b). Intersarcomere dynamics during fixed-end tetanic contractions of frog muscle fibres. *J Physiol* 293, 365-378.
- LIEBER, R. L. & FRIDEN, J. (1998). Musculoskeletal balance of the human wrist elucidated using intraoperative laser diffraction. *J Electromyogr Kinesiol* 8, 93-100.
- LIEBER, R. L., LOREN, G. J. & FRIDEN, J. (1994). In vivo measurement of human wrist extensor muscle sarcomere length changes. *J Neurophysiol* 71, 874-881.
- LOMBARDI, V. & PIAZZESI, G. (1990). The contractile response during steady lengthening of stimulated frog muscle fibres. *J Physiol* **431**, 141-171.
- LOREN, G. J., SHOEMAKER, S. D., BURKHOLDER, T. J., JACOBSON, M. D., FRIDEN, J. & LIEBER, R. L. (1996). Human wrist motors: biomechanical design and application to tendon transfers. *J Biomech* 29, 331-342.
- MORGAN, D. L. (1990). New insights into the behavior of muscle during active lengthening. *Biophys J* 57, 209-221.
- NARDONE, A., ROMANO, C. & SCHIEPPATI, M. (1989). Selective recruitment of high-threshold human motor units during voluntary isotonic lengthening of active muscles. *J Physiol* 409, 451-471.

- PROCHAZKA, A., ELEK, J. & JAVIDAN, M. (1992). Attenuation of pathological tremors by functional electrical stimulation. I: Method. Ann Biomed Eng 20, 205-224.
- PROCHAZKA, A., GILLARD, D. & BENNETT, D. J. (1997). Positive force feedback control of muscles. J Neurophysiol 77, 3226-3236.
- RACK, P. M. & WESTBURY, D. R. (1974). The short range stiffness of active mammalian muscle and its effect on mechanical properties. *J Physiol* 240, 331-350.
- STEIN, R. B., MOMOSE, K. & BOBET, J. (1999). Biomechanics of human quadriceps muscles during electrical stimulation. *J Biomech* 32, 347-357.
- SUGI, H. & TSUCHIYA, T. (1988). Stiffness changes during enhancement and deficit of isometric force by slow length changes in frog skeletal muscle fibres. *J Physiol* 407, 215-229.
- VANCE, T. L., SOLOMONOW, M., BARATTA, R., ZEMBO, M. & D'AMBROSIA, R. D. (1994). Comparison of isometric and load moving length-tension models of two bicompartmental muscles. *IEEE Trans Biomed Eng* 41, 771-781.
- VELTINK, P. H., CHIZECK, H. J., CRAGO, P. E. & EL-BIALY, A. (1992). Nonlinear joint angle control for artificially stimulated muscle. *IEEE Trans Biomed Eng* **39**, 368-380.
- ZAJAC, F. E. (1989). Muscle and tendon: properties, models, scaling, and application to biomechanics and motor control. *Crit Rev Biomed Eng* 17, 359-411.

CHAPTER 3

Contribution of stretch reflexes to locomotor control: a modeling study.

Adapted from an original publication:

Yakovenko S, Gritsenko V, and Prochazka A. Biological Cybernetics 90: 146 - 155, 2004.

3.1 Introduction

At an international symposium on Movement and Sensation held in Cairns, Australia in 2001 we made the provocative suggestion that stretch reflexes do not contribute substantially to load compensation in mammalian locomotion (Prochazka *et al.*, 2002). We made this suggestion on the basis of the relatively small size and long latency of electromyogram (EMG) responses that occur after ground contact in the stance phase of the locomotor step cycle. Some initial biomechanical modeling we had done indicated that these responses may only have a modest effect on the kinematics of quadrupedal gait.

Classical studies demonstrated many years ago the ability of the spinal cord to produce the basic locomotor rhythm in the absence of sensory feedback (Brown, 1911). Brown coined the term "intrinsic factor" to describe the underlying neural mechanism and more recently Grillner renamed this the "central pattern generator" (CPG) (Grillner & Zangger, 1974). The fact that under some circumstances rudimentary weight-bearing locomotion can occur in the absence of sensory input indicates that the biomechanical properties of the limbs provide some flexibility in load compensation (Pearson *et al.*, 2003). However, studies on animal and human subjects have also shown that after sensory loss gait is far less coordinated and less able to adapt to changes in terrain and body posture (Bickel, 1897; Lajoie *et al.*, 1996; Bloem *et al.*, 2002). Thus, two main roles are usually attributed to sensory feedback: it provides control of the stiffness of individual muscles and it allows higher-level control of balance, stability and coordination.

Stretch reflexes associated with locomotion, on which we will concentrate in the following discussion, have been extensively studied with a variety of physiological techniques for over a century. Most of the studies have dwelt on the electrical responses of muscles to electrically-evoked sensory inputs or to imposed muscle stretching and shortening. The amplitude of sensory-evoked 75

EMG responses is modulated throughout the step cycle, (Akazawa *et al.*, 1982; Capaday & Stein, 1986; Dietz *et al.*, 1990; Stein *et al.*, 2000) so that contribution of stretch reflexes has been posited to be highest when the receptor-bearing muscle is active. Estimates of the relative contribution of stretch reflexes to overall muscle EMG during locomotion are in the range 25% -35% (Yang *et al.*, 1991; Bennett *et al.*, 1996; Stein *et al.*, 2000). But if the sensory input due to ground contact is removed at the onset of the stance phase of cat locomotion ("foot-in-hole" experiments), changes in EMG appear surprisingly late (30-40ms) (Gorassini *et al.*, 1994). In another set of experiments, stretch of the ankle extensors after ground contact was artificially exaggerated (Gritsenko *et al.*, 2001). This confirmed the timing of the stretch reflexes and further emphasized the modest size of the reflex components in unperturbed steps.

So what is the functional role of the stretch reflexes during locomotion? After recovery, deafferented animals show little change in yield of the limb during the stance phase of level overground locomotion, though in more demanding tasks abnormally large yielding can occur (Abelew *et al.*, 2000). However, in experiments of this type it could be that compensatory changes in central drive learnt over a period of time may replace the missing stretch reflex components and thereby mask the role of these reflexes under normal conditions. The ideal experiment would be to study locomotor performance when sensory input is suddenly abolished. It is hard to see how this manipulation could be done in real animals, but it is certainly possible in biomechanical simulations. We have used these in this study to test the following two hypotheses. 1) Stretch reflexes are too weak and too delayed to contribute significantly to weight-bearing. 2) The important contributions of sensory input involve state-dependent processing.



Figure 3.1. Structure of neuromuscular locomotor model. A musculo-skeletal model with segment lengths and weights (indicated next to segment centers of mass) based loosely on the hindlimbs of cat. Muscle groups and their origins and insertions are represented by the following musculotendon actuators HF (hip flexors), HE (hip extensors), KF-HE (bifunctional muscle with knee flexor and hip extensor function), KE (knee extensors), AF (ankle flexors) and AE (ankle extensors). Shoulder joint is constrained to 2 degrees of freedom, horizontal translation and rotation.

3.2 Methods

3.2.1 Structure of the locomotor model

We developed a planar locomotor model of two hind limbs attached to a horizontal torso supported at the front by a frictionless wheel (Fig. 3.1). Each hindlimb comprised 4 rigid-body segments (thigh, shank, foot and toes) which were driven by 6 musculo-tendon actuators. All joints were modeled as frictionless revolutes. The model was developed using Matlab version 6.1 software (The MathWorks, USA) coupled to Working Model 2D version 5 software (Knowledge Revolution, USA). The foot interactions with the ground were modeled with the use of Working Model software. The following parameters were chosen to minimize slipping: the coefficient of restitution was set to 0.1 and the coefficient of Coulomb friction to 10. The results were computed using the Kutta-Merson numerical integration method with 0.1 msec fixed steps and 1/100 mm error tolerance.

In the model to be presented, the dimensions, masses of the segments and moment arms of the actuators were chosen approximately to mimic those of a cat. All simulations were started just prior to foot contact, with initial velocity of the trunk and leading leg segments set to 0.7m/s. We intentionally did not strive for a rigorously accurate model as we were seeking conclusions that would generalize across a large range of parametric variation (see Discussion). Internal properties of the individual actuators were modeled by Hill-type force-velocity (Hill, 1938) and linear forcelength (Gillard *et al.*, 2000) relationships. Muscle force was calculated as

A muscle properties



Figure 3.2. a Active and passive force-length-velocity relationships used to model muscle properties of actuators. b EMG profiles used as CPG outputs to the corresponding musculotendon actuators. c Schematic of the organization of control system in our model.

$$F = F_{\max} \cdot f_v \cdot f_l \cdot a_t + F_{\max}^{passive} \cdot f_l^{passive}$$
 Eq. 3.1

where *F* is muscle force, F_{max} is the maximum isometric force, f_v and f_l are the force-velocity and the force-length relations respectively, $F_{max}^{passive}$ is the maximum passive force, $f_l^{passive}$ is the normalized passive force-length relation (Fig. 3.2a) and a_l is muscle activation, whose dynamics was described by He-Zajac-Levine excitation-contraction coupling equation (He *et al.*, 1991):

$$\dot{a}_{t} + \frac{1}{\tau_{act}} \cdot \left(\frac{\tau_{act}}{\tau_{deact}} + \left[1 - \frac{\tau_{act}}{\tau_{deact}}\right] \cdot u_{t}\right) \cdot a_{t} = \frac{1}{\tau_{act}} \cdot u_{t}$$
 Eq. 3.2

where τ_{act} , τ_{deact} are activation and deactivation time constants (20 msec and 40 msec, respectively) and u_t is motoneuron excitation described by Eq. 3.6.

Electromyographic (EMG) profiles of the simulated muscle groups during slow level walking (Fig. 3.2b) served as CPG outputs to the musculotendon actuators. Swing and stance durations were set to 30% and 70% of the step cycle period respectively, which is an appropriate phase relationship for slow gait with a cycle period of 600ms (Goslow *et al.*, 1973; Halbertsma, 1983). The activation profiles were based on a large number of locomotor studies and are described in Chapter 4.

3.2.2 Parametric testing

A direct dynamic approach was applied to compute kinematics and to study the contribution of different types of sensory feedback to motor control of the musculoskeletal system. Instead of using

dynamic optimization methods with set performance goals for certain types of locomotion, we chose to study performance of the model in parametric space, constructed from combinations of stiffnesses of the musculotendon actuators (Fmax in Eq. 3.1). A random search method was used to determine "stable" regions of parametric space, defined as the ability of the model to locomote for at least 12 seconds (equivalent to about 20 steps). Principal component analysis (PCA) was further performed on the "stable" sets. To quantify locomotor stability, we computed the moment of inertia for the "stable" volume defined by the sets of force parameters in Fig. 3.4 using the following formula:

$$I = \sum_{i}^{n} \frac{r_i^2}{n}$$
 Eq. 3.3

where r_i is the vector magnitude defining distance from the center of the "stable" volume to each individual parameter set and n is the total number of parameter sets (100 "stable" simulations). Locomotor performance was analyzed in the plane formed by the two principal component vectors that accounted for most of the variability in the data.

The analysis was repeated in the model where the length of the ankle was increased to match the relative segment lengths of a horse. The reason for repeating the analysis in the "horse" model was to test whether our basic conclusions held in the face of large changes in limb geometry (Fig. 3.1). The length of the ankle relative to the femur and tibia in the horse is double that in most other mammals.

3.2.3 Stretch reflex model

The stretch reflex model comprised simulated feedback from spindle Ia and tendon organ Ib afferents onto homonymous motor pools. Heterogeneous excitation and reciprocal inhibition components of afferent feedback, described by Lundberg, (Engberg & Lundberg, 1969) were neglected for the sake of simplicity. The contributions of Ia and Ib afferents to the output of homonymous motoneuron pools were represented by the following formulae derived from the literature (Prochazka, 1999).

Ia model:

$$f_{la}(l,v) = K_{la} \cdot (21 \cdot v^{0.5} + 200 \cdot l + 60)$$
 Eq. 3.4

Ib model:

$$f_{Ib}(s,F) = K_{Ib} \cdot \frac{(s+0.15) \cdot (s+1.5) \cdot (s+16)}{(s+0.2) \cdot (s+2) \cdot (s+37)} \cdot F$$
 Eq. 3.5

where f_{la} is a time function of the Ia afferent firing rate response to changes of muscle length and velocity, *l* is the muscle length in rest length units, *v* is the muscle velocity expressed in rest length per second, f_{lb} is the tendon organ response in the frequency domain, *s* is a frequency domain operator and K_{la} and K_{lb} are gain coefficients. The middle of the range of motion of each musculotendon actuator was chosen as its rest length. Note that the above equations are in the time and frequency domains respectively. Though equation 3.3 can be written as a differential equation in the time domain, its filtering properties are then very difficult to infer. Inspection of equation 3.3 indicates that tendon organs act as high pass filters, with 20 dB/decade increases in transmission occurring between the following pairs of frequencies: 0.15-0.2, 1.5-2, 16-37 rad/s (i.e. ~0.02-0.03, 82 0.2-0.3 and 2.6-5.9 Hz). Matlab Simulink allows a mixture of time and frequency domain transfer functions within the same model.

The Ia and Ib reflex feedback contributed to the muscle contractile force with a latency of 35ms (Gorassini *et al.*, 1994; Gritsenko *et al.*, 2001). This feedback was active only when the CPG EMG profile for the receptor-bearing muscle was non-zero to satisfy the known modulation of the stretch reflex within the step cycle (Capaday & Stein, 1986). The gain coefficients K_{ia} and K_{ib} were adjusted so that the Ia and Ib signals each added a mean of 15% to the CPG EMG component throughout a full locomotor cycle. Thus, together both signals contributed 30% to the CPG EMG profiles, which is in accordance with the suggested reflex contribution to the EMG during walking (Bennett *et al.*, 1996; Stein *et al.*, 2000). When the stretch reflex component was present, the excitation function in Eq. 3.2 was of the form:

$$u_t = CPG_t + f_{la} + f_{lb}$$
 Eq. 3.6

3.2.4 Finite-state rules

The following IF-THEN rules were used to model higher level control of phase–switching in the step cycle (Granat *et al.*, 1993; Prochazka, 1993):

- 1. Stance to swing transition: IF stance AND ipsilateral hip is extended AND ipsilateral leg is unloaded THEN swing;
- 2. Swing to stance transition: IF swing AND ipsilateral hip is flexed AND ipsilateral knee is extended THEN stance.



Figure 3.3. Contribution of stretch reflexes during locomotion. Two simulation results are shown in the *left* and *right* panels. **a**—**c** Centrally-generated levels of muscle activation were not quite high enough to support stable locomotion. Stretch reflex contributions comprising 30% of the total activation (*black* portions of the activation profiles) were initially present and then suddenly removed. **d**—**f** Centrally-generated levels of muscle activation were sufficient to support stable locomotion; stretch reflex contributions (*black* portions of profiles) were initially absent and then suddenly added. Stick figures of **c** indicate that the stretch reflexes are crucial to maintain stable locomotion when CPG activation alone is insufficient to support load bearing. However, in a trial with sufficient CPG activity (**f**), the addition of stretch reflexes only had the effect of slightly increasing gait velocity, which eventually led to collapse (see supplementary material).

The thresholds for the firing of these rules were determined on a trial-and-error basis. The schematic of all control systems implemented in the model are shown in Fig. 3.2c.

3.3 Results

3.3.1 Simulation of "deafferentation" experiment

Figure 3.3 illustrates two examples of the type of experiment mentioned in the Introduction, where a sudden change in sensory feedback occurs during locomotion. In the left panels (Fig. 3.3a-c) sensory feedback was suddenly withdrawn, whereas in the right panels (Fig. 3.3d—f) sensory feedback was suddenly added. In both cases the sensory inputs contributed about 30% to the overall muscle activation profiles as described above. In the example on the left the amplitudes of the centrally generated components were deliberately chosen to produce forces that alone would be insufficient for weight-bearing, whereas in the example on the right the CPG activation levels were chosen to be sufficient to produce stable locomotion. In the first case, withdrawal of the stretch reflex contribution led to a collapse of the model, from which one would conclude that stretch reflexes are crucial to maintain stable locomotion. This would refute the suggestion we made in Cairns (hypothesis 1 above) that stretch reflexes are too weak and too delayed to contribute significantly to weight-bearing. But in the second case, when the levels of the CPG muscle activation profiles were 30% higher, i.e. enough to support body weight, the sudden addition of the stretch reflexes caused only a modest increase in the rate of locomotion and often decreased stability of the model. From this we would conclude that stretch reflexes had a rather minor role, which supports hypothesis 1. Thus it seems that the importance of stretch reflexes depends on whether central drive is well matched to the anticipated loads or not.



Figure 3.4. Schematic explaining how Principal Component Analysis (PCA) was used to evaluate the locomotor stability over a wide range of parameter variations. a Hip and knee forces must lie within the ellipse for stable locomotion to occur. The larger the ellipse, the greater the range of parameters for which locomotion is stable. b When another parameter is added, (in this case ankle extensor force), the stable region lies within a three-dimensional volume. The larger the volume the greater the range of stability. Orthogonal axes can be identified mathematically that point in the directions of largest variations of the coordinates within the volume. Unit vectors along these axes are called Principal Components. PCA may be extended to any number of parameters, in our case 6, corresponding to the 6 actuators in our model. PCA allowed us to compare the sizes of 6-dimensional stability "volumes" for different types of control. Since the first and second principal components described on average 90% of all variability in the stable parameters, the two-dimensional plots of Figures 3.4 provide nearly all of the information required to compare stability between control schemes over a large range of parameter variation.

To gain a better understanding of the range of central drive over which stretch reflexes would have a major versus a minor role, and also to test the validity of our conclusions in the face of parametric variations in both muscle stiffness parameters and limb geometry, we examined changes in the parametric space defined by forces of musculotendon actuators in the two models. Because six actuators were used in each limb in the models, parametric space was six-dimensional and therefore hard to represent with conventional Cartesian plots. However it is possible to simplify this representation by grouping variations in the six force parameters into "principal components." This is illustrated schematically in Fig. 3.4. First let us consider the effect on locomotion of changes in the peak forces of just two of the six actuators (hip flexors and knee extensors: Fig. 3.4a). Clearly if hip flexion is too forceful (point "a") the leg will overshoot, and locomotion will be destabilized. If there is too little hip flexion coupled with forceful knee extension (point "b"), locomotion will also be unstable. Only certain combinations of the two parameters will be compatible with stable gait. Suppose that these are contained within the shaded ellipse. Now let us add ankle extensors (Fig. 3.4b). Again there will be a restricted range of peak ankle extensor force compatible with stable locomotion. When combined with the stable combinations of the other two variables, the stable region may now be represented by the shaded volume in the 3-dimensional plot. By calculating eigen vectors and eigen values for the data autocorrelation matrix it is possible to define orthogonal axes along which the parameter combinations making up the volume are the most spread out. These axes thus "account for" the largest amount of variation in the data and are called the "principal components (PC)." Principal Component Analysis (PCA) can be extended to all six parameters.



Figure 3.5. Locomotor performance plots of stable trials of models without sensory feedback (gray circles) and with stretch reflex (black upward triangles) are plotted as projections on the planes of three extensor $(\mathbf{a-c})$ and three mainly flexor $(\mathbf{d-f})$ musculotendon actuators. Horizontal and vertical axes represent maximum isometric force at muscle rest length (Fmax in Eq. 3.1) of the corresponding musculotendon actuator. The large region of stability of the model without sensory feedback indicates that the intrinsic stiffness of the muscles sufficed to compensate for kinetic and kinematic variations over a fairly large parametric range. Since the origins of the coordinate systems corresponds to a simulation with zero muscle stiffnesses, the minimum vector to the stable region corresponds to the minimum muscle stiffnesses required for locomotion. Region of stability of the model without sensory contribution $(\mathbf{a-c})$, i.e. when stretch reflexes contribute 30% of overall extensor muscle excitation, less central activation is required to maintain stable locomotion. However, largely overlapping stability regions of flexors $(\mathbf{d-f})$ indicate that stretch reflexes are not essential for flexor activity without additional kinematic constraint.

3.3.2 Locomotor performance in the absence of sensory feedback

Locomotor performance was explored in six dimensional parameter space of musculotendon actuator forces using a gradient-descent method with random initial position. Figure 3.5 shows sets of parameters associated with stable locomotor performance of the model driven by the CPG alone (grey circles) and the CPG with the addition of stretch reflexes (black upward triangles). Figure 3.5a—c show the projections of six-dimensional data on the planes formed by the three extensor actuators. Where panels D, E and F show the projections of the same data on the planes formed by the remaining predominantly flexor actuators. The origin of the plots corresponds to zero force in all actuators. It is clear from Fig. 3.5 that stable locomotion can emerge over a fairly large range of force parameter variation, even in the absence of sensory feedback. Similar results have been previously obtained using inverse dynamics or neural networks to optimize activation patterns (Taga *et al.*, 1991; Taga, 1998; Neptune *et al.*, 2001; Ogihara & Yamazaki, 2001).

3.3.3 Effect of stretch reflexes on locomotor performance

Figure 3.5 shows a stability plot for the cat model where stretch reflexes contributed about 30% of the overall activity (see equation 3.6). In Fig. 3.5a—c the volume of stability associated with the model with sensory feedback (upward triangles) extends much closer to the origin than in the "deafferented" model (grey circles). This was also the case for the "horse" model (not illustrated). This indicates that stretch reflexes are capable of increasing extensor muscle stiffness when central drive is low and thus provide the necessary load compensation. In spite of the long latency of the reflexes and their dynamic nature, their 30% mean contribution essentially takes the place of the

missing 30% of central activation. This explains the destabilizing effect of a sudden removal of the stretch reflex contribution in Fig. 3.3a—c where the central activation levels were low. A contribution of stretch reflexes to flexor activity does not seem essential judging by the largely overlapping stable parameter spaces of both models in Fig. 3.3d—f.

In the example in Fig. 3.3d—f the addition of stretch reflexes when central activation was adequate to provide stable locomotion has a destabilizing effect. However, if we consider all the stable simulations obtained, the moment of inertia for the "stable" sets (see Eq. 3.3) of the model with stretch reflexes was 1.9 times larger than that without ($I_{CPG+SR} = 5.6*10^4 \text{ N}^2$, $I_{CPG} = 3.0*10^4 \text{ N}^2$). This indicates that stretch reflexes can contribute significantly to load compensation and taken over the entire parameter space, may even stabilize locomotion in many cases.

3.3.4 Contribution of state-dependent control to locomotor stability

In the stretch reflex modeling above, the cycle frequency of the CPG pattern was invariant. This placed constraints on the range of gait velocities consistent with stability. Adaptive control of cadence based on sensory information about actual biomechanical states might therefore be expected to increase stability. To test this hypothesis we implemented simple sensory rules to trigger the transitions between flexion and extension phases of the centrally generated pattern of muscle activation profiles. Figure 3.6 shows the extent to which If-Then sensory rules controlling step transitions (see Methods) stabilized the simulations in the cat model with CPG and stretch reflexes active. Figure 3.6a and c shows locomotor performance plots in the plane of two first principal components of the stable sets of parameters associated with the model with stretch reflexes in Fig. 3.5. The stability region is further defined by a contour plot, which shows the amount of time that elapsed in simulations before the model collapsed. In Fig. 3.6c the stable



Figure 3.6. PCA locomotor performance plots and plots of mean gait velocity versus stride length of stable trials of models with CPG and stretch reflexes (**a**, **b**) and with CPG, stretch reflexes and If-Then rules (**c**, **d**). The *contour* lines connect points of equal stability measured in seconds of stable gait. Region of stability in the model with If-Then rules is about 30% bigger, particularly in the direction of increase of hip extensor force. Labeled *straight* lines in **a** & **c** represent projections of the six-dimensional force axes (axis vector magnitude is 250N) on the plane of two first principal components. **b** Velocity was proportional to stride length in stable locomotion because the cycle frequency of the centrally generated pattern was constant in these trials. Stability was confined to a fairly narrow band of velocities. **d** Addition of If-Then rules to the control system shown in **a** and **b** increased the range of stable gait velocities.

region was enlarged by about 30% compared with that in Fig. 3.6a. Note that the main increase of stability coincides with the direction of the increasing force of hip extensor actuator, which regularly led to an increase of forward velocity.

An analysis of the velocities of gait corresponding to the stable and unstable regions showed that stability was associated with a fairly narrow band of gait velocities. Figure 3.6b shows that gait velocity was closely related to stride length for all of the stable simulations. This is not surprising, given that the frequency of the CPG rhythm in all of these simulations was constant. If stride length increases at a constant step cycle frequency, velocity *must* increase, or stability is lost. Consider the case where hip flexor forces are large but body velocity is low. The leg swings far forward and at ground contact the ground reaction force points back, decelerating the body. If velocity drops to zero before the centre of mass moves over the point of support, further forward motion becomes impossible and the model collapses backward. As we shall see, stretch reflexes do not fundamentally change this effect because in essence they just augment the inherent load compensation mechanism provided by muscle stiffness and viscosity (Partridge, 1966). On the other hand, conditional control can increase the range of stable velocities by adapting the cadence (cycle frequency) to the actual kinematics of the limb. The range of velocities of the stable simulations shown in the plot of Fig. 3.6d was greatly increased, showing that adaptive control of cadence granted by If-Then rules was associated with the increased range of stability.

3.3.5 Closed-loop gain of stretch reflex during locomotion

Figure 3.7 shows the dependence of the response of the system on the amplitude of the feedforward commands in our model. It is clear that increases in the central command are



Figure 3.7. Strength of stretch reflexes is modulated by muscle length. **a** Relationship between CPG amplitudes and relative reflex contribution for four sets of reflex gains. Both parameters are normalized to the minimal activity level required to produce stable locomotion in the "deafferented" model. Notice the decrease in the reflex component with the increasing CPG amplitude. **b** Mean extensor force did not change significantly with CPG amplitude except at the highest reflex gains. **c** Mean muscle length decreases with increasing CPG amplitudes. This effectively decreases stretch reflex loop gains (see text).

associated with decreases of the contribution of the stretch reflex loops. This cannot be attributed to changes in the generalized forces because these changes are quite small (Fig. 3.7b). Rather, Figure 3.7c demonstrates that higher activation levels drive the extensor muscles to operate at substantially shorter lengths, which attenuates the closed-loop gains of the length and force feedback loops, thus reducing the relative contribution of reflexes. As has been shown previously (Prochazka *et al.*, 1997), when muscles are short, they respond with a smaller force output to a given motoneuronal input and this in turn attenuates the gains of the feedback loops of which they are a part. Thus, the relative force contribution of proprioceptive reflexes is attenuated.

This mechanism may play a particularly important role when changes in muscle length are restricted, e.g. by experimental design. For example, numerous studies have been performed to investigate control of treadmill locomotion in decerebrate cats where the hips are firmly clamped or pinned to a strong metal frame over the treadmill. This constraint abolishes vertical motion of the hips, which is normally present in locomotion, and confines extensor muscle operation to longer lengths during the stance phase of gait. One would predict that extensor muscles would therefore operate further up the force-length curves, thus increasing the loop gain of the stretch reflex pathways. Estimation of the reflex contribution under these conditions would result in overestimates compared to normal unrestrained locomotion.

Figure 3.8 shows simulation results of "unrestrained" and treadmill locomotion with "fixed hips". The moderate increases of the operating length of extensors due to hip fixation resulted in a dramatic increase of the activation level and generated force. This indicates that experiments of this type, where hip motion is prevented may distort the neural control of locomotion significantly.


Figure 3.8. Effect of abolishing vertical hip motion on EMG, force and length. Restriction of vertical hip movement during walking increases activation levels of extensor muscles (a), increases extensor muscle forces (b), increases extensor muscle lengths during stance (c). This restriction, which is used in many decerebrated locomotor experiments, may lead to overestimations of the level of reflex contribution during locomotion.

3.4 Discussion

The purpose of this study was to gain insight into the contribution of sensory input to the control of locomotion. Though many experiments have been done on locomotor stretch reflexes, most of these have been limited to EMG measurements (Capaday & Stein, 1986; Gorassini *et al.*, 1994; Gritsenko *et al.*, 2001) rather than kinetic or kinematic changes of the whole limb (Sinkjaer *et al.*, 1988; Pearson *et al.*, 1999). The amount of EMG activation attributable to reflexes elicited by muscle length and force changes during the step cycle is generally less than 35% of the total (Stein *et al.*, 2000). Furthermore the reflex components have rather long latencies (Gorassini *et al.*, 1994; Gritsenko *et al.*, 2001). Finally, bursts of sensory input from stretch receptors elicited by electrical stimulation (Whelan & Pearson, 1997) or by muscle vibration (Ivanenko *et al.*, 2000) have surprisingly little impact on the kinematics of locomotion. All of these findings led us to propose that the biomechanical consequences of stretch reflexes elicited in the course of unperturbed gait did not contribute significantly to load compensation during weight-bearing locomotion.

The biomechanical modeling described in this paper was done to test the following hypotheses: 1) Stretch reflexes are too weak and too delayed to contribute significantly to weightbearing. 2) The important contributions of sensory input involve state-dependent processing. The results did not support hypothesis 1, but they did support hypothesis 2.

Hypothesis 1 was not supported, because stretch reflexes could rescue stable locomotion in cases where the amplitudes of the CPG activation profiles were insufficient to support locomotion. On the other hand, adding stretch reflex components to CPG activation profiles that *were* adequate to support stable locomotion tended to increase body height, propulsion and speed. When these increases were inappropriately scaled between muscles, locomotion became unstable. Thus although hypothesis 1 was not supported, nonetheless, stretch reflexes only seem to contribute significantly when central activation levels are low.

Regarding hypothesis 2, the incorporation of If-Then rules increased stability and in particular it greatly increased the range of cadences and velocities of stable locomotion, even though the time course of the actual muscle activation profiles did not change. How could gait velocity vary without the activation profiles changing? The answer is that the triggering of say the stance-swing transition terminates the reading-out of the extensor muscle activation profiles, even if they are only half completed, and initiates the reading-out of the flexor activation profiles. Similarly flexor profiles are terminated earlier in the next half cycle. The interesting point here is that cadence and gait velocity can be varied over a wide range without changing the time-course or amplitude of the activation profiles, but merely by skipping parts of these profiles and resetting to a new part of the step cycle. To our knowledge, this possibility has not been recognized until now.

It is important to mention some of the restrictions and limitations of the modeling performed. A general criticism of modeling of this type is that when the model contains many parameters, there is scope for choosing sets of parameters that fit one's favored hypothesis. To safeguard against this, we deliberately avoided making the models represent faithfully the anatomy of a particular "animal." Furthermore, our approach to the exploration of parameter space using an observer-independent stability search algorithm and PCA allowed for an unbiased representation of stability over several hundred sets of parameters in two models that represented extremes of limb geometry. The conclusions we drew regarding the two hypotheses were therefore broad-based and not model-specific, i.e. they remained valid in the face of a large range of parameter variations.

This is not to say that we were able to explore all types of parameter variations that could influence the conclusions. For example, the muscle models did not take into account the hysteretic

property of short-range stiffness (Rack & Westbury, 1973). We intend to add this characteristic in future modeling work. We did not explore all possible muscle spindle and tendon organ models (Prochazka & Gorassini, 1998), nor did we vary the latency of the stretch reflex, nor try to represent separate short-, medium- and long-latency reflex components or heterogeneous reflex connections. There is in any case considerable uncertainty in the literature regarding the relative sizes of such components. In previous modeling, we have found that varying reflex latencies within reasonable limits did not change muscle kinetic responses substantially (Prochazka *et al.*, 1997). By varying the level of the stretch reflex contribution during locomotion we found that closed-loop gain of the sensory evoked responses depended strongly on muscle length and velocity. This dependence was previously implicated in stabilizing positive force feedback (Prochazka *et al.*, 1997). Since estimation of the gain of the closed-loop reflex system during various tasks poses a considerable experimental challenge, the relative level of contribution of the stretch reflex to muscle excitation remains a contentious issue. Models such as ours may greatly clarify and assist in evaluation of components of feedforward and feedback systems of motor control.

To conclude, our model provided insight into the interplay between neural commands and biomechanical properties of the musculoskeletal actuators of the locomotor system. The main conclusion was that homonymous stretch reflexes are capable of adjusting muscle forces at several joints to achieve weight bearing during gait, and this is particularly significant when central activation is low. Simple finite state rules provide dramatic improvements in flexibility and stability of level overground locomotion in our model. Cadence is constantly adjusted to kinematic state. This also raises the possibility that gait velocity could be controlled by changing the firing thresholds of sensory rules in a state-dependent controller. Our method of testing stability by exploring parametric space using PCA is novel and provides a systematic measure of the generality of the conclusions.

3.5 References

- ABELEW, T. A., MILLER, M. D., COPE, T. C. & NICHOLS, T. R. (2000). Local Loss of Proprioception Results in Disruption of Interjoint Coordination During Locomotion in the Cat. J Neurophysiol 84, 2709-2714.
- AKAZAWA, K., ALDRIDGE, J. W., STEEVES, J. D. & STEIN, R. B. (1982). Modulation of stretch reflexes during locomotion in the mesencephalic cat. J Physiol (Lond) 329, 553-567.
- BENNETT, D. J., DE SERRES, S. J. & STEIN, R. B. (1996). Gain of the triceps surae stretch reflex in decerebrate and spinal cats during postural and locomotor activities. J Physiol 496, 837-850.
- BICKEL, A. (1897). Ueber den einfluss der sensibelen nerven und der labyrinthe auf die bewegungen der thiere. Pfluegers Arch. Ges. Physiol. 67, 299-344.
- BLOEM, B. R., ALLUM, J. H., CARPENTER, M. G., VERSCHUUREN, J. J. & HONEGGER, F. (2002). Triggering of balance corrections and compensatory strategies in a patient with total leg proprioceptive loss. Experimental Brain Research 142, 91-107.
- BROWN, T. G. (1911). The intrinsic factor in the act of progression in the mammal, vol. 84. Proc. Roy. Soc., London.
- CAPADAY, C. & STEIN, R. B. (1986). Amplitude modulation of the soleus H-reflex in the human during walking and standing. Journal of Neuroscience 6, 1308-1313.
- DIETZ, V., DISCHER, M., FAIST, M. & TRIPPEL, M. (1990). Amplitude modulation of the human quadriceps tendon jerk reflex during gait. Experimental Brain Research 82, 211-213.
- ENGBERG, I. & LUNDBERG, A. (1969). An electromyographic analysis of muscular activity in the hindlimb of the cat during unrestrained locomotion. Acta Physiologica Scandinavica 75, 614-630.
- GILLARD, D. M., YAKOVENKO, S., CAMERON, T. & PROCHAZKA, A. (2000). Isometric muscle length-tension curves do not predict angle-torque curves of human wrist in continuous active movements. Journal of Biomechanics 33, 1341-1348.
- GORASSINI, M. A., PROCHAZKA, A., HIEBERT, G. W. & GAUTHIER, M. J. A. (1994). Corrective responses to loss of ground support during walking in intact cats. Journal of Neurophysiology 71, 603-610.
- GOSLOW, G. E., REINKING, R. M. & STUART, D. G. (1973). The cat step cycle: Hind limb joint angles and muscle lengths during unrestrained locomotion. J. Morph. 141, 1-42.

- GRANAT, M. H., HELLER, B. W., NICOL, D. J., BAXENDALE, R. H. & ANDREWS, B. J. (1993). Improving limb flexion in FES gait using the flexion withdrawal response for the spinal cord injured person. Journal of Biomedical Engineering 15, 51-56.
- GRILLNER, S. & ZANGGER, P. (1974). Locomotor movements generated by the deafferented spinal cord. Acta Physiologica Scandinavica 91, 38-39A.
- GRITSENKO, V., MUSHAHWAR, V. & PROCHAZKA, A. (2001). Adaptive changes in locomotor control after partial denervation of triceps surae muscles in the cat. Journal of Physiology 533, 299-311.
- HALBERTSMA, J. M. (1983). The stride cycle of the cat: the modelling of locomotion by computerized analysis of automatic recordings. Acta Physiologica Scandinavica. Supplementum 521, 1-75.
- HE, J., LEVINE, W. S. & LOEB, G. E. (1991). Feedback gains for correcting small perturbations to standing posture. IEEE Transactions on Automatic Control 36, 322-332.
- HILL, A. V. (1938). The heat of shortening and the dynamic constants of muscle. Proceedings of Royal Society 126B, 136-195.
- IVANENKO, Y. P., GRASSO, R. & LACQUANITI, F. (2000). Influence of leg muscle vibration on human walking. Journal of Neurophysiology 84, 1737-1747.
- LAJOIE, Y., TEASDALE, N., COLE, J. D., BURNETT, M., BARD, C., FLEURY, M., FORGET, R., PAILLARD, J. & LAMARRE, Y. (1996). Gait of a deafferented subject without large myelinated sensory fibers below the neck. Neurology 47, 109-115.
- NEPTUNE, R. R., KAUTZ, S. A. & ZAJAC, F. E. (2001). Contributions of the individual ankle plantar flexors to support, forward progression and swing initiation during walking. J Biomech 34, 1387-1398.
- OGIHARA, N. & YAMAZAKI, N. (2001). Generation of human bipedal locomotion by a bio-mimetic neuro-musculo-skeletal model. Biological Cybernetics 84, 1-11.
- PARTRIDGE, L. D. (1966). Signal-handling characteristics of load-moving skeletal muscle. American Journal of Physiology 210, 1178-1191.
- PEARSON, K. G., FOUAD, K. & MISIASZEK, J. E. (1999). Adaptive changes in motor activity associated with functional recovery following muscle denervation in walking cats. Journal of Neurophysiology 82, 370-381.
- PEARSON, K. G., MISIASZEK, J. E. & HULLIGER, M. (2003). Chemical ablation of sensory afferents in the walking system of the cat abolishes the capacity for functional recovery after peripheral nerve lesions. Exp Brain Res 150, 50-60.

- PROCHAZKA, A. (1993). Comparison of natural and artificial control of movement. IEEE Transactions on Rehabilitation Engineering 1, 7-16.
- PROCHAZKA, A. (1999). Quantifying proprioception. In Peripheral and spinal mechanisms in the neural control of movement. ed. BINDER, M. D. Elsevier, Amsterdam.
- PROCHAZKA, A., GILLARD, D. & BENNETT, D. J. (1997). Implications of positive feedback in the control of movement. J Neurophysiol 77, 3237-3251.
- PROCHAZKA, A. & GORASSINI, M. (1998). Models of ensemble firing of muscle spindle afferents recorded during normal locomotion in cats. Journal of Physiology 507, 277-201.
- PROCHAZKA, A., GRITSENKO, V. & YAKOVENKO, S. (2002). Sensory control of locomotion: reflexes versus higher-level control. In Sensori-Motor Control. ed. GANDEVIA, S. G., PROSKE, U. & STUART, D.G., pp. In Press. Kluwer Academic/Plenum Publishers, London, New York.
- RACK, P. M. & WESTBURY, D. R. (1973). The short range stiffness of active mammalian muscle. Journal of Physiology 229, 16P-17P.
- SINKJAER, T., TOFT, E., ANDREASSEN, S. & HORNEMANN, B. C. (1988). Muscle stiffness in human ankle dorsiflexors: intrinsic and reflex components. Journal of Neurophysiology 60, 1110-1121.
- STEIN, R. B., MISIASZEK, J. E. & PEARSON, K. G. (2000). Functional role of muscle reflexes for force generation in the decerebrate walking cat. Journal of Physiology 525, 781-791.
- TAGA, G. (1998). A model of the neuro-musculo-skeletal system for anticipatory adjustment of human locomotion during obstacle avoidance. Biological Cybernetics 78, 9-17.
- TAGA, G., YAMAGUCHI, Y. & SHIMIZU, H. (1991). Self-organized control of bipedal locomotion by neural oscillators in unpredictable environment. Biological Cybernetics 65, 147-159.
- WHELAN, P. J. & PEARSON, K. G. (1997). Comparison of the effects of stimulating extensor group I afferents on cycle period during walking in conscious and decerebrate cats. Experimental Brain Research 117, 444-452.
- YAKOVENKO, S., MUSHAHWAR, V., VANDERHORST, V., HOLSTEGE, G. & PROCHAZKA, A. (2002). Spatiotemporal Activation of Lumbosacral Motoneurons in the Locomotor Step Cycle. J Neurophysiol 87, 1542-1553.
- YANG, J. F., STEIN, R. B. & JAMES, K. B. (1991). Contribution of peripheral afferents to the activation of the soleus muscle during walking in humans. Exp Brain Res 87, 679-687.

CHAPTER 4

Spatiotemporal activation of lumbosacral motoneurons in the locomotor step cycle

Adapted from an original publication:

S. Yakovenko, V. Mushahwar, V. VanderHorst, G. Holstege and A. Prochazka Journal of Neurophysiology, 87, p.1542-1553, 2002

4.1 Introduction

It has been known for centuries that for a short time after cervical spinalization, many animals can still generate a locomotor rhythm (Mettrie de La, 1745). T. Graham Brown coined the term "intrinsic factor" to describe the rhythm generator in the spinal cord and suggested a simple "half-center" oscillator as the underlying mechanism (Brown, 1911). Brown's intrinsic factor was renamed the "central pattern generator" (CPG): (Grillner & Zangger, 1975). Many experiments have been done in different species to localize and characterize the basic neuronal machinery of locomotor CPGs (Kiehn et al., 1998). Broadly speaking, opinion is split between those who believe that the vertebrate CPG is fairly localized (Cazalets et al., 1995) and those who think it is distributed (Deliagina et al., 1983; Arshavsky et al., 1997; Kremer & Lev-Tov, 1997; Kiehn & Kjaerulff, 1998). One version of the "distributed" hypothesis envisions a series of semi-autonomous "unit" oscillators distributed along the neuraxis and dynamically coupled by propriospinal pathways (Grillner & Wallen, 1985). A consensus seems to be emerging that rhythmogenesis is more robust in one or two spinal cord segments just rostral to the lumbosacral enlargement than in more caudal segments, and that these rostral segments may "lead" the other segments in generating the locomotor rhythm (Deliagina et al., 1983; Arshavsky et al., 1997; Kiehn & Kjaerulff, 1998; Marcoux & Rossignol, 2000).

It is impossible with current techniques to measure directly the activity of large ensembles of neurons of a particular functional type in the mammalian spinal cord during normal behaviour. Though this may be possible in the future by identifying neurons according to some feature such as size, molecular affinity or luminance, to our knowledge this has not been attempted to date. Consequently, all of the experiments on localization of pattern generators within the spinal cord have been performed on reduced preparations of different types, including isolated embryonic 105 spinal cords of the rat (Lev-Tov & O'Donovan, 1995), in vitro portions of fetal rat spinal cords (Hochman et al., 1994; Kiehn & Kjaerulff, 1998), spinalized fictive locomotor cat (Andersson et al., 1978; Grillner & Zangger, 1979; Deliagina et al., 1983) and lamprey (Zhang et al., 1996; Grillner et al., 1998). In these various experimental situations the locomotor rhythm is usually initiated and maintained by the application to the spinal cord of pharmacological agents such as dihydroxyphenylalanine (DOPA) or N-methyl-D-aspartate (NMDA) or by electrical stimulation applied to descending or peripheral sensory pathways. The rhythms generated in muscle nerves or ventral roots can often have many of the specific characteristics of those seen in normal locomotion, so the assumption is made that the patterns of central neural activity studied in this way are similar to those occurring in normal behavior. However, not only are descending inputs absent in these preparations but in most cases so are sensory inputs from actively moving limbs. Descending control originating in cerebral, cerebellar and midbrain areas has a profound influence on the amplitude and timing of muscle contractions during locomotion (Armstrong & Marple-Horvat, 1996; Rho et al., 1997). Step-cycle phase transitions, as well as modulation of motoneuron (MN) activity during swing and stance phases depend heavily on sensory input during normal locomotion (Grillner & Zangger, 1975; Grillner & Rossignol, 1978; Pearson, 1995; Prochazka, 1996).

The aim of our study was to characterize the spatiotemporal migration of neuronal activity in the cat spinal cord during normal locomotion. We combined two sets of data: the known distribution of MN pools innervating different muscles within the spinal grey matter and the known EMG activity profiles of these muscles in the locomotor step cycle.

The approximate rostrocaudal distribution of MN pools in the cat spinal cord was originally deduced from the distribution of ventral roots innervating different hindlimb muscles (Sherrington, 1892). Several important studies have greatly extended the scope and accuracy of this information

(Romanes, 1951; Vanderhorst & Holstege, 1997). In addition, there is now a large body of information on the location and reflex connections of interneurons of different types (Jankowska, 1992) and on the distribution of the terminals of sensory axons (Brown, 1981).

Each action potential in a MN propagates along the efferent axon and gives rise to a motor unit action potential (MUAP) in the muscle the MN innervates. All of the MUAPs generated by the numerous active MNs sum in a slightly nonlinear manner (see Discussion) to produce the recordable signal called the electromyogram (EMG). The rectified EMG thus provides an indirect measure of the net firing of MNs of that muscle in the spinal cord at any particular moment. In numerous functional studies over the last three decades the EMG patterns of cat hindlimb muscles recorded during locomotion have been described in great detail (see Methods).

The model of MN distributions in the spinal cord and the model of spatiotemporal activity described in this paper are in digital form. This will allow them to be used in other laboratories not only to test for differences in spinal activity patterns in different preparations (e.g. fictive locomotor versus normal), but also to help predict which muscles will be activated or inactivated by local electrical stimulation of the spinal cord (Mushahwar & Horch, 1998; Mushahwar *et al.*, 2000), lesions or local excitation or inhibition with microinjections of pharmacological agents (Marcoux & Rossignol, 2000). A preliminary report on this work has been published (Yakovenko *et al.*, 2000).



Figure 4.1. Anatomical models of the organization of the MN pools within the cat lumbar enlargement. The pools innervating muscles acting about the hip (left), knee (middle) and ankle (right) are presented separately. The anatomical data were digitized from 30 summary drawings or "slices" (Vanderhorst & Holstege, 1997). The rostrocaudal axis is normalised and divided into 100 levels. On average slices were 0.96 mm thick and corresponded to 3.3 levels. The models show a structure 29 mm long on average, spanning the caudal L₄ to rostral S1 segments of the spinal cord. The fusiform MN pools within the grey matter are long and intertwined (MN pools are identified by colour, abbreviation and a number in each of the three representations). The schematic bars on the left show their rostrocaudal extent. The centers of mass are shown as open circles on these bars and their positions in the transverse plane are shown in the bottom panels. The lines leading from the abbreviations point to the location of the corresponding center of mass. Note the rostral locations of hip flexor pools (psoas, SRT, iliacus) and more caudal locations of hip extensor pools (e.g. SMa, BFa). The reverse relationship is true of MN pools innervating knee muscles: knee flexor MN pools (e.g. BFp, ST) are more caudal and knee extensor MN pools (e.g. VL, RF) more rostral. Ankle MN pools are relatively close together in the caudal region, with flexors slightly more rostral than extensors. BFp, SMp and ST are bifunctional muscles that flex the knee and extend the hip, so their MN pools are represented in both the middle and left panels.

4.2 Methods

4.2.1 Digitization

In a recent study the horseradish peroxidase (HRP) staining technique was used to determine the distributions within the cat spinal cord of 50 MN pools innervating hindlimb, pelvic floor and lower back muscles (Vanderhorst & Holstege, 1997). The entire lumbosacral enlargement of the frozen spinal cord (mean length 28.8 mm) was cut into 40 μ m sections. Every second section was collected. Groups of 12 such sections were used to construct composite digital drawings showing the positions of all stained MNs in the group (i.e. each drawing corresponded to a 12 x 2 x 0.04 = 0.96-mm thick slice of spinal cord). The rostral and caudal ends of the enlargement were identified on the basis of distinctive transitions in shape of the ventral horn. Because the enlargements varied in length between animals, a normalization procedure was used to divide each enlargement into 100 levels. A summary figure of 30 slices was presented (Vanderhorst & Holstege, 1997) (Fig. 28), corresponding to a mean length of the enlargement of 30 x 0.96 = 28.8 mm.

Based on the EMG profiles available to us (see following text), we digitized the positions of MNs of the following 27 hindlimb muscles: adductor femoris magnus (AdFM), biceps femoris anterior and posterior (BFa, BFp), caudofemoralis (CF), extensor digitorum longus (EDL), flexor digitorum longus (FDL), flexor hallucis longus (FHL), gluteus maximus and medius (GlutMax, GlutMed), gastrocnemius lateralis and medialis (LG, MG), gracilis, iliacus, plantaris, psoas, rectus femoris (RF), sartorius anterior and sartorius medialis (SRTa, SRTm), soleus, semimembranosus anterior and posterior (SMa, SMp), semitendinosus (ST), tibialis anterior (TA), tibialis posterior (TP), vastus intermedius, vastus lateralis and vastus medialis (VI, VL, VM).

Initially we classified MNs as innervating either flexors or extensors, according to the predominant biomechanical action of the corresponding muscles or parts of muscles activated in the flexor and crossed extensor reflexes (Sherrington, 1910; Eccles & Lundberg, 1958; Carrasco & English, 1999). Flexors: BFp, EDL, gracilis, iliacus, psoas, RF, SMp, SRTa, SRTm, ST, TA; extensors: AdFM, BFa, FDL, FHL, GlutMax, GlutMed, LG, MG, plantaris, soleus, SMa, TP, VI, VL, VM. However, we will argue later that in locomotion BFp, SMp and ST muscles and possibly Gracilis and FDL form a separate functional group and their MNs produce a distinct locus of activation in the spinal cord.

In the anatomical study (Vanderhorst & Holstege, 1997) two characteristics of the spinal cord were recognised when determining the spatial relation between lumbosacral MN pools. First, there were major differences in the organization of spinal rootlets between cats and so segmental localisation varied. Second, the length of the lumbar enlargement varied considerably (mean: 28.8 mm \pm 2.4 mm (SD), range: 23.0 – 34.1mm (n = 120)). To normalise the data across animals, two easily recognizable landmarks were chosen to identify the rostral and caudal ends of the enlargement. The rostral landmark (defined as level 0) was a sudden lateral extension or bulging of the ventrolateral corner of the ventral horn into the white matter in the midlumbar cord. The caudal landmark (defined as level 100) was in the sacral cord where the border of the ventral horn changes from a rounded profile to a straight line joining the ventromedial and dorsolateral corners of the horn. These changes in shape at levels 0 and 100 took place over a distance of 0.5 - 1.0 mm (-1.5 to 3 levels). Locating them accurately required careful comparisons of several sections in the vicinity of the transitions. The 0- to 100-level scale allowed the relative distributions of many MN pools to be compared. In the numerous case studies illustrated (Vanderhorst & Holstege, 1997) the rostrocaudal locations of the centers of mass of individual MN pools ranged over maximally 10 levels (i.e., localisation was reliable to within 10% of the length of the enlargement). It should be



Figure 4.2. MN distribution along the rostrocaudal axis of the lumbar enlargement of the cat spinal cord. The enlargement was divided into 30 slices as described in the text. The number of MNs digitized from the source data in each slice is plotted. As no distinction was made between alpha and gamma MNs, the number of alpha MNs is probably overestimated (see text). The total population distribution is shown as a dashed line in each panel, and reveals a significant imbalance between the rostral and caudal halves of the lumbosacral enlargement. *A.* flexor MNs are concentrated mainly in the rostral part of the spinal cord (levels 0 to 50). *B.* Extensor MNs are more caudal, with peak density in the L_7 segment. Two distributions are shown, one that includes FDL MNs (dotted) and one that does not (solid: see text). *C.* Distribution of MNs innervating BFp, gracilis, SMp and ST (with and without FDL). We argue in the text that these muscles, normally classified as flexors, have a biomechanical role in the step cycle that is best described as retraction.

Table 4.1. Rostral (L_5 - L_6) and caudal (L_7 - S_1) populations of extensor, BFp-FDL-gracilis-SMp-ST and flexor MNs in the cat lumbar enlargement.

	extensor motoneurons		BFp, FDL, gracilis, SMp,ST motoneurons		is, flexor moton	flexor motoneurons		all motoneurons	
rostral half (L_4 - L_6) caudal half (L_7 - S_1)	549 1817	23% 77%	155 752	17% 83%	620 141	81% 19%	1324 2710	33% 67%	
$full (L_4 - S_1)$	2366		907		761		4034		

noted that this normalisation procedure was seen as an important advance over previous methods, in which relative positions of MN pools could only be studied accurately in individual animals.

To enter the positions of MNs into the model we scanned Figure 28 of (Vanderhorst & Holstege, 1997), at high resolution (600 x 600 dpi). A software program written in Matlab (version 5.3, The MathWorks) was used to display each scanned slice on a 17 inch computer monitor and to digitize the coordinates of individual MNs together with the outline of the grey matter. The coordinates, stored in ASCII format, were imported into a three-dimensional (3-D) model constructed with a second Matlab software program. In the model, the slices were all represented as being 0.96-mm thick and accordingly the length of the enlargement was represented as 28.8 mm. It should be stressed that in the original study lumbar enlargements varied in length over the range 23.0 - 34.1 mm (Vanderhorst & Holstege, 1997). In this paper we will therefore usually refer to the 100 normalized levels, however certain calculations such as that of the center of activation require absolute positions. The slices were aligned using the central canal for lateral centering and the midline between dorsal and ventral horns for dorsoventral centering. As the dimensions of individual cross-sections varied, the absolute positions of MNs and the boundaries of MN pools in the slices can only be considered accurate to within the variance of the original dimensions. In unpublished data from five cats, one of the authors (V. Mushahwar) made measurements of the width and thickness of the spinal cord at different segmental levels. The mean distance from the midline to edge of the white matter at L₇ level was 3.6 ± 0.04 mm and the corresponding mean dorso-ventral thickness of the spinal cord was 5.5 ± 0.1 mm. As most MNs were located about halfway from the midline to the edge of the white matter and less than half of the thickness of the cord from the central canal, we estimate SDs of the absolute positions of MNs to be no more than 0.02 mm anterolaterally and 0.05 mm dorsoventrally.

The Matlab functions for visualization of 3-D structures were used to generate a 3-D model of the lumbar enlargement. Figure 4.1 shows graphical displays of this model in which surfaces have been "rendered" with SURFdriver version 3 software (Surfdriver.com).

EMG patterns of cat limb muscles have been studied extensively by many researchers. In a previous paper from this laboratory we reviewed the available data and constructed a chart of standardised EMG profiles of a variety of muscles in the cat step cycle (Prochazka et al., 1989). In the present study we again reviewed all the currently available literature and collated as many examples as possible of segments of EMG locomotor recordings from hindlimb muscles. From this collection we constructed EMG profiles representing the activity in one step cycle for each of 27 muscles. We considered doing this mathematically by digitizing and processing the individual traces from the different studies to produce averages aligned to foot contact and normalised to one cycle period, however we decided against this for the following reasons. The EMG recordings were not of uniform quality or format. Some EMG signals were rectified and smoothed while others were unrectified. Some were averages of several steps while others were raw traces from single cycles. Some recordings included reliable measurements of ground contact and foot lift while others did not. Cycle durations varied and as a result so did the relative durations of swing and stance phases. Even when EMG recordings are made in the same laboratory with the same techniques there is some variation in the time course of EMG profiles from a given muscle that appears to depend on electrode dimensions and placement as well as possible differences in gait between animals. Taking all of these factors into consideration, we therefore opted to construct each profile from linear segments after carefully comparing the available records in the collection and using our judgment on critical features such as the timing of onset and duration of the EMG signal, the timing of the peaks and the overall shape of the signal. Profiles were constructed from a number of piecewise

linear segments, the onset, duration and slope of which were entered digitally in a standardized spreadsheet of the step cycle in which the landmarks were the moments of foot lift and foot contact. The swing phase occurring between these landmarks was standardized to 30% of the cycle, a proportion associated with gait velocities in the range 0.6 - 0.8 m/s and cycle periods of about 700ms (Goslow et al., 1973; Halbertsma, 1983). Particular attention was paid to the time course of EMG signals at and around these critical points in the cycle. When possible, concurrent EMG signals from ankle extensors MG, LG or soleus were used as a reference for foot contact, as the mean delay between the onset of EMG in these muscles in the E1 phase and the moment of foot contact is 60ms, with a range of ± 5 ms (Trend, 1987; Prochazka & Gorassini, 1998). Muscle length or joint angle traces, when available, also helped us to identify timing landmarks. Each profile was normalized, i.e., it reached a peak of 100% at least once in the cycle. The assumption implicit in this normalization is that all muscles are fully recruited at least once in the step cycle. Clearly this does not occur in slow locomotion (Walmsley et al., 1978). However we viewed this simply as a matter of scaling; the important point was that all MN pools, whether flexor, extensor or retractor, reached the same maximal level of recruitment when their EMG signals were maximal. Different relative levels of recruitment of flexors and extensors were then explored to address the issue of parametric sensitivity.

The step cycle was divided into 100 equal intervals corresponding to about 7ms each for gait of moderate speed. For each of these intervals the normalized level of EMG of each hindlimb muscle was obtained from the corresponding profile in the chart. The EMG levels from each muscle were used to scale the number of dots displayed within the volume of the corresponding MN pool in the digital model. This assumes a linear relationship between EMG and MN firing rate (Hoffer *et al.*, 1987), an assumption we discuss in detail later. To simplify the simulation, we also assumed a

random topographical distribution of active MNs within each MN pool. The 100 consecutive images or frames were saved in bitmap format and when displayed in rapid sequence with Corel Photo-Paint software they formed a continuous-loop movie showing the spatiotemporal activity of hindlimb MN pools. The effect of our simplifying assumptions on the results and conclusions will be dealt with in more detail in the Discussion. However, to gain an overall idea of the parametric sensitivity of our main conclusions to details of EMG profiles, we did a set of simulations in which the profiles were converted to all-or-nothing step functions such that all non-zero values were assigned the value 100%.

4.2.2 Calculation of the center of MN activity

The locus of MN activity was analyzed by calculating the center of mass (r_{cm}) of the n active MNs (the "center of activity") at each of the 100 moments in the step cycle, using the following formula:

$$r_{cm} = \frac{\sum_{i=1}^{n} r_i}{n}$$
 Eq. 4.1

where r_i is the vector from the origin to the *i*th MN (The origin being defined as the rostral-most point of the neuraxis). As mentioned above, the center of activity was expressed in terms of absolute position within the standard 28.8 -mm long enlargement.

4.3 Results

4.3.1 Anatomical distribution of MN pools in the lumbosacral spinal cord

Figure 4.1 shows three views of the 3-D model of the lumbar enlargement of the cat spinal cord, rendered from the digitized transverse slices of HRP-stained MNs described above. Each of the three

displays shows the outlines of the gray and white matter as well as MN pools of muscles acting about the hip, knee and ankle respectively. The Figure illustrates some interesting general features of MN pool distributions. The pools are long, spindle-shaped and intertwined. The bars on the left of each model were included to show the rostrocaudal extent and the center of mass of each pool.

As is well known from classical myotomal charts (Romanes, 1951; Sharrard, 1955), there is a correspondence between the rostrocaudal position of MN pools and the anatomical position of their respective muscles in the leg. It follows that there must be a correlation between the position of MN pools and the biomechanical action with which they are associated. Thus starting at level 0 at the rostral end of the enlargement and moving caudally to level 80 we find the MN pools of muscles distributed in proximal to distal order in the anterior compartment of the leg: hip flexors (psoas, SRTm, SRTa, iliacus) knee extensors (VL, VM, RF, VI) and ankle flexors (EDL, TA). When the leg is pendant, these muscles all act to move the foot forward. Starting from level 40 and moving caudally to level 100 (S₁ segment), we find the hip extensor pools (AdFM, Gracilis, SMa, SMp), followed by toe plantarflexors (FDL, FHL) and ankle extensors (MG, LG, Pl, soleus). The most caudal part of the enlargement also contains MN pools of proximal muscles with complex actions about hip and knee (GlutMed, ST, CF, BFa, BFp). When the leg is pendant these muscles all assist in moving the foot and toe back. Note that in terrestrial locomotion, some of the muscles assume a load-compensating role which can no longer be described in terms of a simple forward-backward dichotomy (see

Discussion).



Figure 4.3. EMG patterns of hindlimb muscles during forward level locomotion at moderate speeds (~ 0.6 m/s). The normalized patterns are plotted for a full step cycle, starting from the onset of swing. The left and right panels show respectively the EMG profiles of extensors, flexors and the muscles we classify as retractors. FDL, which has a variable EMG pattern, is shown dashed in both extensor and retractor groups. These EMG profiles were constructed from piece-wise linear segments from numerous sources in the literature, the numbers in each panel referring to the papers listed below. The profiles were used in the subsequent figures to modulate the number of MNs shown as being active in the spatiotemporal model.

References: 1. (Abraham et al. 1985), 2. (Belanger et al. 1996), 3. (Buford and Smith 1990a), 4. (Carrier et al. 1997), 5. (Chanaud et al. 1991a), 6. (Duysens and Loeb 1980), 7. (Engberg 1964), 8. (Engberg and Lundberg 1969), 9. (Forssberg 1979b), 10. (Forssberg et al. 1980), 11.

(Gritsenko et al. 2000), 12. (Halbertsma 1983), 13. (Hoffer et al. 1987a), 14. (Hoffer et al. 1987b), 15. (Loeb and Hoffer 1985), 16. (Loeb et al. 1985), 17. (O'Donovan et al. 1982), 18. (Pratt et al. 1996), 19. (Pratt et al. 1991), 20. (Prochazka et al. 1976), 21. (Prochazka et al. 1989),

22. (Rasmussen et al. 1978), 23. (Rossignol et al. 1986), 24. (Smith et al. 1998), 25. (Trank and Smith 1996), 26. (Wand et al. 1980). The numerical distributions of the digitized MNs are illustrated in Fig. 4.2. There is a significant imbalance between the sizes of rostral and caudal populations of MNs. The caudal half of the enlargement (level 50 – 100) contains far more MNs than the rostral half. Indeed it is well known that the caudal ventral horns have lateral extensions that accommodate the increased numbers of MNs (Vanderhorst & Holstege, 1997). Most of the MNs in this region innervate extensor muscles (Fig. 2B). Over 75% of all hindlimb extensor MNs are in this region (Table 1) and their density reaches a peak at about level 80. Flexor MNs (Fig. 4.2A) are less tightly clumped but even so, over 80% of them are found in the rostral half. The distributions of BFp, FDL, gracilis, SMp and ST MNs are illustrated separately in Fig. 4.2C. These MNs are mainly in the caudal half of the enlargement. We will argue that the function of these muscles in locomotion is to provide the final extensor thrust during the stance phase, and then to provide ground clearance for the swing phase by extending the hip while at the same time flexing the knee and the toes. For this reason we have borrowed the term "retractors" from the invertebrate literature to describe them (see Discussion).

4.3.2 Analysis of the spatiotemporal pattern of MN activity

Figure 4.3 summarizes the EMG profiles of all the muscles we included in the model during level forward walking, starting from the onset of swing (see Methods for details on how the profiles were constructed). The muscles in each group behave rather uniformly during locomotion. The flexors are active mainly in the early and mid-swing phase, when the foot is off the ground. The extensors become active during the last 100 ms of swing. In fact this part of the step cycle is usually designated E1, the first extensor phase of the step cycle (Philippson, 1905). The bifunctional muscles BFp, gracilis and ST show two EMG bursts in the step cycle, at the transitions from stance to swing and from swing to stance respectively (SMp does not always show this burst clearly). These muscles produce a backward and upward motion of the foot at the end of the stance phase, a 119

function we will argue below is best described as *retraction*. FDL may contribute to this action by plantarflexing the paw. FDL was originally considered an extensor, along with its synergist FHL (Sherrington 1910). Several EMG studies supported this view, showing FHL and FDL to have extensor bursts similar to those of the gastronemii and soleus. However, in a study that focussed specifically on the biomechanical actions and EMG of FDL and FHL, it was concluded that unlike FHL, FDL acted predominantly as a toe flexor at the stance-swing transition with a brief, intense burst of EMG (O'Donovan *et al.*, 1982). It was suggested that the EMG during the stance phase recorded by others may be due to cross-talk. The flexor burst pattern of FDL is supported in neurograms recorded during fictive locomotion (Fleshman *et al.*, 1984) and in more recent EMG recordings in normal cats (Loeb, 1993; Trank & Smith, 1996; Smith *et al.*, 1998), though Loeb's group reported variable amounts of stance phase activity that was unlikely to be due to cross-talk. Given the ambiguity regarding FDL in the literature, we performed two separate analyses of the data, in one case assuming a flexor (retractor) burst EMG profile for FDL and in the other an extensor profile (see dashed profiles in Fig. 4.3).

We were unable to include some muscles with MNs in the lumbar enlargement for the following reasons. There were no recordings from intrinsic hip muscles such as obturator, pyriformis and gemellus in the literature. Tensor fascia latae, a thin, flat muscle overlying the vasti, has anterior and posterior portions which flex or extend the hip respectively (Chanaud *et al.*, 1991). These portions were not separated in the anatomical study (Vanderhorst & Holstege, 1997), and so EMG profiles could not be duly assigned. For the same reason, the small intrinsic muscles of the foot are omitted from our analysis. A case in point is extensor digitorum brevis. This muscle is innervated by branches of the deep peroneal nerve and so its MNs are probably close to those of EDL, in caudal L_6 and rostral L_7 segments. EDB EMG shows a burst of activity at the onset of the

stance phase (Trank & Smith, 1996), so this might slightly alter the activity in our model. Given the likely small number of MNs involved, this is unlikely to be a major effect.

Conversely, though the distribution of MNs of the minor portions of adductor femoris (brevis and longus) were available, the only EMG recordings in the literature were from the major portion (magnus), so the minor portions are not represented in our model.

The MN distributions of the axial muscles multifidi and longissimus dorsi showed a sparse population between L_6 and S_1 and included some tail MNs at the caudal end (Vanderhorst & Holstege, 1997). The only EMG recordings from these two muscle groups we could find in the literature were made at L_4 level in the normal cat (Carlson *et al.*, 1979) or at L_4 and L_7 levels in the decerebrate or spinal cat (Zomlefer *et al.*, 1984). In the normal cats, for locomotion at about 1 m/s these muscles showed biphasic, low-level EMG in both swing and stance phases of the step cycle. In the decerebrate cats, for locomotion slower than 0.7 - 0.8 m/s, there was virtually no detectable EMG signal in the vertebral muscles (Zomlefer *et al.*, 1984). To gain some perspective on this issue, we obtained EMG recordings from the longissimus muscle between the L_5 and L_6 spinous processes in one cat with the use of percutaneous electrodes. The EMG pattern was low level and fairly constant over the step cycle, which agrees with the above observations. In light of the incomplete nature of the data on vertebral muscles, it did not seem appropriate to include these in the analysis. Given their small numbers and apparently low level of recruitment in slow stepping, their omission is unlikely to affect our main conclusions, though it would of course be desirable to confirm this if and when more EMG and anatomical data become available.



Figure 4.4. Modelled spinal MN activity in the lumbosacral enlargement during locomotion. The digitized MN distributions shown in Fig. 4.1 were combined with the EMG profiles shown in Fig. 3 to produce modelled activation patterns of MNs during two step cycles of forward, level locomotion (in this figure and the next, FDL was assigned the flexor (retractor) profile of Fig. 4.3). The phases of the step cycle are indicated above the 3-D plot. Each step cycle was divided into 100 time intervals corresponding to about 7 ms for slow walking. The plot shows the time course of activity of MNs in 30 transverse slices of the cord (level 0 most rostral, level 100 most caudal, each slice corresponds to \sim 3 levels or \sim 1mm (Vanderhorst & Holstege, 1997)). Note the dominant peaks of activity in the caudal slices in the stance phases and the relatively small rostral peaks during the swing phases. Note too the rapid migration of activity in the caudal half of the enlargement.

The main focus of this study, the spatiotemporal pattern of the MN activity during locomotion, is illustrated in Fig. 4.4. The plot shows the distribution of the active MNs along the rostrocaudal axis in two full step cycles. As described in Methods, the step cycle was divided into 100 equal intervals and for each interval the normalized EMG level of each muscle in Fig. 4.3 was used to compute the number of active MNs per slice (a slice corresponds to approximately 1 mm of spinal cord or 3.3 normalized levels (Vanderhorst & Holstege, 1997)). Figure 4.4 reveals a large imbalance in MN activity in swing and stance phases, the latter associated with the large number of extensor MNs in the caudal half of the enlargement. Note the relatively small size of the rostral peak of activity centered between L_4 and L_5 in the swing phase.

The movie animation of Fig. 4.4 (which may be obtained in digital form from the authors) created the impression of a rostrocaudal oscillation during the step cycle. This was confirmed by computing and plotting the rostrocaudal movement of the center of activity in a plan view of the same data (Fig. 4.5A). The locus of the center of MN activity (open circles) was calculated in equal 1% increments of the cycle period, (corresponding to 7-ms intervals for a 700-ms cycle period typical of medium-speed locomotion). The distance between the open circles is proportional to the velocity of the center of activity; the velocity is maximal at the transition from swing to stance. During the stance phase there is a period in which the center of activity dwells in the region about level 70. Note that the vertical scales in the panels of Fig. 4.4 and 4.5 vary because we normalized to the peak number of active MNs per slice in each case so as to use the full range of the color spectrum.



Figure 4.5. Variations in assumed activation patterns to assess parametric sensitivity. A. graded recruitment reaching a maximum of 100% in all MN pools. This plot is a plan view of the plot in Fig. 4.4. The positions of the center of mass of net MN activity were calculated at each time increment of the step cycle and displayed as open circles. The migration of these centers of activity was most rapid at the swing-stance transition. At the stance-swing transition the center of activation moves caudally (arrow a), then rostrally again. This caudal detour is associated with activity in BFp, gracilis, SMp and ST (arrow b) and FDL (arrow c). B. All-or-nothing recruitment: the EMG profiles of Fig. 4.3 were converted to step functions such that all non-zero values were assigned the value 100%. These step functions were then used to modulate the numbers of active MNs in a given pool between zero and 100%. This resulted in activity plots that had similar general features to those in A but transitions were more abrupt and the spatiotemporal migration of activity lost some of its subtleties. The caudal detour at the stance-swing transition was present, though attenuated. C. graded recruitment reaching a maximum of 15% in flexor and retractor MN pools and 40% in extensor MN pools. This is probably the case that most closely mimics real-life locomotion. (retractor profile used for FDL). D. graded recruitment reaching a maximum of 40% in flexor and retractor MN pools and 15% in extensor MN pools. The general features of the rostrocaudal movement of the center of activity shown in A were preserved in B and C. The brief caudal detours of activity at the stance-swing transition are attenuated in C and exaggerated in D. Note that because the total number of active MNs varied according to the assumptions, the colour scales were adjusted to preserve contrast and detail in each plot.

The large stance phase peak centered between L_6 and L_7 in Fig. 4.4 is followed by a smaller peak of activation in the L₇ segment associated with activity of FDL, BFp, gracilis, SMp and ST MN pools at the transition from stance to swing. This peak also shows up clearly in the plan view in Figure 4.5A. For a brief instant in this phase of the step cycle this is virtually the only part of the whole lumbosacral enlargement with active MNs. The transitions from swing to stance and retraction to swing are abrupt events, in which the focus of activity jumps from one end of the enlargement to the other: the computed center of activity moves between levels 36 and 74, i.e., 38 levels, equivalent to 10.9 mm in the average-sized enlargement. As the centers of activation of the most proximal and distal MN pools (Psoas and BFp) are at 4.3 and 25.8 mm respectively in the average enlargement, the maximal possible displacement of the center of activity is on average 21.5mm, i.e., 75 levels. Thus the center of activity moves 51% of its maximal possible range. The swing-stance transition is a little faster than the transition from retraction to stance: the center of activity reaches a peak velocity of 361 mm/s in the swing-stance transition and 309 mm/s during the retraction-swing transition. The figure also shows up a transient spread of activity rostrally to slice 11-15 about two-thirds of the way through the stance phase. This transient is associated with EMG activity of VL and RF muscles. Like VL, RF extends the knee but it also flexes the hip. Its activation is followed immediately by that of its antagonists, BFp, gracilis, SMp and ST, with an associated caudal shift of the center of activation in the spinal cord.

The EMG profiles on which Fig. 4.4 and 4.5A were based were derived from many studies and their construction involved qualitative judgments and approximations. How sensitive are such data to parametric variations in the EMG profiles? Three parametric variations are illustrated in Fig. 4.5, B - D. In Fig. 4.5B we eliminated incremental modulations in the amplitudes of EMG profiles and used only the timing information (i.e., the profiles of Fig. 4.3 were converted to all-or-nothing step functions such that all non-zero values were assigned the value 100%). As may be seen, the main effect of this was to reduce the net displacements of activity and to create more abrupt changes at phase transitions. The caudal detour of activity at the stance-swing transition was reduced. However the main conclusions drawn from Fig. 4.4 and 4.5A still held.

In Fig. 4.5, C and D we assessed the sensitivity of the plots in Fig. 4.4 and 4.5A to variations in the relative level of MN recruitment in flexor, extensor and retractor muscles. Fig. 4.5C shows the data of Fig. 4.4 recalculated on the assumption that extensor MNs were recruited to a maximum of 40% at peak EMG and flexor and retractor MNs were recruited to a maximum of 15% (40% and 15% were chosen on the basis of data comparing locomotion with more forceful movements such as paw-shakes in which recruitment approaches 100% (Chanaud *et al.*, 1991)). These values were switched in Fig. 4.5D: flexor and retractor MNs were assumed to reach 40% recruitment at peak EMG and extensor MNs were assumed to reach a maximum of 15% recruitment. In all cases the general features of the rostrocaudal movement of the center of activity shown in Fig. 4.5A are preserved.

In a separate analysis, we replotted Fig. 4.5A on the assumption that FDL had the extensor activity profile shown in Fig. 4.3 (*left*). This was performed in case the inclusion of FDL as a retractor may have given undue weight to the caudal detour of activity at the stance-swing transition. The plot obtained was hard to distinguish from that of Fig. 4.5A. We conclude that the stance-swing transition does indeed represent a distinct phase of migration of MN activity associated with retraction.

4.4 Discussion

In this study two sets of data were brought together from the literature to generate a spatiotemporal model of spinal MN activation during the locomotor step cycle. Several interesting 126

aspects of the functional anatomy of the spinal cord and the migration of MN activity during locomotion emerged. Though some of these were known in vague terms or could have been deduced from the literature, the model presents them in a precise form.

First, it is important to identify the assumptions and limitations in our analysis. First, the anatomical study on which the work depends (Vanderhorst & Holstege, 1997) did not distinguish between alpha and gamma MNs, so the number of alpha MNs in a given pool is probably overestimated. We assume that the same bias occurred in all the MN pools and so although the absolute numbers of alpha MNs is overestimated, the data provides an accurate picture of *relative* numbers of alpha MNs across pools.

Second, by making the number of active MNs in the spinal cord proportional to the amplitude of the EMG profile of a muscle, we tacitly assumed that EMG level is linearly related to net MUAP rate (and by extension to the net firing rate of corresponding spinal MNs). This is an important issue because if the relationship were highly nonlinear, the MN activity shown in the model could be significantly distorted. In an early human study, MU firing rates were found to increase with muscle force over the whole physiological range (Bigland & Lippold, 1954). In separate studies muscle force was shown to be fairly linearly related to EMG (Basmajian & De Luca, 1985). Taking these results together, it is safe to infer that MU firing rates increase with increasing EMG, but no firm conclusions can be made regarding the linearity of the relationship. In chronic recordings from single MNs during treadmill locomotion in cats, (Hoffer *et al.*, 1987) found that MN firing rates in most muscles were linearly related to the corresponding mass EMG, at least over the range of gait velocities studied.

The relationship between EMG and net MN firing rate has also been analyzed in two theoretical studies. In the first (Person & Libkind, 1970) EMG was modeled as the sum of triangular

127

MUAP waveforms. The integrated EMG was roughly proportional to the square root of net MUAP rate when MUAPs were assumed to be asynchronous. The relationship became more linear with increasing synchronization. Rate modulation of MUAPs was neglected in this study, but in a more recent simulation using large numbers of realistic MUAP waveforms derived from single unit EMG recordings (Day & Hulliger, 2001), this and other more physiological activation patterns were modeled. In most cases EMG was more linearly related to the number of MUAPs than in the Person study, e.g., in the range 0-50% MU recruitment, corresponding to medium-speed walking (Walmsley *et al.*, 1978), mean EMG increased fairly linearly with the net MU firing rate (deviations from linearity <10%). Taking all this evidence together, the assumption of a linear relationship between EMG and net MUAP rate, though it is a simplification, is unlikely to produce major distortions in the estimates of net activity of MNs in the simulation.

Third, our EMG profiles were normalized, so the assumption implicit in Figs. 4.4 and 4.5A was that MN recruitment reaches the same maximal level in all muscles. This is also an oversimplification, and so we added two simulations in which the relative recruitment in flexors and extensors was varied. These parametric variations showed that although some of the subtleties of the migration of activity were either attenuated or exaggerated, overall, the main features of spatiotemporal migration of activity were present in all the tests. The modeling would of course be better focused if it were possible to record from a large number of muscles in the same animal, with similarly placed electrodes and identical signal processing.

Finally, some MN pools were not included in our analysis, either because EMG profiles were unavailable or because the anatomical data did not include them in enough detail. Generally these MN pools were much smaller than those we did include, the only notable exceptions being the large paravertebral muscles. Further data would be needed to fill in these gaps. However we do not think that their inclusion would change our basic conclusions: first, the paravertebral muscles and their MN pools are distributed evenly over the whole lumbosacral region, so their activation profiles would add uniformly to those of the other pools and not significantly change the rostrocaudal migration of the center of activation. Second, back muscles are in any case not very active electromyographically during medium-paced locomotion at speeds less than 0.8 m/s in the cat (Zomlefer *et al.*, 1984).

Regarding our results, the first remarkable feature that emerges is the imbalance between the numbers of flexor and extensor MNs in the cat spinal cord. This leads to an imbalance in net MN activity during swing and stance phases, with a dominant seat of activity at a surprisingly caudal location (levels 80-85). If anything our simplifying assumption that all MN pools reach the same maximal recruitment level underestimated the imbalance, as flexors such as TA and EDL are not strongly activated during swing phase compared to their activity in powerful voluntary withdrawal movements (Prochazka, 1986), whereas MN recruitment in soleus, for example, may be virtually complete in locomotion(Walmsley *et al.*, 1978). The actual recruitment balance during mediumpaced locomotion is therefore probably most closely represented in Fig. 4.5C, where it is assumed that extensors are recruited to a maximum of 40% and flexors to a maximum of 15%.

The preponderance of extensor MNs in both numbers and activity has a bearing on ideas about the structure of the locomotor CPG. Though Graham Brown probably did not mean his term "half-center" to be taken literally, the term does imply an equal weighting of flexor and extensor subdivisions (Brown, 1911). The familiar schematic of the CPG, a circle bisected into flexor and extensor halves, continues this tradition. Our model does not of course give any information about the distributions of interneurons, of which the CPG may be largely composed (Tresch & Kiehn, 1999), but at least at the MN level, it puts a different slant on flexor-extensor weighting. Note that it

129

has been suggested that MNs are in fact integral elements of CPGs (Marder, 1991; O'Donovan *et al.*, 1998). Furthermore, recent studies have shown that interneurons and MNs in the same spinal segment share the same phase of activity in the step cycle (Tresch & Kiehn, 1999), suggesting that the foci of MN activity in our model may also describe the activity of many associated interneurons, but of course not necessarily all. Finally one should remember that MN firing rate profiles do not necessarily correspond completely to the time course of their locomotor drive potentials, e.g., some MNs may depolarize in two phases of the locomotor cycle but only fire action potentials in one phase (Perret & Cabelguen, 1980).

The next feature to emerge is the rostrocaudal oscillation of foci of activity during the step cycle. As MN pools innervating muscles normally thought of as flexors (including BFp and ST) are distributed through the whole length of the lumbosacral enlargement, the alternation between a caudal focus of MN activity in the stance phase and a rostral focus in late swing was not a given. Our first impressions from the movie were of a wave of activation propagating smoothly back and forth along the spinal cord. Initially we felt that this was supported by the smoothness of the trajectories of the centers of activation we published in abstract form (Yakovenko *et al.*, 2000). Propagation of activity along the neuraxis has been implicated in the orderly sequencing of activation of segmented musculature in different species (Orlovsky *et al.*, 1999). However, the rostrocaudal transitions shown in Figs. 4.4 and 4.5 are in fact quite abrupt, so if smooth propagation from one set of neurons to the next is involved, it is rapid. Our model seems more consistent with the idea of abrupt switching between three groups of MNs: extensor MNs active during late swing and most of stance, retractor MNs being active in late stance and early swing and ankle and hip flexor MNs being active in swing only.

In the intact animal, stance-swing and swing-stance transitions in normal over-ground locomotion are under sensory control: when certain sensory conditions are met, phase switching is triggered (Prochazka, 1993). It will therefore be interesting to compare the present results with those obtained when sensory input is absent (e.g., in fictive locomotion). In Results we pointed out a possible link between MN distribution and biomechanical action: in the pendent limb, rostral MNs would evoke forward movements of the foot and caudal MNs would evoke backward movements of the foot. However in real-life locomotion, movements are affected by weight-bearing so the activity of some muscles no longer fits this simple dichotomy. The spatiotemporal migration of activity in the spinal cord may then differ significantly from that in unloaded movements such as air-stepping and paw-shakes.

Regarding BFp, gracilis, SMp, ST and FDL, these muscles extend the hip and flex the knee and toes (gracilis also has an adductor role). Their main EMG bursts occur when the foot lifts off at the end of the stance phase (Engberg & Lundberg, 1969; Halbertsma, 1983), though FDL may also exhibit some activity earlier in stance (O'Donovan *et al.*, 1982; Loeb, 1993). In mid-stance the moment arms of the posterior portions of BFp, gracilis, SMp and ST about the knee are larger than those about the hip (Goslow *et al.*, 1973). But in late stance and early swing, if the knee becomes very extended, the moment arms about hip and knee become similar. The biomechanical action at foot lift-off is therefore a backward and upward movement of the foot and toes with respect to the hip, this motion being crucial for ground clearance. Thus BFp, gracilis, SMp, ST and FDL complete the backward movement of the foot that develops in the stance phase and initiate the upward movement that is then maintained by flexor muscles during swing. Sherrington listed gracilis, BFp and ST muscles as being engaged in the flexion reflex of decerebrate cats, yet in his Table 2 he lists BFp, FDL and SM as being inhibited along with pure extensors when the flexion reflex was re-
applied during a rebound extensor contraction (Sherrington, 1910). During reflex stepping and standing, BFp, gracilis and ST behaved as pure flexors while SM and FDL were classified as extensors. Short-latency (presumed group Ia) reflex connections measured between various hip and knee muscles in lightly anesthetized cats added to the uncertainty of classifying BFp, gracilis and ST as flexors (Eccles & Lundberg, 1958).

In view of these various anomalies, as well as the results of our spinal cord modeling we suggest that in locomotion BFp, gracilis, SMp, ST and FDL muscles be classified as *retractors*, their role being to complete the backward movement of the foot and toes with respect to the hip (i.e., to complete the extensor thrust of the stance phase) and then to lift the foot in preparation for forward swing. This is not to deny that they can act differently in other tasks: e.g., BFp is strongly activated during flexion phases of withdrawal movements of the limb (Prochazka, 1986) and gracilis has a significant adductor role (Sherrington, 1910). From the point of view of the spatiotemporal flow of activity in the spinal cord the activation of the BFp, gracilis, SMp and ST MN pools involves a small transient shift caudal from the focus of extensor activity in the stance phase. Activity then switches rapidly to the rostral flexor centers.

Regarding the hypothesis of discrete "unit" oscillators controlling hip, knee and ankle (Grillner & Wallen, 1985; Kiehn & Kjaerulff, 1998), Fig. 4.1 shows the actual anatomical positions of the three sets of MN pools that would be involved. If such unit oscillators exist and MNs are part of them (Marder, 1991; O'Donovan *et al.*, 1998), they are evidently widely spaced rostrocaudally and so they would require long and overlapping propriospinal connections. For example, if we equate the centers of the unit oscillators with the centers of mass of their respective MN pools, the two components of the hip oscillator would be separated by 61 levels (17.5 mm in an average-sized spinal cord), those of the knee oscillator by 58 levels (16.7mm) and those of the ankle oscillator by

only 20 levels (approximately 5.6 mm). The rostrocaudal order of the knee flexor and extensor halfcenters would be the reverse of that of the hip and ankle half-centers.

The approach introduced in this paper opens up a variety of interesting possibilities. Spatiotemporal patterns of spinal cord activation could be modeled and compared in various preparations (e.g., normal, decerebrate and spinal locomotion), in different types of cyclical movement (e.g., paw-shake response and scratching) and using different representations of MN activity, e.g., intracellular recording. Models of this type could be extremely useful if and when spinal cord neuroprostheses are developed for people with spinal cord injury. The locomotor system of the cat spinal cord is not the only pattern generator amenable to this form of analysis. Any system in which the CNS distribution of neurons and the activity profiles of their target organs in the periphery are reasonably well characterized could be analyzed in this way. The dynamic models and the spinal cord movie described in this paper may be obtained in digital form from the authors.

4.5 References

- ANDERSSON, O., FORSSBERG, H., GRILLNER, S. & LINDQUIST, M. (1978). Phasic gain control of the transmission in cutaneous reflex pathways to motoneurones during 'fictive' locomotion. *Brain Res* 149, 503-507.
- ARMSTRONG, D. M. & MARPLE-HORVAT, D. E. (1996). Role of the cerebellum and motor cortex in the regulation of visually controlled locomotion. *Can J Physiol Pharmacol* 74, 443-455.
- ARSHAVSKY, Y. I., DELIAGINA, T. G. & ORLOVSKY, G. N. (1997). Pattern generation. *Curr Opin Neurobiol* 7, 781-789.
- BASMAJIAN, J. V. & DE LUCA, C. J. (1985). EMG signal amplitude and force. In *Muscles Alive. Their functions Revealed by Electromyography*, pp. 187-222. Williams and Wilkins, Baltimore, MD.
- BIGLAND, B. & LIPPOLD, O. C. (1954). The relation between force, velocity and integrated electrical activity in human muscles. *J Physiol* 123, 214-224.
- BROWN, A. G. (1981). Organization in the Spinal Cord: The Anatomy and Physiology of Identified Neurons. Springer, Berlin.
- BROWN, T. G. (1911). The intrinsic factors in the act of progression in the mammal. Proc Roy Soc Lond B Biol Sci 84, 308-319.
- CARLSON, H., HALBERTSMA, J. & ZOMLEFER, M. (1979). Control of the trunk during walking in the cat. Acta Physiol Scand 105, 251-253.
- CARRASCO, D. I. & ENGLISH, A. W. (1999). Mechanical actions of compartments of the cat hamstring muscle, biceps femoris. *Prog Brain Res* 123, 397-403.
- CAZALETS, J. R., BORDE, M. & CLARAC, F. (1995). Localization and organization of the central pattern generator for hindlimb locomotion in newborn rat. *J Neurosci* 15, 4943-4951.
- CHANAUD, C. M., PRATT, C. A. & LOEB, G. E. (1991). Functionally complex muscles of the cat hindlimb.
 V. The roles of histochemical fiber-type regionalization and mechanical heterogeneity in differential muscle activation. *Exp Brain Res* 85, 300-313.
- DAY, S. J. & HULLIGER, M. (2001). Experimental simulation of cat electromyogram: evidence for algebraic summation of motor-unit action-potential trains. *J Neurophysiol* 86, 2144-2158.
- DELIAGINA, T. G., ORLOVSKY, G. N. & PAVLOVA, G. A. (1983). The capacity for generation of rhythmic oscillations is distributed in the lumbosacral spinal cord of the cat. *Exp Brain Res* 53, 81-90.

- ECCLES, R. M. & LUNDBERG, A. (1958). Integrative pattern of Ia synaptic actions on motoneurones of hip and knee muscles. *J Physiol* 144, 271-298.
- ENGBERG, I. & LUNDBERG, A. (1969). An electromyographic analysis of muscular activity in the hindlimb of the cat during unrestrained locomotion. *Acta Physiologica Scandinavica* 75, 614-630.
- FLESHMAN, J. W., LEV-TOV, A. & BURKE, R. E. (1984). Peripheral and central control of flexor digitorum longus and flexor hallucis longus motoneurons: the synaptic basis of functional diversity. *Exp Brain Res* 54, 133-149.
- GOSLOW, G. E., JR., REINKING, R. M. & STUART, D. G. (1973). The cat step cycle: hind limb joint angles and muscle lengths during unrestrained locomotion. *J Morphol* 141, 1-41.
- GRILLNER, S., PARKER, D. & EL MANIRA, A. (1998). Vertebrate locomotion--a lamprey perspective. Ann N Y Acad Sci 860, 1-18.
- GRILLNER, S. & ROSSIGNOL, S. (1978). On the initiation of the swing phase of locomotion in chronic spinal cats. *Brain Res* 146, 269-277.
- GRILLNER, S. & WALLEN, P. (1985). Central pattern generators for locomotion, with special reference to vertebrates. *Annu Rev Neurosci* 8, 233-261.
- GRILLNER, S. & ZANGGER, P. (1975). How detailed is the central pattern generation for locomotion? Brain Res 88, 367-371.
- GRILLNER, S. & ZANGGER, P. (1979). On the central generation of locomotion in the low spinal cat. Exp Brain Res 34, 241-261.
- HALBERTSMA, J. M. (1983). The stride cycle of the cat: the modelling of locomotion by computerized analysis of automatic recordings. Acta Physiologica Scandinavica. Supplementum 521, 1-75.
- HOCHMAN, S., JORDAN, L. M. & MACDONALD, J. F. (1994). N-methyl-D-aspartate receptor-mediated voltage oscillations in neurons surrounding the central canal in slices of rat spinal cord. J Neurophysiol 72, 565-577.
- HOFFER, J. A., SUGANO, N., LOEB, G. E., MARKS, W. B., O'DONOVAN, M. J. & PRATT, C. A. (1987). Cat hindlimb motoneurons during locomotion. II. Normal activity patterns. *Journal of Neurophysiology* 57, 530-553.
- JANKOWSKA, E. (1992). Interneuronal relay in spinal pathways from proprioceptors. *Prog Neurobiol* 38, 335-378.

- KIEHN, O., HARRIS-WARRICK, R. M., JORDAN, L. M., HULTBORN, H. & KUDO, N. (1998). Neuronal mechanisms for generating locomotor activity, vol. 860.
- KIEHN, O. & KJAERULFF, O. (1998). Distribution of central pattern generators for rhythmic motor outputs in the spinal cord of limbed vertebrates. *Ann NY Acad Sci* 860, 110-129.
- KREMER, E. & LEV-TOV, A. (1997). Localization of the spinal network associated with generation of hindlimb locomotion in the neonatal rat and organization of its transverse coupling system. *J Neurophysiol* 77, 1155-1170.
- LEV-TOV, A. & O'DONOVAN, M. J. (1995). Calcium imaging of motoneuron activity in the en-bloc spinal cord preparation of the neonatal rat. *J Neurophysiol* 74, 1324-1334.
- LOEB, G. E. (1993). The distal hindlimb musculature of the cat: interanimal variability of locomotor activity and cutaneous reflexes. *Exp Brain Res* 96, 125-140.
- MARCOUX, J. & ROSSIGNOL, S. (2000). Initiating or blocking locomotion in spinal cats by applying noradrenergic drugs to restricted lumbar spinal segments. *J Neurosci* 20, 8577-8585.
- MARDER, E. (1991). Modifiability of pattern generation. Curr Opin Neurobiol 1, 571-576.
- METTRIE DE LA, J. O. (1745). L'Homme-Machine (Man a Machine; and, Man a Plant). Gryphon Editions, New York.
- MUSHAHWAR, V. K., COLLINS, D. F. & PROCHAZKA, A. (2000). Spinal cord microstimulation generates functional limb movements in chronically implanted cats. *Exp Neurol* 163, 422-429.
- MUSHAHWAR, V. K. & HORCH, K. W. (1998). Selective activation and graded recruitment of functional muscle groups through spinal cord stimulation. *Ann NY Acad Sci* 860, 531-535.
- O'DONOVAN, M. J., PINTER, M. J., DUM, R. P. & BURKE, R. E. (1982). Actions of FDL and FHL muscles in intact cats: functional dissociation between anatomical synergists. *J Neurophysiol* 47, 1126-1143.
- O'DONOVAN, M. J., WENNER, P., CHUB, N., TABAK, J. & RINZEL, J. (1998). Mechanisms of spontaneous activity in the developing spinal cord and their relevance to locomotion. *Ann N Y Acad Sci* **860**, 130-141.
- ORLOVSKY, G. N., DELIAGINA, T. G. & GRILLNER, S. (1999). Neuronal Control of Locomotion. Oxford Univ. Press, Oxford, UK.
- PEARSON, K. G. (1995). Proprioceptive regulation of locomotion. Curr Opin Neurobiol 5, 786-791.
- PERRET, C. & CABELGUEN, J. M. (1980). Main characteristics of the hindlimb locomotor cycle in the decorticate cat with special reference to bifunctional muscles. *Brain Res* 187, 333-352.

- PERSON, R. S. & LIBKIND, M. S. (1970). Simulation of electromyograms showing interference patterns. *Electroencephalogr Clin Neurophysiol* 28, 625-632.
- PHILIPPSON, M. (1905). L'autonomie et la centralisation dans le systeme merveux des animaux. *Trav Lab Inst Solvay* 7, 1-208.
- PROCHAZKA, A. (1986). Proprioception during voluntary movement. Can J Physiol Pharmacol 64, 499-504.
- PROCHAZKA, A. (1993). Comparison of natural and artificial control of movement. *IEEE Trans Rehab* Eng 1, 7-16.
- PROCHAZKA, A. (1996). Proprioceptive feedback and movement regulation. In Handbook of Physiology. Exercise: Regulation and Integration of Multiple Systems, pp. 89-127. Am. Physiol. Soc., New York.
- PROCHAZKA, A. & GORASSINI, M. (1998). Ensemble firing of muscle afferents recorded during normal locomotion in cats. *J Physiol* **507** (**Pt 1**), 293-304.
- PROCHAZKA, A., TREND, P., HULLIGER, M. & VINCENT, S. (1989). Ensemble proprioceptive activity in the cat step cycle: towards a representative look-up chart. *Progress in Brain Research* 80, 61-74; discussion 57-60.
- RHO, M. J., CABANA, T. & DREW, T. (1997). Organization of the projections from the pericruciate cortex to the pontomedullary reticular formation of the cat: a quantitative retrograde tracing study. *J Comp Neurol* **388**, 228-249.
- ROMANES, G. J. (1951). The motor cell columns of the lumbo-sacral spinal cord of the cat. J Comp Neurol 94, 313-363.
- SHARRARD, W. J. (1955). The distribution of the permanent paralysis in the lower limb in poliomyelitis; a clinical and pathological study. *J Bone Joint Surg Br* **37-B**, 540-558.
- SHERRINGTON, C. S. (1892). Notes on the arrangemet of some motor fibres in the lumbo-sacral plexus. *J Physiol (Lond)* 13, 621-772.
- SHERRINGTON, C. S. (1910). Flexion-reflex of the limb, crossed extension-reflex, and reflex stepping and standing. *J Physiol (Lond)* 40, 28-121.
- SMITH, J. L., CARLSON-KUHTA, P. & TRANK, T. V. (1998). Forms of forward quadrupedal locomotion. III. A comparison of posture, hindlimb kinematics, and motor patterns for downslope and level walking. *Journal of Neurophysiology* 79, 1702-1716.

- TRANK, T. V. & SMITH, J. L. (1996). Adaptive control for backward quadrupedal walking VI. metatarsophalangeal joint dynamics and motor patterns of digit muscles. *Journal of Neurophysiology* 75, 678-679.
- TREND, P. (1987). Gain Control in Proprioceptive Pathways. In *Physiology*. University of London, London, UK.
- TRESCH, M. C. & KIEHN, O. (1999). Coding of locomotor phase in populations of neurons in rostral and caudal segments of the neonatal rat lumbar spinal cord. *J Neurophysiol* 82, 3563-3574.
- VANDERHORST, V. G. & HOLSTEGE, G. (1997). Organization of lumbosacral motoneuronal cell groups innervating hindlimb, pelvic floor, and axial muscles in the cat. *J Comp Neurol* **382**, 46-76.
- WALMSLEY, B., HODGSON, J. A. & BURKE, R. E. (1978). Forces produced by medial gastrocnemius and soleus muscles during locomotion in freely moving cats. *J Neurophysiol* 41, 1203-1216.
- YAKOVENKO, S., MUSHAHWAR, V. K., VANDERHORST, V. G., HOLSTEGE, G. & PROCHAZKA, A. (2000). Locus of activation of spinal motoneuron pools in the cat step cycle. *J Physiol (Lond)* **525P**, 42P.
- ZHANG, W., POMBAL, M. A., EL MANIRA, A. & GRILLNER, S. (1996). Rostrocaudal distribution of 5-HT innervation in the lamprey spinal cord and differential effects of 5-HT on fictive locomotion. J Comp Neurol 374, 278-290.
- ZOMLEFER, M. R., PROVENCHER, J., BLANCHETTE, G. & ROSSIGNOL, S. (1984). Electromyographic study of lumbar back muscles during locomotion in acute high decerebrate and in low spinal cats. *Brain Res* 290, 249-260.

CHAPTER 5

The adjustments of phase and cycle durations in MLR-induced fictive cat locomotion. Implications for organization of the pattern generating circuitry.

5.1 Introduction

The study of locomotion dates back to La Mettrie's observations of rhythm generation in animals with cervical spinalization (Mettrie, 1745) and the first cinematic recordings of Muybridge (Muybridge, 1887). Apart from the neurophysiological interest, a detailed knowledge about the control of locomotion is useful in devising medical strategies to relieve locomotor disabilities in people due to trauma or debilitating diseases. The work of T. Graham Brown and Sir Charles Sherrington in the early 20th century was very influential in establishing the basic neural circuitry involved in the generation of locomotor activity. Even in the absence of supraspinal control the locomotor rhythm can be generated by the spinal cord and is subject to modification by sensory feedback. This spinal pattern generating apparatus with at least two antagonistically organized groups of neurons named half-centers was first called "the intrinsic factor" (Brown, 1911) and was later renamed the "central pattern generator" (CPG) (Grillner, 1975). A number of experiments have localized and studid the locomotor pattern generating circuitry in reduced animal preparations (Deliagina et al., 1983; Ho & O'Donovan, 1993; Kjaerulff & Kiehn, 1996; Arshavsky et al., 1997; Cowley & Schmidt, 1997; Kremer & Lev-Tov, 1997). In such studies, the locomotor rhythm was usually evoked by the application of pharmacological agents such as L-DOPA (L-3,4dihydroxy-phenylalanine), mechanical stimulation of the perineum or electrical stimulation of brainstem structures or dorsal roots. Studies have shown that the control of timing and amplitude of motoneuron discharge is mediated not only by the descending pathways (Shik et al., 1966; Shik & Orlovsky, 1976; Armstrong & Drew, 1984; Drew, 1988; Beloozerova & Sirota, 1993), but also by movement-evoked sensory feedback (Conway et al., 1987; Guertin et al., 1995; Hiebert et al., 1996; Prochazka, 1996; Whelan & Pearson, 1997; Lam & Pearson, 2001). Though the location and neural structure of the locomotor network in vertebrates are poorly understood, some details of neural 140

connectivity, membrane properties and correlation with function are available for invertebrate species (Kristan & Weeks, 1983; Kopell, 1988; Marder & Bucher, 2001; Cymbalyuk *et al.*, 2002).

Attempts have been made to deduce the general organization of the CPG on the basis of the mutability of the patterns of muscle activation during different locomotor behaviors, e.g. comparing forward, backward, uphill and downhill walking (Grillner, 1981; Buford & Smith, 1993; Trank et al., 1996; Smith et al., 1998). In these studies, Brown's simple model of half-center oscillators is usually subdivided further into oscillators controlling synergies of muscles around each joint. Recently, a spatio-temporal analysis of the hindlimb motoneuron activity underlying cat locomotion was performed, which combined information about EMG activity in hindlimb muscles with the known distribution of motoneuron pools within the lumbosacral spinal cord (Yakovenko et al., 2002). This revealed a rostro-caudal oscillation of motoneuron activity in the spinal cord, which has intriguing implications for the organization of the controlling pattern generating circuitry. If some or all of the interneuronal network of the CPG is distributed and co-localized with corresponding motoneurons (Ho & O'Donovan, 1993; Kjaerulff & Kiehn, 1996; Raastad & Kiehn, 2000; Tresch & Kiehn, 2000) the locus of activity of the motoneurons may also represent that of the entire controlling CPG network. The results of the analysis of motoneuron activity during normal gait are consistent with the hypothesis that flexor and extensor activity is spatio-temporally separated in the step cycle and that phase transitions are switch-like and not wave-like, as suggested elsewhere (Bonnot et al., 2002; Kaske et al., 2003; Patel et al., 2003).

In a separate modeling study we became aware that the timing of transitions between the locomotor phases is important for adaptation of the CPG rhythm to relevant behavioral demands, e.g. an increase in the speed of locomotion. During normal overground gait an increase in the speed of locomotion is usually accompanied by a decrease of the step cycle duration, mostly due to

the shortening of the stance phase. The duration of the swing phase remains relatively constant (Goslow *et al.*, 1973; Halbertsma, 1983). The same modulation strategy of flexor and extensor phases was reported 1) in the locomotor activity of spinalized cats elicited by perineal stimulation and the administration of clonidine (Forssberg & Grillner, 1973; Pearson & Rossignol, 1991), and 2) in fictive locomotion of decerebrate cats evoked either by stimulation of the mesencephalic locomotor region (MLR) (Jordan, 1998), intravenous administration of L-DOPA (Pearson & Rossignol, 1991; Leblond & Gossard, 1997) or occurring spontaneously (Dubuc *et al.*, 1985). These results suggest that the CPG is hard-wired to produce adjustments of step cycle duration by modulating extensor, rather than flexor bursts. Here we report a novel observation that during fictive MLR-induced locomotion, cycle duration is modulated by mainly adjusting the duration of flexor bursts.

5.2 Methods

5.2.1 Surgical procedures

All data reported in this paper originated from experiments done for other reasons in David McCrea's laboratory, University of Manitoba (Gosgnach *et al.*, 2000; Quevedo *et al.*, 2000). All surgical and experimental protocols were in accordance with the guidelines set out by the Canadian Council for Animal Care. Experiments were performed on 31 cats (weight 2.2 - 3.2 kg). The animals were anesthetized with halothane delivered in a mixture of 30% oxygen and 70% nitrous oxide for the duration of the surgery. Surgical procedures consisted of a tracheotomy, cannulation of arteries and veins, dissections of various hindlimb nerves and a craniotomy. Mechanical, precollicular-postmamillary decerebration was performed by removing all the tissue rostral to the transection. This procedure makes animals insentient (legally dead) and allows discontinuation of anesthetic. The neuromuscular blocker gallamine triethiodide (Flaxedil, 2-3 mg kg-1 h-1) was used to abolish movement and the animals were artificially ventilated. Fictive locomotion was elicited by unilateral or bilateral monopolar electrical stimulation of the mesencephalic locomotor region (MLR, 30-200 μ A, 0.5-1 ms pulses, 7-30 Hz) with tungsten microelectrodes. Selected hindlimb nerves were placed on conventional bipolar hook electrodes for recording. The following nerves were included: biceps femoris posterior (BFp) or sometimes BFp and semitendinosus together (PBST), extensor digitorum longus (EDL), flexor digitorum longus (FDL), flexor hallucis longus (FHL), FDL and FHL (FDHL), gastrocnemius and soleus (LGS), peroneus longus (PerL), plantaris (Plan), rectus femoris (RF), sartorius (Sart), semimembranosus and biceps femoris anterior (SMAB), tibialis anterior (TA), and vasti. In some experiments vasti and RF were mounted on the same recording electrode.

Up to 12 nerves were recorded simultaneously in each cat. Since these preparations were intended for different studies, a laminectomy was performed usually from the L4 to the S1 regions in some animals but in others it was more extensive including thoracic and cervical segments. A lethal injection of pentobarbital anesthetic was administered at the end of the experiment.

5.2.2 Data recording and analysis

Electroneurogram (ENG) activity in hindlimb muscle nerves listed above was filtered (30 Hz - 3 KHz), rectified and digitized at 500 Hz. The smoothed waveforms were then analyzed to determine onsets and terminations of bursts using an adaptive threshold crossing method described below. Recordings with a signal-to-noise ratio less than 5 were rejected. Cycle period was determined as the time between two onsets of the ENG envelope. We intentionally restricted the

analysis to cycles with periods between 0.5 and 1.5 sec corresponding to speeds of overground locomotion between 0.2 and 1.2 m/sec (Halbertsma, 1983). Regression lines were fitted to scatter plots of burst duration and cycle period for individual nerves. One-way analysis of variance (ANOVA) and multiple comparison of regression line slopes with Tukey's honestly significant difference criterion was performed to determine significant differences (α =0.05).

5.2.3 Adaptive threshold crossing method

Consider a portion of a noisy, time-varying signal. Now consider a horizontal line representing a threshold level, moving upward through the signal (Fig. 5.1A). Within the noise band, there are many cross-overs of the noisy signal and the threshold line. This number reaches a peak somewhere within the noise band. As the threshold line moves above the noise band and into the region of the "real" signal, the number of cross-overs drops and then remains fairly constant. We called the minimal threshold above the noise as an optimal threshold (dotted line c in Fig.5.2). Finally, as the line moves up into the region of signal peaks and beyond, the number of crossovers first increases and then declines to zero. Thus, we can use a simple algorithm to detect the optimal threshold and identify periods of noise from periods of signal. Then we can quantify the mean and the standard deviation of the noise and signal periods to calculate the signal-to-noise ratio. If the quality of the recording is satisfactory, we can identify onsets and terminations of bursts by calculating the sign of the slope at the intersection points (positive slope corresponds to onset, negative slope corresponds to termination). Finally, the mean and the standard deviation of the noise periods can be used to adjust the time of the onsets and the terminations to the smallest values of the signal statistically different from the noise. The detection quality of the method was similar to that of manual selection with the obvious advantages of automation and objectivity.



Figure 5.1. An adaptive optimal threshold method for detection of onsets and terminations of bursts. Horizontal and vertical *dotted* lines represent various threshold levels (*c* is the optimal threshold) *A*. A section of ENG recording analyzed with the optimal threshold method. Upward and downward triangles correspond respectively to the adjusted positions of onsets and terminations for each burst (see Methods for details). *B*. The number of intersections (*solid* line) and its differential (*dashed* line) are plotted as functions of the threshold level. The optimal threshold level (*c*), which corresponds to the minimum level above noise, can be found as a minimum threshold following a positive and then a negative slope of the function of intersections (*solid* line).

5.3 Results

As discussed above previous studies of the variations of swing and stance durations during locomotion have demonstrated that changes in cycle period are due less to changes in swing phase and corresponding flexor bursts than to the duration of stance phase and extensor bursts. Our study originated from a chance observation that this did not always hold in MLR-induced fictive locomotion.

In any given cat, we found that either the extensor bursts or the flexor bursts occupied more than 50% of the step cycle period in nearly all cycles (i.e. a given cat showed extensordominated or flexor-dominated behavior). Furthermore, the *amount of variation* in burst duration *also* seemed to depend on the cycle occupied. We therefore initially decided to divide the data into two groups, corresponding to extensor-dominated and flexor-dominated behavior.

Figure 5.1A shows a typical recording of nerve activity during fictive locomotion evoked by the stimulation of MLR in a paralyzed preparation. Note that the burst duration in this recording dominated the duration of the locomotor cycle, it occupied more than 50% of the cycle duration. During overground locomotion in the normal cat this relationship is characteristic of extensorrelated activity. However, in this case the depicted activity is that of a flexor nerve (Sart). This intriguing relationship prompted us to examine the phase vs. cycle relationship in this preparation. Using the adaptive threshold crossing analysis described in *Methods* we were able to identify onsets and terminations of bursts during fictive locomotion in 31 animals. The locomotor patterns were divided into groups based on the percentage of the cycle occupied by bursts of nerves defined as flexors: Sart, TA, EDL, PerL and the following as extensors: GS, FHL, Plan, SMAB. This is in accord with the generally accepted classification of nerves, based on the predominant biomechanical



Figure 5.2. The relationship between phase and cycle duration for the hindlimb nerve activity when the cycle duration is dominated by either flexors (A) or extensors (B). The *dashed* line represents a slope of 0.5, which corresponds to the burst duration equal to 50% of the cycle duration. Thick line is the median of the sample data.

A. flexor-dominated







Figure 5.3. The relationship between phase and cycle duration for flexor (solid line), extensor (dashed line) and retractor (dotted line) hindlimb nerve activity when the step cycle duration is dominated by either flexors (A) or extensors (B).

action of the corresponding muscles in flexor and crossed extensor reflexes (Sherrington, 1910; Eccles & Lundbert, 1958) and the spatio-temporal activation of the corresponding motoneurons in the spinal cord during overground locomotion (Yakovenko *et al.*, 2002).

We found that our sample contained locomotor patterns dominated by flexor activity in 22 experiments and by extensor activity in 9 experiments. Figure 5.2 shows the lines of best fit in the relationship between burst duration and cycle duration of various hindlimb nerves. Notice that during the flexor-dominated fictive locomotion in Figure 5.2A the lines of best fit of flexor activity are mostly above the dashed line. The dashed lines represent a slope of 0.5 or 50% of cycle duration, while those of extensor activity are below this line. The opposite relationship is demonstrated in Figure 5.2B. Notice that the slopes of flexor lines of best fit in Figure 5.2A are steeper than those in Figure 5.2B; the opposite is true for the extensors. The significance of this is discussed later.

Figure 5.3 shows the mean lines of best fit in Figure 5.2 now combined into *three* groups: flexors, extensors and retractors. The latter group contains the data from FDL and PBST nerves, which showed more variability than other nerves (Fig.5.4), and have elsewhere been distinguished as a separate "retractor" group based on their spatio-temporal activity during normal gait and their biomechanical action (Yakovenko *et al.*, 2002). The slopes of the extensor and flexor lines of best fit are similar to those found in normal cats during overground gait (see Fig.6 in Halbertsma, 1983). In contrast, in the flexor-dominated group in Figure 5.3A, flexor slopes were greater than extensor slopes. Taking Fig.5.3A and 5.3B together, a more general statement can be made. The variation in burst duration depends on the percentage of the cycle occupied. Thus, extensors vary the most if they are dominant; flexors vary the most if they are dominant. During flexor-dominated locomotion



Figure 5.4. The result of a one-way ANOVA of the slopes of the relationship between phase and cycle duration of hindlimb nerve activity when the step cycle duration was dominated by either flexors (A) or extensors (B). The "whiskers" show the 95% confidence interval about the median (*circle*) of a sample.



Figure 5.5. The result of the one-way ANOVA and multiple comparisons of the slopes of the phase modulation with Tukey's honestly significant difference criterion in flexor (A) and extensor (B) dominated fictive locomotor activity. The "whiskers" show the 95% confidence interval about the median of a sample.

(Fig. 5.3A) the mean slope of flexors (solid line) was 0.69, while that of the extensors was 0.17. Though the burst duration of the retractor (bifunctional) group was less than 50% of the step cycle for both the flexor and the extensor -dominated cycles, the slope of the retractors was close to that of extensors during flexor-dominated locomotion (0.36) and it was closer to flexors during extensor-dominated locomotion (0.12) (Fig. 5.3B). This result is in agreement with a study (Perret & Cabelguen, 1980), which demonstrated that activity of this group was shaped by converging inputs that excite flexors and extensors.

Fig. 5.4 and 5.5 deal with the statistical analysis. Figure 5.4 shows the results of a one-way ANOVA of the slopes of the phase adjustments during flexor and extensor -dominated fictive locomotion. It is worth noting the large variability in the slopes of FDL and PBST nerves, which again could be an indication of the convergent influences from the flexor and the extensor pattern generating networks. The mean RF slope was not significantly different from those of flexors or extensors. This might indicate a mixed functional role, or it may simply have been due to the RF nerve having been mounted together with the nerve of vasti in some experiments. Figure 5.5 shows a comparison of the means in the groups of Fig.5.3. In flexor-dominated (Fig. 5.5A) and extensor-dominated (Fig.5.5B) fictive locomotion, the slopes of the flexor and extensor groups of nerves were significantly different. The slopes of the retractor and flexor groups were significantly different from those of other during flexor-dominated locomotion. During extensor-dominated locomotion, the slopes of the retractors were significantly different from those of the extensors.

5.4 Discussion

During normal overground level locomotion, adjustments of step cycle duration are mainly due to variations in stance duration (Goslow *et al.*, 1973; Halbertsma, 1983). Several features of

152

overground locomotion show a bias toward extensor-related activity (Fig. 5.6A). During slow walking the duration of stance dominates the step cycle period (Prochazka *et al.*, 1989; Yakovenko *et al.*, 2002). Because of the anti-gravity role of the extensor muscles, they exert proportionally higher forces than flexors during locomotion (Chanaud *et al.*, 1991). In addition, sensory pathways reinforce extensor activity during stance (Duysens & Pearson, 1980; Whelan *et al.*, 1995). Does this imply that the pattern generating circuitry is hard-wired to favor extensor-related activity?

We addressed this question by analyzing the variability of nerve activity during MLRinduced fictive locomotion. When the duration of the extensor phase dominated the duration of the step cycle (>50%) in fictive locomotion, the phase adjustments during spontaneous changes in cycle duration were similar to those during normal gait. The novel finding of this study is that the reverse behavior was also observed during MLR-induced fictive locomotion. When cycle duration was dominated by flexor activity, the phase adjustments were mainly due to variations in flexor phase duration. Grillner and Zangger noted similar behavior in one of the fourteen spinalized animals during DOPA-induced fictive locomotor activity (see Fig.6 in Grillner & Zangger, 1979). This result suggests that the locomotor pattern generating networks can be equally biased towards either extensor or flexor activity. This agrees with Brown's prediction that the locomotor pattern generating networks for each limb contain at least a "pair of antagonistic opposites", or analogous antagonistic half-centers (Brown, 1911). A. normal overground locomotion



B. fictive "flexor-biased" locomotion



Figure 5.6. Simple schematic of the locomotor pattern generator. A. During normal overground locomotion, activity of extensor half-center is biased by descending and sensory pathways (*thick lines*) to cope with weight-bearing. B. During MLR-induced fictive locomotion, motion-related feedback is lost and descending pathways are possibly biased towards flexors.

We also found a relationship between phase duration and the variation in burst duration during spontaneous changes in cycle period. From this we propose the following rule: the variation in burst duration depends on the percent of the cycle occupied. This rule describes phase adjustments during normal overground locomotion (Goslow *et al.*, 1973; Halbertsma, 1983), spontaneous locomotion in decorticate preparation (Grillner & Dubuc, 1988), locomotion induced either by perineal stimulation (Pearson & Rossignol, 1991) or by noradrenergic agonists in chronically spinalized preparation (Grillner & Zangger, 1979).

Clearly, the speed of overground gait can be adjusted using two distinct strategies, by changing either the duration of the step cycle or the stride length. From Halbertsma's regression Eq. 1-4 (Halbertsma, 1983) for slow and fast walking, it is clear that the step cycle duration strategy contributes more than the stride length strategy to the adjustment of speed during overground locomotion. This agrees with the finding that an increase of the stride length leads to a higher loss of mechanical energy due to the collisions with the support surface (Donelan *et al.*, 2002). The locomotor system may try to minimize this energy loss. For this reason the step cycle duration strategy may be the preferred solution.

Why does the nervous system avoid adjusting the duration of swing to the same degree as that of stance during normal gait (Goslow *et al.*, 1973; Halbertsma, 1983)? Perhaps, the interactions of the locomotor system with the ground impose mechanical constraints that prevent the use of flexor phase adjustments of speed. To illustrate this suggestion, consider the situation in which an increase in flexor muscle activity and swing duration allows a more forward placement of the foot relative to the body. Then, more kinetic energy is lost due to ground contact and still more work is required to bring the centre of mass over the more distant point of support. Without an additional increase in extensor muscle thrust and / or prolongation of the stance phase the body would lose

all its forward momentum and stall. On the other hand, an increase in the stride length due to increased extensor thrust imparts a higher mechanical energy to the body, which is sufficient to counteract collision losses and allows stable locomotion at a higher speed. The locomotor system is therefore predisposed in favor of extensor phase adjustments strategy during overground locomotion. The alternative flexor phase adjustment strategy might be observed only when interactions with the ground are abolished. This hypothesis was supported by our results and by the increased flexor phase adjustments during air-stepping (Smith *et al.*, 1986).

We propose that the pattern generated by the locomotor CPG depends on the distribution of excitation between flexor and extensor half-centers, which are analogous and antagonistic. In this model, the level of excitation of each half-center can then be modified by inputs from descending and sensory pathways. A shift in the excitatory bias to one of the half-centers may lead to an increase in the duration of the corresponding phase of the cycle and a decrease in the antagonistic phase through reciprocal neural interactions (Cymbalyuk *et al.*, 2002). In our study, 22 out of 31 cats were predisposed to produce flexor dominated fictive locomotion, which may be explained by inputs to the flexor half-center through the descending pathways evoked by stimulation of MLR in decerebrated cats (Orlovsky, 1970; Shimamura *et al.*, 1982) and the absence of motion-related feedback, which would normally provide excitatory input to the extensor halfcenter (Fig. 5.6B).

The proposed model can describe a number of reported observations of phase modulation. Unloading in stance decreases stance phase duration in cats (Duysens & Pearson, 1980) and human infants (Pang & Yang, 2000). Electrical stimulation of extensor group I afferents increases the extensor phase duration in decerebrate (Whelan *et al.*, 1995) and fictive (Guertin *et al.*, 1995) locomotor preparations. During fictive locomotion of cats a few days after spinalization, progressive extension of the hindlimb, which increases proprioceptive feedback to flexors, increased the rate of flexor phase modulation (Pearson & Rossignol, 1991). The flexor phase duration can be also increased by electrical stimulation of flexor group II afferents during stance (Perreault *et al.*, 1995; Hiebert *et al.*, 1996) and flexor group I afferents during swing (flexor) phase (Perreault *et al.*, 1995; Hiebert *et al.*, 1996; Lam & Pearson, 2001) in decerebrate and fictive cat locomotor preparations.

The functional relevance of the proposed balance of activity between flexor and extensor half-centers may also explain the phase changes observed during upslope and downslope locomotion (Carlson-Kuhta *et al.*, 1998; Smith *et al.*, 1998). For example, during downslope locomotion, progressive increase in slope decreases loading of the hindlimb extensors and requires higher activation of flexors. In this case, the excitatory motion-related sensory feedback to the extensor half-center would be decreased and that to the flexor half-centers would be increased. Our model would then predict that the duration of the extensor (stance) phase would decrease with the increasing downward slope, a result observed in intact cats (Smith *et al.*, 1998). During upslope locomotion, loading of both extensors and flexors is increased. However, a small increase in slope changes extensor loading only slightly, but requires a significant increase in flexor activity most likely to prevent tripping. Thus, we can predict that the duration of swing will increase for locomotion on shallow and moderate upward slopes and decrease for steep slopes, because of the increased extensor load. This pattern of phase modulation is observed experimentally in cats walking up variable slopes (Carlson-Kuhta *et al.*, 1998).

To conclude, in this paper we report a previously unreported flexibility of the locomotor pattern generating circuitry, which can be explained by changes in excitation bias from descending and sensory pathways in the half-center model of the locomotor pattern generator. The proposed mechanism provides a new way of looking at the role of descending inputs and proprioceptive feedback in regulation of stereotypical behaviors.

5.5 References

- ARMSTRONG, D. M. & DREW, T. (1984). Topographical localization in the motor cortex of the cat for somatic afferent responses and evoked movements. J Physiol 350, 33-54.
- ARSHAVSKY, Y. I., DELIAGINA, T. G. & ORLOVSKY, G. N. (1997). Pattern generation. Curr Opin Neurobiol 7, 781-789.
- BELOOZEROVA, I. N. & SIROTA, M. G. (1993). The role of the motor cortex in the control of vigour of locomotor movements in the cat. J Physiol 461, 27-46.
- BONNOT, A., WHELAN, P. J., MENTIS, G. Z. & O'DONOVAN, M. J. (2002). Spatiotemporal pattern of motoneuron activation in the rostral lumbar and the sacral segments during locomotor-like activity in the neonatal mouse spinal cord. J Neurosci 22, RC203.
- BROWN, T. G. (1911). The intrinsic factor in the act of progression in the mammal, vol. 84. Proc. Roy. Soc., London.
- BUFORD, J. A. & SMITH, J. L. (1993). Adaptive control for backward quadrupedal walking. III. Stumbling corrective reactions and cutaneous reflex sensitivity. J Neurophysiol 70, 1102-1114.
- CARLSON-KUHTA, P., TRANK, T. V. & SMITH, J. L. (1998). Forms of forward quadrupedal locomotion. II. A comparison of posture, hindlimb kinematics, and motor patterns for upslope and level walking. J Neurophysiol 79, 1687-1701.
- CHANAUD, C. M., PRATT, C. A. & LOEB, G. E. (1991). Functionally complex muscles of the cat hindlimb.
 V. The roles of histochemical fiber-type regionalization and mechanical heterogeneity in differential muscle activation. Exp Brain Res 85, 300-313.
- CONWAY, B. A., HULTBORN, H. & KIEHN, O. (1987). Proprioceptive input resets central locomotor rhythm in the spinal cat. Exp Brain Res 68, 643-656.
- COWLEY, K. C. & SCHMIDT, B. J. (1997). Regional distribution of the locomotor pattern-generating network in the neonatal rat spinal cord. J Neurophysiol 77, 247-259.
- CYMBALYUK, G. S., GAUDRY, Q., MASINO, M. A. & CALABRESE, R. L. (2002). Bursting in leech heart interneurons: cell-autonomous and network-based mechanisms. J Neurosci 22, 10580-10592.

- DELIAGINA, T. G., ORLOVSKY, G. N. & PAVLOVA, G. A. (1983). The capacity for generation of rhythmic oscillations is distributed in the lumbosacral spinal cord of the cat. Exp Brain Res 53, 81-90.
- DONELAN, J. M., KRAM, R. & KUO, A. D. (2002). Simultaneous positive and negative external mechanical work in human walking. J Biomech 35, 117-124.
- DREW, T. (1988). Motor cortical cell discharge during voluntary gait modification. Brain Res 457, 181-187.
- DUBUC, R., CABELGUEN, J. M. & ROSSIGNOL, S. (1985). Rhythmic antidromic discharges of single primary afferents recorded in cut dorsal root filaments during locomotion in the cat. Brain Res 359, 375-378.
- DUYSENS, J. & PEARSON, K. G. (1980). Inhibition of flexor burst generation by loading ankle extensor muscles in walking cats. Brain Res 187, 321-332.
- ECCLES, R. M. & LUNDBERT, A. (1958). Integrative pattern of Ia synaptic actions on motoneurones of hip and knee muscles. J Physiol (Lond) 144, 271–298.
- FORSSBERG, H. & GRILLNER, S. (1973). The locomotion of the acute spinal cat injected with clonidine i.v. Brain Res 50, 184-186.
- GOSGNACH, S., QUEVEDO, J., FEDIRCHUK, B. & MCCREA, D. A. (2000). Depression of group Ia monosynaptic EPSPs in cat hindlimb motoneurones during fictive locomotion. J Physiol 526, 639-652.
- GOSLOW, G. E., JR., REINKING, R. M. & STUART, D. G. (1973). The cat step cycle: hind limb joint angles and muscle lengths during unrestrained locomotion. J Morphol 141, 1-41.
- GRILLNER, S. (1975). Locomotion in vertebrates: central mechanisms and reflex interaction. Physiol Rev 55, 247-304.
- GRILLNER, S. (1981). Control of locomotion in bipeds, tetrapods, and fish. In Handbook of physiology. Sec2, The nervous system. ed. BROOKHARDT, J. M. & MOUNTCASTLE, V. B., pp. 1179–1236. American Physiological Society, Bethesda.
- GRILLNER, S. & DUBUC, R. (1988). Control of locomotion in vertebrates: spinal and supraspinal mechanisms. Adv Neurol 47, 425-453.
- GRILLNER, S. & ZANGGER, P. (1979). On the central generation of locomotion in the low spinal cat. Exp Brain Res 34, 241-261.

- GUERTIN, P., ANGEL, M. J., PERREAULT, M. C. & MCCREA, D. A. (1995). Ankle extensor group I afferents excite extensors throughout the hindlimb during fictive locomotion in the cat. J Physiol 487 (Pt 1), 197-209.
- HALBERTSMA, J. M. (1983). The stride cycle of the cat: the modelling of locomotion by computerized analysis of automatic recordings. Acta Physiol Scand Suppl 521, 1-75.
- HIEBERT, G. W., WHELAN, P. J., PROCHAZKA, A. & PEARSON, K. G. (1996). Contribution of hind limb flexor muscle afferents to the timing of phase transitions in the cat step cycle. J Neurophysiol 75, 1126-1137.
- Ho, S. & O'DONOVAN, M. J. (1993). Regionalization and intersegmental coordination of rhythmgenerating networks in the spinal cord of the chick embryo. J Neurosci 13, 1354-1371.
- JORDAN, L. M. (1998). Initiation of locomotion in mammals. Ann N Y Acad Sci 860, 83-93.
- KASKE, A., WINBERG, G. & COSTER, J. (2003). Traveling-wave pattern generator controls movement and organization of sensory feedback in a spinal cord model. Biol Cybern 88, 11-19.
- KJAERULFF, O. & KIEHN, O. (1996). Distribution of networks generating and coordinating locomotor activity in the neonatal rat spinal cord in vitro: a lesion study. J Neurosci 16, 5777-5794.
- KOPELL, N. (1988). Toward a theory of modeling central pattern generators. In Neural Control of Rhythmic Movements in Vertebrates, vol. 10. ed. COHEN, A. H., ROSSIGNOL, S. & GRILLNER, S., pp. 369-413. Wiley, New York.
- KREMER, E. & LEV-TOV, A. (1997). Localization of the spinal network associated with generation of hindlimb locomotion in the neonatal rat and organization of its transverse coupling system. J Neurophysiol 77, 1155-1170.
- KRISTAN, W. B., JR. & WEEKS, J. C. (1983). Neurons controlling the initiation, generation and modulation of leech swimming. Symp Soc Exp Biol 37, 243-260.
- LAM, T. & PEARSON, K. G. (2001). Proprioceptive Modulation of Hip Flexor Activity During the Swing Phase of Locomotion in Decerebrate Cats. J Neurophysiol 86, 1321-1332.
- LEBLOND, H. & GOSSARD, J. P. (1997). Supraspinal and segmental signals can be transmitted through separate spinal cord pathways to enhance locomotor activity in extensor muscles in the cat. Exp Brain Res 114, 188-192.
- MARDER, E. & BUCHER, D. (2001). Central pattern generators and the control of rhythmic movements. Curr Biol 11, R986-996.
- METTRIE, D. L., J.O. (1745). L'Homme-Machine. English translation: Man a machine; and Man a plant (1994). Maya Hackett, Indianapolis.

MUYBRIDGE, E. (1887). Animal Locomotion, Philadelphia.

ORLOVSKY, G. N. (1970). Work of reticulo-spinal neurons during locomotion. Biofizika 15, 728-764.

- PANG, M. Y. & YANG, J. F. (2000). The initiation of the swing phase in human infant stepping: importance of hip position and leg loading. J Physiol 528 Pt 2, 389-404.
- PATEL, U. K., GARFINKEL, A. & EDGERTON, V. R. (2003). Toward a model of locomotion pattern generation by traveling waves. In Society for Neuroscience Meeting, vol. 278.11, Orlando.
- PEARSON, K. G. & ROSSIGNOL, S. (1991). Fictive motor patterns in chronic spinal cats. J Neurophysiol 66, 1874-1887.
- PERREAULT, M. C., ANGEL, M. J., GUERTIN, P. & MCCREA, D. A. (1995). Effects of stimulation of hindlimb flexor group II afferents during fictive locomotion in the cat. J Physiol 487 (Pt 1), 211-220.
- PERRET, C. & CABELGUEN, J. M. (1980). Main characteristics of the hindlimb locomotor cycle in the decorticate cat with special reference to bifunctional muscles. Brain Res 187, 333-352.
- PROCHAZKA, A. (1996). Proprioceptive feedback and movement regulation. In Handbook of Physiology. Section 12. Exercise: Regulation and Integration of Multiple Systems. ed. ROWELL, L. & SHEPERD, J. T. American Physiological Society, New York.
- PROCHAZKA, A., TREND, P., HULLIGER, M. & VINCENT, S. (1989). Ensemble proprioceptive activity in the cat step cycle: towards a representative look-up chart. Progress in Brain Research 80, 61-74; discussion 57-60.
- QUEVEDO, J., FEDIRCHUK, B., GOSGNACH, S. & MCCREA, D. A. (2000). Group I disynaptic excitation of cat hindlimb flexor and bifunctional motoneurones during fictive locomotion. J Physiol 525, 549-564.
- RAASTAD, M. & KIEHN, O. (2000). Spike coding during locomotor network activity in ventrally located neurons in the isolated spinal cord from neonatal rat. J Neurophysiol 83, 2825-2834.
- SHERRINGTON, C. S. (1910). Flexion-reflex of the limb, crossed extension-reflex, and reflex stepping and standing. Journal of Physiology (London) 40, 28-121.
- SHIK, M. L. & ORLOVSKY, G. N. (1976). Neurophysiology of locomotor automatism. Physiol Rev 56, 465-501.
- SHIK, M. L., SEVERIN, F. V. & ORLOVSKII, G. N. (1966). [Control of walking and running by means of electric stimulation of the midbrain]. Biofizika 11, 659-666.

- SHIMAMURA, M., KOGURE, I. & WADA, S. (1982). Reticular neuron activities associated with locomotion in thalamic cats. Brain Res 231, 51-62.
- SMITH, J. L., BRADLEY, N. S., CARTER, M. C., GIULIANI, C. A., HOY, M. G., KOSHLAND, G. F. & ZERNICKE, R. F. (1986). Rhythmical movements of the hindlimbs in spinal cat: considerations for a controlling network. In Development and Plasticity of the Mammalian Spinal Cord, vol. 3. ed. GOLDBERGER, M. E., GORIO, A. & MURRAY, M. Liviana Press, Padova.
- SMITH, J. L., CARLSON-KUHTA, P. & TRANK, T. V. (1998). Forms of Forward Quadrupedal Locomotion. III. A Comparison of Posture, Hindlimb Kinematics, and Motor Patterns for Downslope and Level Walking. J Neurophysiol 79, 1702-1716.
- TRANK, T. V., CHEN, C. & SMITH, J. L. (1996). Forms of forward quadrupedal locomotion. I. A comparison of posture, hindlimb kinematics, and motor patterns for normal and crouched walking. J Neurophysiol 76, 2316-2326.
- TRESCH, M. C. & KIEHN, O. (2000). Population reconstruction of the locomotor cycle from interneuron activity in the mammalian spinal cord. J Neurophysiol 83, 1972-1978.
- WHELAN, P. J., HIEBERT, G. W. & PEARSON, K. G. (1995). Stimulation of the group I extensor afferents prolongs the stance phase in walking cats. Exp Brain Res 103, 20-30.
- WHELAN, P. J. & PEARSON, K. G. (1997). Comparison of the effects of stimulating extensor group I afferents on cycle period during walking in conscious and decerebrate cats. Exp Brain Res 117, 444-452.
- YAKOVENKO, S., MUSHAHWAR, V., VANDERHORST, V., HOLSTEGE, G. & PROCHAZKA, A. (2002). Spatiotemporal activation of lumbosacral motoneurons in the locomotor step cycle. J Neurophysiol 87, 1542-1553.

CHAPTER 6

The propriospinal bypothesis of spinal cord regeneration

6.1 Introduction

Many recovery strategies are currently under development to minimize extent of the spinal cord injury and to restore lost functions. Unlike in the peripheral nervous system (PNS), damaged neurons in the central nervous system (CNS) fail to regenerate because of the prohibitive environment attributed to the presence of growth inhibitory molecules associated mostly with proteins in myelin (Schwab & Thoenen, 1985; Caroni & Schwab, 1988) and formation of scar tissue. To overcome this non-permissive environment, Aguayo and colleagues conducted a pioneering study, where long distance axon regeneration was achieved in to a graft of peripheral nerves in the CNS (Aguayo *et al.*, 1981; Bray *et al.*, 1987).

Recently, several groups reported successful long distance regeneration across the spinal cord lesion in completely spinalized rats, which exhibited significant functional improvements, i.e. restoration of weight-bearing and coordinated locomotion (Cheng *et al.*, 1996; Ramon-Cueto *et al.*, 2000; Coumans *et al.*, 2001). These results are particularly crucial because rats do not recover spontaneously (Basso *et al.*, 1996) without extensive locomotor training (Edgerton *et al.*, 2001). Another surprising feature of these results is that significant behavioural improvements were achieved with a limited amount of regeneration. While the circuitry controlling locomotion is located in lumbar spinal segments (Cazalets *et al.*, 1995; Cowley & Schmidt, 1997; Kiehn & Kjaerulff, 1998), the regenerated axons extend only one or two spinal segments below a thoracic lesion (T8-T9), in most cases about 20 mm short of the lumbosacral locomotor region (Ramon-Cueto *et al.*, 1998; Rapalino *et al.*, 1998; Bamber *et al.*, 2001).

What is the mechanism of the functional recovery? One likely explanation is that regenerating axons make connections with propriospinal neurons, which extend caudally and activate hindlimb locomotor circuitry in the lumbar spinal cord (Figure 6.1A). Recently, new evidence in support of this hypothesis has demonstrated that injured axons of the corticospinal tract make spontaneous connections with propriospinal neurons (Bareyre *et al.*, 2004). If this were also the case in the studies with long axon regeneration through a complete spinal cord lesion, non-specific activation of descending propriospinal pathways immediately below the lesion might explain the observed improvements in motor performance of completely spinalized animals (Figure 6.1B). The goal of this study was to test this hypothesis.

6.2 Methods

Animals. The experiments were performed on 9 adult Sprague-Dawley female rats weighing 200-250g. Animal treatment and surgical procedures were approved by the University of Alberta Health Sciences Animal Welfare Committee. They conformed to the guidelines of the Canadian Council on Animal Care.

Spinal cord transection. Animals were deeply anesthetised with isoflurane and underwent an aseptic laminectomy at thoracic T8 spinal level. Complete spinal cord transections 2-3 mm wide were produced with the use of fine suction pipettes at T8 spinal segment until no neural tissue remained and the inside of the dura could be visualized over the entire circumference. The musculature and skin were suture closed, analgesic (Buprenorphine, 5 μ g/kg every 8 h) was administered and animals were placed in a heated environment until full postoperative recovery.



Figure 6.1. Regeneration hypothesis and methods. A. The "propriospinal hypothesis" attributes observed behavioural improvements of the axon regeneration through nerve grafts to nonspecific innervation of propriospinal neurons, which activate lumbosacral locomotor circuitry several segments below the lesion. B. Completely spinalized animals were implanted the ISMS arrays below the lesion to activate intermediate and ventral grey matter. C. Tips of microwires (diameter 25μ m) were deinsulated and cut at a $\sim 30^{\circ}$ angle.

The bladder was emptied manually 3-4 times daily for the next two weeks until reflex voiding was established.

Implantation of electrode array. Approximately 2-3 weeks after complete spinal cord transection, when bladder and bowel functions had recovered, the animals were anesthetised again and underwent a laminectomy at thoracic T9-T12 spinal segments. An array of 6 platinum-iridium microwires (diameter - 25μ m), insulated except for the bevelled tip, arranged in two rows and spaced 2 mm apart was implanted to stimulate intermediate and ventral grey matter one segment below the lesion (Figure 6.1 B). The implantation and stabilization procedures were based on a technique developed for intraspinal microstimulation (ISMS) (Mushahwar *et al.*, 2003). The electrode leads were spot-glued to the dura with cyanoacrylate and covered by a thin piece of plastic film to prevent connective tissue from dislodging the implant. An elastic tube carrying the microwires was secured to the T11 spinous process with dental acrylic. The microwires led percutaneously to a connector, which was secured to the skin with sutures. Post-operative recovery and subsequent bladder and bowel management was as described above.

Behavioural testing. Following full postoperative recovery, we filmed and rated locomotor performance before and during microstimulation using the standard BBB open-field locomotor rating scale (Basso *et al.*, 1995). Animals were rated without intraspinal stimulation for 4 minutes in the first session, which was immediately followed by another full session with intraspinal stimulation. Trains of stimulus pulses (biphasic, 200μ sec, $50s^{-1}$) with amplitudes 0.8-0.9 times threshold for activation of trunk and abdominal muscles ($20-200\mu$ A) were delivered through the ISMS array (Figure 6.1C).
Histology. In order to determine the location of the implanted microwires, the animals were deeply anesthetised with sodium pentobarbital and perfused through the heart with a 3.7% formaldehyde solution. The thoracic segments of the spinal cord with implanted electrodes were extracted and stored in formaldehyde. The spinal cord was subsequently sectioned and the relative positions of electrode tips were established visually.

6.3 Results

To test the "propriospinal hypothesis" nine rats were implanted with the ISMS arrays about one spinal segment below the T8 lesion. Since the length of the electrode array was 4mm, the implant physically occupied just over one thoracic segment (Waibl, 1973). Using thoracic vertebrae as reference, the position of the arrays were confirmed histologically to be within the T9-T11 spinal segments. Figure 6.2A shows location of the electrode tips in the transverse plane relative to the grey matter outline. Generally, the electrodes were positioned in ventral and intermediate grey matter (Fig. 6.2A).

Motor thresholds of each of the implanted electrodes were determined 30 minutes prior to the experimental session. Stimulation elicited small contractions of abdominal and back muscles, whose motoneurons are located in the implanted segments (Nicolopoulos-Stournaras & Iles, 1983). None of the electrodes evoked directly contractions of individual hindlimb muscles or rhythmic leg movements. However, it is possible in some cases that rostral areas of thoracic spinal segment T12 were stimulated due to stimulus spread from the most caudal microwires.

The experimental sessions consisted of two 4 min trials. The first trial without ISMS was followed immediately by the second (treatment) trial with ISMS. The stimulus amplitudes for each



Figure 6.2. Results. A. Location of the implanted electrode tips in thoracic T10-T11 spinal segments. B. Locomotor performance of individual animals before (*open circles*) and during (*filled circles*) ISMS. The mean values for both groups are plotted in red.

electrode were set to 0.8-0.9T with respect to direct muscle activation. ISMS was delivered through all active electrodes simultaneously for the duration of the treatment trial.

Figure 6.2B shows the locomotor performance of individual animals before and during intraspinal stimulation measured with the BBB scale (Basso *et al.*, 1995). A modest, but significant (α =0.05) improvement in the BBB score (mean 1.5±0.6 s.e.m.) was observed during the stimulation. The null hypothesis (treatment had no effect) was rejected (p<0.02). None of the tested animals showed an adequate level of coordination and weight-bearing for walking.

Instead, they used their forelimbs to propel themselves forward before and during ISMS. Generally, most of the animals with ISMS exhibited some hindlimb rhythmicity and coordination between left and right limbs, only explicitly considered in the BBB score. Two animals scored high on the BBB scale before and during the stimulation (>5). These were the only animals with persistent bladder infections. Another feature frequently observed in several animals during the stimulation trials was tonic hip flexion of one leg and hip and knee extension of the contralateral leg. During this state, episodic movements as rated by BBB scale were more likely to be observed on the extended side, while there were rarely any episodic movements on the flexed side.

These results indicate that non-specific electrical activation of the high thoracic spinal segments can improve performance of rats with complete spinal transections assessed by the BBB locomotor rating scale.

6.4 Discussion

Overall, our results support the "propriospinal hypothesis". Though the improvement in motor performance was marginal, it is similar to that reported in recent studies of various

intervention strategies after spinal cord injury (McDonald *et al.*, 1999; GrandPre *et al.*, 2002; Hausmann *et al.*, 2002; Tuszynski *et al.*, 2003; Verdu *et al.*, 2003). In fact, because these studies tested recovery after incomplete spinal cord injuries, the rate of the spontaneous recovery in control animals was higher than that after complete spinal cord transections. Thus, it could be argued that the relative amount of recovery reported in the studies with incomplete transections was actually lower than that reported in this paper.

Propriospinal pathways have been recognized as potential targets of spinal cord repair treatments (Jordan & Schmidt, 2002). Propriospinal neurons are located in ventral as well as in dorsal laminae of the spinal cord and in fact are likely to represent the majority of spinal neurons (Skinner *et al.*, 1979; Chung *et al.*, 1984; Menetrey *et al.*, 1985; Chung *et al.*, 1987). Our choice of intermediate and ventral areas as targets of stimulation was based on evidence that propriospinal neurons in these areas are more involved in the control of locomotion than those in more dorsal areas (Kiehn & Kjaerulff, 1998; Mrowczynski *et al.*, 2001; You *et al.*, 2003). However, this does not rule out the potential importance of dorsal propriospinal pathways, which may also be involved in the activation of the hindlimb locomotor region (Gerasimenko *et al.*, 2003).

Motor improvements observed in graft regeneration studies can be alternatively explained by the possible mechanical coupling between muscles of trunk and hindlimbs (Giszter *et al.*, 1998). For example, strong contractions of abdominal muscles could stretch hip flexors. This may evoke a proprioceptive feedback to the lumbosacral pattern generator and improve the performance of rats on the open-field locomotor scale. In this case, axons regenerating through a graft need only to activate segmental pathways to the local thoracic motoneurons. We cannot dissociate possible contribution of this pathway in this study. However, a recent study of corticospinal tract behaviour

in response to injury supports the "propriospinal hypothesis" by demonstrating that axons start sprouting and form spontaneously functional circuits with short and long propriospinal neurons (Bareyre *et al.*, 2004).

A recent study of the effect of ISMS (Mushahwar *et al.*, 2003) indicates that sensory axons have a lower threshold than cell bodies. This implies that the stimulation in our study is not limited to activation of propriospinal neurons and may also involve activation of segmental afferent pathways. The current experimental approach cannot dissociate the contribution of this pathway to the observed improvement.

The limited amount of functional improvement in our results points to possible limitations of the recovery strategy based only on non-specific activation of thoracic descending systems. In our experiments, none of the animals before or during ISMS developed sufficient weight-bearing or intra- and inter- limb coordination necessary for locomotion. On the other hand, mechanical stimulation applied during bladder expression often produced long-lasting bouts of air-stepping, not observed in the same animal during ISMS. This observation supports the recent finding of pathways from sacrocaudal afferents to the lumbosacral locomotor pattern generator in neonatal rats (Strauss & Lev-Tov, 2003). However, it is possible that the efficiency of input from the thoracic propriospinal system could be increased if combined with other recovery strategies, e.g. locomotor training (Edgerton *et al.*, 2001).

6.5 References

- AGUAYO, A. J., DAVID, S. & BRAY, G. M. (1981). Influences of the glial environment on the elongation of axons after injury: transplantation studies in adult rodents. J Exp Biol 95, 231-240.
- BAMBER, N. I., LI, H., LU, X., OUDEGA, M., AEBISCHER, P. & XU, X. M. (2001). Neurotrophins BDNF and NT-3 promote axonal re-entry into the distal host spinal cord through Schwann cell-seeded mini-channels. Eur J Neurosci 13, 257-268.
- BAREYRE, F. M., KERSCHENSTEINER, M., RAINETEAU, O., METTENLEITER, T. C., WEINMANN, O. & SCHWAB, M.
 E. (2004). The injured spinal cord spontaneously forms a new intraspinal circuit in adult rats. Nat Neurosci 7, 269-277.
- BASSO, D. M., BEATTIE, M. S. & BRESNAHAN, J. C. (1995). A sensitive and reliable locomotor rating scale for open field testing in rats. J Neurotrauma 12, 1-21.
- BASSO, D. M., BEATTIE, M. S. & BRESNAHAN, J. C. (1996). Graded histological and locomotor outcomes after spinal cord contusion using the NYU weight-drop device versus transection. Exp Neurol 139, 244-256.
- BRAY, G. M., VIDAL-SANZ, M. & AGUAYO, A. J. (1987). Regeneration of axons from the central nervous system of adult rats. Prog Brain Res 71, 373-379.
- CARONI, P. & SCHWAB, M. E. (1988). Antibody against myelin-associated inhibitor of neurite growth neutralizes nonpermissive substrate properties of CNS white matter. Neuron 1, 85-96.
- CAZALETS, J. R., BORDE, M. & CLARAC, F. (1995). Localization and organization of the central pattern generator for hindlimb locomotion in newborn rat. J Neurosci 15, 4943-4951.
- CHENG, H., CAO, Y. & OLSON, L. (1996). Spinal cord repair in adult paraplegic rats: partial restoration of hind limb function. Science 273, 510-513.
- CHUNG, K., KEVETTER, G. A., WILLIS, W. D. & COGGESHALL, R. E. (1984). An estimate of the ratio of propriospinal to long tract neurons in the sacral spinal cord of the rat. Neurosci Lett 44, 173-177.
- CHUNG, K., LANGFORD, L. A. & COGGESHALL, R. E. (1987). Primary afferent and propriospinal fibers in the rat dorsal and dorsolateral funiculi. J Comp Neurol 263, 68-75.
- COUMANS, J. V., LIN, T. T., DAI, H. N., MACARTHUR, L., MCATEE, M., NASH, C. & BREGMAN, B. S. (2001). Axonal regeneration and functional recovery after complete spinal cord transection in rats by delayed treatment with transplants and neurotrophins. J Neurosci 21, 9334-9344.

- COWLEY, K. C. & SCHMIDT, B. J. (1997). Regional distribution of the locomotor pattern-generating network in the neonatal rat spinal cord. J Neurophysiol 77, 247-259.
- EDGERTON, V. R., LEON, R. D., HARKEMA, S. J., HODGSON, J. A., LONDON, N., REINKENSMEYER, D. J., ROY, R. R., TALMADGE, R. J., TILLAKARATNE, N. J., TIMOSZYK, W. & TOBIN, A. (2001). Retraining the injured spinal cord. J Physiol 533, 15-22.
- GERASIMENKO, Y. P., AVELEV, V. D., NIKITIN, O. A. & LAVROV, I. A. (2003). Initiation of locomotor activity in spinal cats by epidural stimulation of the spinal cord. Neurosci Behav Physiol 33, 247-254.
- GISZTER, S., GRAZIANI, V., KARGO, W., HOCKENSMITH, G., DAVIES, M. R., SMERASKI, C. S. & MURRAY, M. (1998). Pattern generators and cortical maps in locomotion of spinal injured rats. In Annals of the New York Academy of Sciences, vol. 860, pp. 554-555.
- GRANDPRE, T., LI, S. & STRITTMATTER, S. M. (2002). Nogo-66 receptor antagonist peptide promotes axonal regeneration. Nature 417, 547-551.
- HAUSMANN, O. N., FOUAD, K., WALLIMANN, T. & SCHWAB, M. E. (2002). Protective effects of oral creatine supplementation on spinal cord injury in rats. Spinal Cord 40, 449-456.
- JORDAN, L. M. & SCHMIDT, B. J. (2002). Propriospinal neurons involved in the control of locomotion: potential targets for repair strategies? Prog Brain Res 137, 125-139.
- KIEHN, O. & KJAERULFF, O. (1998). Distribution of central pattern generators for rhythmic motor outputs in the spinal cord of limbed vertebrates. In Annals of the New York Academy of Sciences, vol. 860. ed. KIEHN, O., HARRIS-WARRICK, R. M., JORDAN, L. M., HULTBORN, H. & KUDO, N., pp. 110–129. New York Academy of Sciences, New York.
- MCDONALD, J. W., LIU, X. Z., QU, Y., LIU, S., MICKEY, S. K., TURETSKY, D., GOTTLIEB, D. I. & CHOI, D. W. (1999). Transplanted embryonic stem cells survive, differentiate and promote recovery in injured rat spinal cord. Nat Med 5, 1410-1412.
- MENETREY, D., DE POMMERY, J. & ROUDIER, F. (1985). Propriospinal fibers reaching the lumbar enlargement in the rat. Neurosci Lett 58, 257-261.
- MROWCZYNSKI, W., GROTTEL, K. & KRUTKI, P. (2001). Neurones in the cervical enlargement of the cat spinal cord antidromically activated from sacral segments and the inferior cerebellar peduncle. Neurosci Res 41, 135-141.
- MUSHAHWAR, V. K., PROCHAZKA, A., ELLAWAY, P. H., GUEVREMONT, L. & GAUNT, R. A. (2003). Microstimulation in CNS excites axons before neuronal cell bodies. In Society for Neuroscience Meeting, vol. 276.6, Orlando.

- NICOLOPOULOS-STOURNARAS, S. & ILES, J. F. (1983). Motor neuron columns in the lumbar spinal cord of the rat. J Comp Neurol 217, 75-85.
- RAMON-CUETO, A., CORDERO, M. I., SANTOS-BENITO, F. F. & AVILA, J. (2000). Functional recovery of paraplegic rats and motor axon regeneration in their spinal cords by olfactory ensheathing glia. Neuron 25, 425-435.
- RAMON-CUETO, A., PLANT, G. W., AVILA, J. & BUNGE, M. B. (1998). Long-distance axonal regeneration in the transected adult rat spinal cord is promoted by olfactory ensheathing glia transplants. J Neurosci 18, 3803-3815.
- RAPALINO, O., LAZAROV-SPIEGLER, O., AGRANOV, E., VELAN, G. J., YOLES, E., FRAIDAKIS, M., SOLOMON, A., GEPSTEIN, R., KATZ, A., BELKIN, M., HADANI, M. & SCHWARTZ, M. (1998). Implantation of stimulated homologous macrophages results in partial recovery of paraplegic rats. Nat Med 4, 814-821.
- SCHWAB, M. E. & THOENEN, H. (1985). Dissociated neurons regenerate into sciatic but not optic nerve explants in culture irrespective of neurotrophic factors. J Neurosci 5, 2415-2423.
- SKINNER, R. D., COULTER, J. D., ADAMS, R. J. & REMMEL, R. S. (1979). Cells of origin of long descending propriospinal fibers connecting the spinal enlargements in cat and monkey determined by horseradish peroxidase and electrophysiological techniques. J Comp Neurol 188, 443-454.
- STRAUSS, I. & LEV-TOV, A. (2003). Neural pathways between sacrocaudal afferents and lumbar pattern generators in neonatal rats. J Neurophysiol 89, 773-784.
- TUSZYNSKI, M. H., GRILL, R., JONES, L. L., BRANT, A., BLESCH, A., LOW, K., LACROIX, S. & LU, P. (2003). NT-3 gene delivery elicits growth of chronically injured corticospinal axons and modestly improves functional deficits after chronic scar resection. Exp Neurol 181, 47-56.
- VERDU, E., GARCIA-ALIAS, G., FORES, J., LOPEZ-VALES, R. & NAVARRO, X. (2003). Olfactory ensheathing cells transplanted in lesioned spinal cord prevent loss of spinal cord parenchyma and promote functional recovery. Glia 42, 275-286.
- WAIBL, H. (1973). Zur topographie der medulla spinalis der albinoratte (Rattus norvegicus). Spronger-Verlag, Berlin, New York.
- YOU, S. W., CHEN, B. Y., LIU, H. L., LANG, B., XIA, J. L., JIAO, X. Y. & JU, G. (2003). Spontaneous recovery of locomotion induced by remaining fibers after spinal cord transection in adult rats. Restor Neurol Neurosci 21, 39-45.

CHAPTER 7

General discussion and future directions

7.1 Summary

The control of locomotion is a complex problem of controlling the movement of a mechanical body with multiple degrees of freedom in a variable and often unpredictable environment. This problem is solved in animals by integrated neuro-musculo-skeletal mechanisms for propulsion and stabilization. Chapter 1 of this thesis reviews elements of the locomotor control system and discusses the role of passive dynamic properties of the skeletal system and the contributions of sensory feedback and spinal rhythmogenic neural networks to the control of muscle recruitment and coordination. It also describes challenges and rehabilitation approaches to restoration of motor function after spinal cord trauma. The projects presented in this thesis address questions on integration and contribution of the reviewed systems to locomotion.

The study presented in Chapter 2 addresses a question of whether isometric force-length curves describe muscle during continuous movement. If these static relationships were adequate for the description of muscle behaviour during cyclic changes in muscle length, as in overground locomotion, a region of instability would exist when the muscle is stretched beyond its optimal length. The data from this study showed that isometric force-length curves do not predict muscle behaviour during continuous movement influenced by history-dependent intrinsic muscle properties.

The study of locomotor stability in Chapter 3 takes into account findings in Chapter 2 in the development of an adequate muscle model. This study showed that spring-like muscle properties can be responsible for locomotor stability in the absence of sensory feedback. The data indicated that locomotor stretch reflexes can play a crucial role for weight-bearing when the excitation from higher centers is low. However, the sensory contribution decreases with the

increase in the centrally generated signal and may play a secondary stabilizing role of load redistribution between muscles. These results led to examining the involvement of spring-like muscle properties (Chapter 2) in the regulation of reflex contributions. It was found that muscles could adjust the gain of the sensory feedback loop (loops 2 in Fig. 1.2) during locomotion.

The next system in the hierarchical organization of locomotor control described in Chapter 1 is the central pattern generator (CPG). Its properties were investigated in Chapters 3-6. The study in Chapter 3 used a simple model of the locomotor CPG based on clock-like activation patterns of hindlimb muscles during locomotion. The study in Chapter 4 addressed a question of spatio-temporal activation of hindlimb motoneurons (MNs) controlled by the locomotor CPG during walking. The results support the hypothesis that activation of the lumbosacral MNs during walking is not wave-like, but it is a switch-like oscillation of activity. Because the CPG is thought to span the entire lumbosacral enlargement and consist of interneurons, which may be active in phase with co-localized MNs, these results suggest that activation of the CPG is switch-like as well.

The finding of the switch-like activation of the lumbosacral MNs during walking led to a study of the patterns of MN activity during fictive locomotion, which are not influenced by motion-related sensory feedback controlling transitions between locomotor phases investigated in Chapter 3 (see loop 4 in Fig. 1.2). The analysis of the fictive locomotor patterns in the study described in Chapter 5 revealed that the CPG is capable of adjusting cycle duration by modifying duration of the flexor phase primarily and not the extensor phase. This is opposite of the extensor dominated adjustments during overground locomotion. The data is consistent with the following rule: the variation in the burst duration during spontaneous changes in cycle duration depends on the percent of cycle occupied. These results are explained by a conceptual model of the CPG with two half-centers receiving excitatory bias from descending inputs and proprioceptive feedback

pathways. This model can generalize recent data in the literature on primary and secondary afferent stimulation during normal and fictive locomotion.

Finally, the study in Chapter 6 tests a "propriospinal hypothesis". It is thought that thoracic propriospinal pathways are involved in the improvements of the locomotor performance in animals with complete spinal cord transections bridged by grafts, which allow regeneration of transected descending pathways below the lesion. To test this hypothesis, intraspinal microstimulation (ISMS) was used for non-specific activation of thoracic propriospinal pathways while the locomotor performance was assessed. ISMS produced significant improvements in the locomotor performance. Thus, the results of this study support the propriospinal hypothesis.

7.2 Discussion and future directions

7.2.1 Intrinsic muscle properties

Chapter 2 investigates the dependence of muscle force generation on muscle length during continuous movements. This study was initially designed to test a prediction that isometric angle-torque curves of active wrist extensors would differ from those of flexors (Loren *et al.*, 1996). To address this issue, we measured the angle-torque relationship about the wrist. This relationship would be identical to the muscle force-length curve if angle and torque changes were proportional to muscle displacement and force. It is known that the moment arms of the wrist muscles vary with wrist angle. Using torque-length-angle relationship in cadavers, two groups (Horii *et al.*, 1993; Loren *et al.*, 1996) estimated muscle moment arms from full flexion to full extension of the wrist. Both groups reported that the extensor moment arms decrease when wrist is moved to full flexion. Therefore, when the wrist is flexed, the extensor muscles are supposed to operate on the descending limb of their force-length curve and their moment arm is small. Thus, the extensor

angle-torque curves were expected to exhibit decreasing torque production when the wrist is moved in flexion. However, in our study extensor torque decreased only in isometric trials and it increased with angle during continuous slow $\pm 10^{\circ}$ angular wrist excursions throughout the physiological range of motion (ROM). Thus, in this study we showed that isometric length-tension curves of active wrist muscles do not apply to continuous movements. This result raises a general question of applicability of isometric length-tension curves used in biomechanical studies to describe continuous movement (e.g. Gerritsen *et al.*, 1998).

During small continuous wrist movements, the torque-angle curves exhibited prominent hysteresis. In these trials, the torque increased sharply with stretch and remained high during further lengthening. During shortening the torque declined sharply and remained depressed until lengthening. The observations of sudden changes in muscle stiffness in response to displacement were most consistent with a history-dependent muscle property called short-range stiffness (SRS) (Rack & Westbury, 1974). Because the torque did not return to its isometric level throughout small continuous wrist movements, we suggested that other history-dependent muscle properties, such as force enhancement (Edman *et al.*, 1978; Herzog & Leonard, 2002), force depression (Herzog *et al.*, 2000), yielding and sarcomere length non-uniformities (Abbot & Aubert, 1951; Edman *et al.*, 1978, 1982) may contribute to the force production in our experiment. However, formation of sarcomere length non-uniformities should increase at longer muscle length (Stein *et al.*, 1999). Nevertheless, in our experiment, the torque-angle curves at long muscle lengths were in fact smaller than those at shorter lengths. Therefore, we concluded that the observed hysteresis could be attributed to SRS, force enhancement and depression but not to the sarcomere length nonuniformities. In future studies, contribution of SRS to force generation will be investigated in humans by comparing muscle resistance to stretch before and after initial pre-stretch displacement, which exceeds the range of SRS. It would be intriguing to investigate the contribution of this hysteretic muscle property to control of locomotion and balance in a type of model described in Chapter 3.

7.2.2 Contribution of stretch reflexes during locomotion

This study in Chapter 3 investigated the contribution of stretch reflexes during locomotion. It is known that the spring-like properties of muscles provide automatic load compensation during weight-bearing. Stretch reflexes also contribute to load compensation. So the question is how crucial is the role of sensory feedback during locomotion? To address this question a neuromuscular model of the hindlimbs of a quadruped was used to test the following two hypotheses: 1) the stretch reflexes are too weak and too delayed to contribute significantly to weight-bearing; 2) the important contributions of sensory input involve state-dependent processing. We used the monotonic positive stiffness property of muscles described in Chapter 2 in the muscle models developed for this biomechanical study.

Numerous simulations of gait were performed with varying combinations of muscle activation derived from pre-set ("centrally-generated") patterns and sensory reflexes. In addition, the sensitivity of our conclusions to parametric variations in the model was studied by varying the leg anatomy and the peak forces of the different muscles. Locomotor stability in these wide-ranging simulations was assessed during locomotion in open-loop ("de-afferented") trials and in trials with feedback control based on either sensory-evoked stretch reflexes or finite-state rules.

The results led us to the following conclusions. 1) In the absence of sensory control, the intrinsic stiffness of limb muscles driven by a stereotyped rhythmical pattern can produce

surprisingly stable gait. This corroborates the conclusions of a previous locomotor modeling study (Gerritsen et al., 1998). 2) When the level of central activation of muscles is low, the contribution of stretch reflexes to load compensation can be crucial. However, when the central drive is strong enough to provide adequate load compensation, the contribution of stretch reflexes diminishes. Indeed, the biological controller of locomotion is in general terms a hybrid of feedforward and feedback systems. Animals in which proprioceptive feedback has been abolished can adjust the feedforward command based on visual input, so that their limb kinematics during level walking is practically indistinguishable from normal (Abelew et al., 2000). In addition, regulation of the relative gains of the feedforward and feedback systems may modify an animal's ability to resist perturbations (Kuo, 2002). The relative gains may also be adjusted continuously in intact animals to cope with rough terrain. The regulation of relative gains of the central command and sensory feedback pathways during locomotion remains a question for future studies. 3) Finite-state control can greatly extend the adaptive capability of the locomotor system. It would be intriguing to use the neuro-musculo-skeletal models to investigate the dynamic requirements of coupling between the intrinsic cycling of the locomotor central pattern generator (CPG) and mechanical oscillations due to the interactions of limbs with the ground. The analysis of gait modifications required to preserve stability may single out sensory signals involved in the regulation of phase switching during locomotion.

7.2.3 Patterns of spatiotemporal activation of the lumbosacral motoneurons

At the time of the development of the biomechanical locomotor model, we used a novel technique to analyse the spatiotemporal activity of lumbosacral motoneurons during normal level locomotion. This study is presented in Chapter 4. The aim of the study was to produce a dynamic

model of the spatiotemporal activation of ensembles of alpha motoneurons (MNs) in the cat lumbosacral spinal cord during the locomotor step cycle. The coordinates of MNs of 27 hindlimb muscles of the cat were digitized from transverse sections of spinal cord (Vanderhorst & Holstege, 1997) spanning the entire lumbosacral enlargement from the caudal part of L4 to the rostral part of S1 segments to generate a three dimensional model. We compiled a chart of EMG profiles of the same 27 muscles during the cat step cycle from previous studies. The level of EMG of each muscle was used to scale the activation level of the corresponding MN pool in the digital model. For the full step cycle, activity of the lumbosacral MNs was reconstructed as a sequence of frames that could be played as a continuous movie. A rostrocaudal oscillation of activity in hindlimb MN pools emerged. This was confirmed by computing the locus of the center of activation of the MNs in the consecutive frames of the movie. The caudal third of the lumbosacral enlargement showed intense MN activity during the stance phase of locomotion. During the swing phase, the focus of activation shifted abruptly to the rostral part of the enlargement. At the stance-swing transition, a transient focus of activity formed in the most caudal part of the lumbosacral enlargement. This was associated with activation of gracilis, posterior biceps, posterior semimembranosus, and semitendinosus muscles. These muscles move the foot back and up to clear the ground during locomotion, a role that could be described as retraction. The spatiotemporal distribution of neuronal activity in the spinal cord during normal locomotion with descending control and intact sensory inputs has not been visualized before.

Can dynamics of the lumbosacral MNs give insight into the structure of the pattern generating network? It is well known that MNs receive inputs from multiple pathways, some of which may not be involved in pattern generation. In addition, the pattern generating networks, which could be fairly localized (Cazalets *et al.*, 1995), may use propriospinal projections to activate MNs in different spinal segments. Indeed, it is challenging to dissociate pattern generating elements from those merely following input commands. Even MNs may be considered (Marder, 1991; O'Donovan *et al.*, 1998) an integral part of the pattern generator. Because of the intrinsic dynamic properties such as plateau potentials, MNs respond nonlinearly to the input command and may continue to fire after cessation of stimulation (Conway *et al.*, 1988). An increasing number of studies support the hypothesis that the locomotor CPG is distributed throughout the lumbosacral enlargement (Kiehn & Kjaerulff, 1998). Of course, our MN model does not give any information about the distributions of interneurons, of which the CPG may be largely composed, but it may reflect general activity of the interneuronal populations segmentally co-localized with the MNs (Tresch & Kiehn, 1999).

The finding of rostrocaudal oscillation within the lumbosacral enlargement during locomotion raised a question whether the dynamics of this activity corresponded to a "wave" or it was "switch-like". An experimental study used calcium-imaging to visualize activity of MNs during locomotion in neonatal rats (Bonnot *et al.*, 2002). It showed that activation of several motoneuron pools in L1-L2 and S1-S3 spinal segments supports the "wave" hypothesis. However, it remains to be seen whether this conclusion describes activation of all lumbosacral MN pools. Also, the mode of locomotion has to be established in fictive preparations, i.e. the observed behaviour may correspond to swimming rather than to walking. Two recent modeling studies also supported the "wave" hypothesis (Kaske *et al.*, 2003; Patel *et al.*, 2003). However, unrealistic manipulation of either the somatotopic organization of MNs or their activation profiles was required to fit the conclusion. For example, Kaske et al. assumed that the knee extensor MNs were located caudally in the lumbosacral enlargement, when Patel et al. showed co-activation of ankle flexors with other extensors throughout the stance phase of quadrupedal locomotion. The results of our study show

clear spatiotemoral separation of activity during swing and stance. This pattern is consistent with the "switch-like" activation of the hindlimb MNs during walking.

From the differences between the spatiotemporal patterns of various stereotypic movements of intact and reduced preparations, information on the locomotor pattern generating circuitry may be deduced. Future studies will focus on the classification of the MN activity during stereotypical movements in intact and reduced preparations. The analysis of the spatiotemporal activity of the hindlimb MNs during fictive locomotion elicited by electrical stimulation of the mesencephalic locomotor region (MLR) is currently in progress (Yakovenko *et al.*, 2001).

7.2.4 Adjustments of the locomotor phases with changes in cycle duration

Chapter 5 includes a study of variability in the locomotor patterns during fictive locomotion, when motion-related feedback is absent. In this study, we discovered a novel flexibility of the pattern generating circuitry. Previous studies (Goslow *et al.*, 1973; Halbertsma, 1983) have shown that during intact cat locomotion the duration of the extensor phase (Te) varies more than the flexor phase (Tf) with changes in cycle period (Tc). The cycle duration is also dominated by the extensor phase (see Chapter 4). We now report that this relationship can be reversed in MLR-induced fictive locomotion when flexor phase duration dominates the step cycle. Cyclical bursts of nerve activity during fictive locomotion were analyzed using a threshold crossing method to identify onsets and cessations of activity. Regression analysis of bursting of individual nerves showed that Te and Tf increased linearly with Tc. Similarly to the intact cat, when the extensor phase was longer than the flexor phase, Te line of best fit had a steeper slope, than Tf. However, when the locomotor cycles were dominated by the flexor activity, the slope was steeper for flexors than for extensors.

Based on this finding, we proposed that the pattern generated by the locomotor CPG depends on the distribution of excitation between flexor and extensor half-centers, which are analogous and antagonistic. In this model, descending and sensory pathways can increase the level of excitation of each half-center. A shift in the excitatory bias to one of the half-centers may lead to an increase in the duration of the corresponding phase of the cycle and a decrease in the antagonistic phase through reciprocal neural interactions (Cymbalyuk *et al.*, 2002).

The proposed model can describe a number of reported observations of phase adjustments in response to unloading (Duysens & Pearson, 1980; Pang & Yang, 2000), electrical stimulation of group I afferents of extensors (Guertin *et al.*, 1995; Whelan *et al.*, 1995) and group I & II afferents of flexors (Perreault *et al.*, 1995; Hiebert *et al.*, 1996) and progressive extension of the hindlimb during fictive locomotion of cats a few days after spinalization (Pearson & Rossignol, 1991). The functional relevance of the proposed balance of activity between flexor and extensor half-centers may also explain the phase changes observed during upslope and downslope locomotion (Carlson-Kuhta *et al.*, 1998; Smith *et al.*, 1998) and air-stepping (Smith *et al.*, 1986).

Overall, the observed behavior is consistent with a theory that the CPG, controlling locomotion of each limb, consists of at least two antagonistic analogous oscillators, each of which is capable of dominating the activity in the step cycle (Brown, 1911).

7.2.5 The "propriospinal hypothesis"

Finally, Chapter 6 investigates a hypothesis, which may explain improvements of motor functions observed in spinalized rats after long distance axon regeneration. Many laboratories have reported the successful regeneration of neurons across damaged portions of the spinal cord (Cheng *et al.*, 1996; Ramon-Cueto *et al.*, 2000; Coumans *et al.*, 2001). Functional connections have been

inferred from improvements in locomotor movements. But regenerated neurons generally extend only about 10mm beyond the lesion site, about 20mm short of the lumbosacral locomotor region. The motor improvements have been explained by possible excitation of propriospinal neurons, which in turn relay this excitation to the hindlimb CPG. However, the observed motor improvements in graft regeneration studies can be alternatively explained by the possible mechanical coupling between muscles of the trunk and hindlimbs (Giszter *et al.*, 1998). For example, sensory excitation may be generated by strong contractions of abdominal muscles, which in turn stretch hip flexors. In this case, axons regenerating through a graft need only to activate segmental pathways to the local thoracic motoneurons. If trunk muscles were activated at a low level insufficient for stretching hip flexors, the contribution of this pathway would be diminished.

We tested the "propriospinal hypothesis" by using intraspinal microstimulation (ISMS) of the thoracic spinal cord to activate descending propriospinal neurons. The level of the ISMS was adjusted to reduce possible contribution of the mechanical coupling between trunk and hindlimb segments. We rated locomotion before, during and immediately after ISMS using a standard open field locomotor rating scale, the BBB scale. Stimulation significantly increased the BBB scores and increased rhythmicity of hindlimb movements. We conclude that electrically evoked activity in thoracic spinal segments isolated from supraspinal control may propagate caudally and activate lumbar pattern generating networks. Thus, this study supports the propriospinal mediation hypothesis of regeneration.

Future directions of this study will address questions about pathways activated by ISMS of thoracic spinal cord. A recent study of the effect of ISMS (Mushahwar *et al.*, 2003) indicates that sensory axons have a lower threshold than cell bodies. This implies that ISMS in our study is not limited to activation of propriospinal neurons and may also involve activation of segmental afferent

pathways. In addition, ISMS may activate sympathetic and parasympathetic preganglionic efferent pathways, which may affect activity of the spinal cord below the lesion. The current experimental approach cannot dissociate the contribution of these pathways to the observed improvement. Another future direction of this study may address a question of whether or not the efficiency of input evoked by ISMS of the thoracic propriospinal system could be increased if combined with other recovery strategies, e.g. locomotor training (Edgerton *et al.*, 2001).

7.2.6 Synthesis and impact of the presented studies

The studies completed during my Ph.D. program address a number of questions in the field of Motor Control. The common goal of these studies was to investigate integration of skeletal, muscular and neural systems for the control of locomotion. Some of the main results presented above have led to research in other laboratories around the world. The movies generated as part of the work on spatiotemporal activity of lumbosacral motoneurons (Chapter 4) are now used for neuroscience teaching in universities in North America and Europe.

The studies in my thesis have opened up the following questions for further research. What is the contribution of muscle history-dependent properties, e.g. short-range stiffness, to modulation of sensory feedback and locomotor stability? How is the contribution of locomotor stretch reflexes modulated during different modes of locomotion, e.g. walking vs. running? Is the finite control system described in Chapter 3 modified to stabilize walking over uneven terrain? Does the lumbar enlargement contain long propriospinal connections coordinating activity of rostral hip flexors to caudal hip extensors? What are the dynamics of the spatiotemporal activity during different types of locomotion? Can two coupled generic oscillators describe the behaviour of the CPG model with excitatory bias presented in Chapter 5? Does ISMS of the thoracic sensory pathways improve motor performance in animals with complete spinal cord transection? Studies presented in Chapters 4 & 5 have raised question about the workings of the CPG. This has led to the initiation of at least two studies in other laboratories interested in fictive and decerebrate locomotion.

In conclusion, this thesis describes how the problem of locomotor control is solved by a hierarchically organized, neuro-musculo-skeletal system with dynamic interactions between its parts.

7.3 References

- ABBOT, B. C. & AUBERT, X. M. (1951). The force exerted by active striated muscle during and after change in length. J Physiol 117, 77-86.
- ABELEW, T. A., MILLER, M. D., COPE, T. C. & NICHOLS, T. R. (2000). Local loss of proprioception results in disruption of interjoint coordination during locomotion in the cat. J Neurophysiol 84, 2709-2714.
- BONNOT, A., WHELAN, P. J., MENTIS, G. Z. & O'DONOVAN, M. J. (2002). Spatiotemporal pattern of motoneuron activation in the rostral lumbar and the sacral segments during locomotor-like activity in the neonatal mouse spinal cord. J Neurosci 22, RC203.
- BROWN, T. G. (1911). The intrinsic factor in the act of progression in the mammal, vol. 84. Proc. Roy. Soc., London.
- CARLSON-KUHTA, P., TRANK, T. V. & SMITH, J. L. (1998). Forms of forward quadrupedal locomotion. II. A comparison of posture, hindlimb kinematics, and motor patterns for upslope and level walking. J Neurophysiol 79, 1687-1701.
- CAZALETS, J. R., BORDE, M. & CLARAC, F. (1995). Localization and organization of the central pattern generator for hindlimb locomotion in newborn rat. J Neurosci 15, 4943-4951.
- CHENG, H., CAO, Y. & OLSON, L. (1996). Spinal cord repair in adult paraplegic rats: partial restoration of hind limb function. Science 273, 510-513.
- CONWAY, B. A., HULTBORN, H., KIEHN, O. & MINTZ, I. (1988). Plateau potentials in alphamotoneurones induced by intravenous injection of L-dopa and clonidine in the spinal cat. J Physiol 405, 369-384.

- COUMANS, J. V., LIN, T. T., DAI, H. N., MACARTHUR, L., MCATEE, M., NASH, C. & BREGMAN, B. S. (2001). Axonal regeneration and functional recovery after complete spinal cord transection in rats by delayed treatment with transplants and neurotrophins. J Neurosci 21, 9334-9344.
- CYMBALYUK, G. S., GAUDRY, Q., MASINO, M. A. & CALABRESE, R. L. (2002). Bursting in leech heart interneurons: cell-autonomous and network-based mechanisms. J Neurosci 22, 10580-10592.
- DUYSENS, J. & PEARSON, K. G. (1980). Inhibition of flexor burst generation by loading ankle extensor muscles in walking cats. Brain Res 187, 321-332.
- EDGERTON, V. R., LEON, R. D., HARKEMA, S. J., HODGSON, J. A., LONDON, N., REINKENSMEYER, D. J., ROY, R. R., TALMADGE, R. J., TILLAKARATNE, N. J., TIMOSZYK, W. & TOBIN, A. (2001). Retraining the injured spinal cord. J Physiol 533, 15-22.
- EDMAN, K. A., ELZINGA, G. & NOBLE, M. I. (1978). Enhancement of mechanical performance by stretch during tetanic contractions of vertebrate skeletal muscle fibres. J Physiol 281, 139-155.
- EDMAN, K. A., ELZINGA, G. & NOBLE, M. I. (1982). Residual force enhancement after stretch of contracting frog single muscle fibers. J Gen Physiol 80, 769-784.
- GERRITSEN, K. G., VAN DEN BOGERT, A. J., HULLIGER, M. & ZERNICKE, R. F. (1998). Intrinsic muscle properties facilitate locomotor control - a computer simulation study. Motor Control 2, 206-220.
- GISZTER, S., GRAZIANI, V., KARGO, W., HOCKENSMITH, G., DAVIES, M. R., SMERASKI, C. S. & MURRAY, M. (1998). Pattern generators and cortical maps in locomotion of spinal injured rats. In Annals of the New York Academy of Sciences, vol. 860, pp. 554-555.
- GOSLOW, G. E., REINKING, R. M. & STUART, D. G. (1973). The cat step cycle: Hind limb joint angles and muscle lengths during unrestrained locomotion. J. Morph. 141, 1-42.
- GUERTIN, P., ANGEL, M. J., PERREAULT, M. C. & MCCREA, D. A. (1995). Ankle extensor group I afferents excite extensors throughout the hindlimb during fictive locomotion in the cat. J Physiol 487 (Pt 1), 197-209.
- HALBERTSMA, J. M. (1983). The stride cycle of the cat: the modelling of locomotion by computerized analysis of automatic recordings. Acta Physiologica Scandinavica. Supplementum 521, 1-75.
- HERZOG, W. & LEONARD, T. R. (2002). Force enhancement following stretching of skeletal muscle: a new mechanism. J Exp Biol 205, 1275-1283.
- HERZOG, W., LEONARD, T. R. & WU, J. Z. (2000). The relationship between force depression following shortening and mechanical work in skeletal muscle. J Biomech 33, 659-668.

- HIEBERT, G. W., WHELAN, P. J., PROCHAZKA, A. & PEARSON, K. G. (1996). Contribution of hind limb flexor muscle afferents to the timing of phase transitions in the cat step cycle. J Neurophysiol 75, 1126-1137.
- HORII, E., AN, K. N. & LINSCHEID, R. L. (1993). Excursion of prime wrist tendons. J Hand Surg [Am] 18, 83-90.
- KASKE, A., WINBERG, G. & COSTER, J. (2003). Traveling-wave pattern generator controls movement and organization of sensory feedback in a spinal cord model. Biol Cybern 88, 11-19.
- KIEHN, O. & KJAERULFF, O. (1998). Distribution of central pattern generators for rhythmic motor outputs in the spinal cord of limbed vertebrates. In Annals of the New York Academy of Sciences, vol. 860. ed. KIEHN, O., HARRIS-WARRICK, R. M., JORDAN, L. M., HULTBORN, H. & KUDO, N., pp. 110–129. New York Academy of Sciences, New York.
- KUO, A. D. (2002). The relative roles of feedforward and feedback in the control of rhythmic movements. Motor Control 6, 129-145.
- LOREN, G. J., SHOEMAKER, S. D., BURKHOLDER, T. J., JACOBSON, M. D., FRIDEN, J. & LIEBER, R. L. (1996). Human wrist motors: biomechanical design and application to tendon transfers. J Biomech 29, 331-342.
- MARDER, E. (1991). Modifiability of pattern generation. Curr Opin Neurobiol 1, 571-576.
- MUSHAHWAR, V. K., PROCHAZKA, A., ELLAWAY, P. H., GUEVREMONT, L. & GAUNT, R. A. (2003). Microstimulation in CNS excites axons before neuronal cell bodies. In Society for Neuroscience Meeting, vol. 276.6, Orlando.
- O'DONOVAN, M. J., WENNER, P., CHUB, N., TABAK, J. & RINZEL, J. (1998). Mechanisms of spontaneous activity in the developing spinal cord and their relevance to locomotion. Ann N Y Acad Sci 860, 130-141.
- PANG, M. Y. & YANG, J. F. (2000). The initiation of the swing phase in human infant stepping: importance of hip position and leg loading. J Physiol 528 Pt 2, 389-404.
- PATEL, U. K., GARFINKEL, A. & EDGERTON, V. R. (2003). Toward a model of locomotion pattern generation by traveling waves. In Society for Neuroscience Meeting, vol. 278.11, Orlando.
- PEARSON, K. G. & ROSSIGNOL, S. (1991). Fictive motor patterns in chronic spinal cats. J Neurophysiol 66, 1874-1887.
- PERREAULT, M. C., ANGEL, M. J., GUERTIN, P. & MCCREA, D. A. (1995). Effects of stimulation of hindlimb flexor group II afferents during fictive locomotion in the cat. J Physiol 487 (Pt 1), 211-220.

- RACK, P. M. & WESTBURY, D. R. (1974). The short range stiffness of active mammalian muscle and its effect on mechanical properties. J Physiol 240, 331-350.
- RAMON-CUETO, A., CORDERO, M. I., SANTOS-BENITO, F. F. & AVILA, J. (2000). Functional recovery of paraplegic rats and motor axon regeneration in their spinal cords by olfactory ensheathing glia. Neuron 25, 425-435.
- SMITH, J. L., BRADLEY, N. S., CARTER, M. C., GIULIANI, C. A., HOY, M. G., KOSHLAND, G. F. & ZERNICKE, R. F. (1986). Rhythmical movements of the hindlimbs in spinal cat: considerations for a controlling network. In Development and Plasticity of the Mammalian Spinal Cord, vol. 3. ed. GOLDBERGER, M. E., GORIO, A. & MURRAY, M. Liviana Press, Padova.
- SMITH, J. L., CARLSON-KUHTA, P. & TRANK, T. V. (1998). Forms of Forward Quadrupedal Locomotion. III. A Comparison of Posture, Hindlimb Kinematics, and Motor Patterns for Downslope and Level Walking. J Neurophysiol 79, 1702-1716.
- STEIN, R. B., MOMOSE, K. & BOBET, J. (1999). Biomechanics of human quadriceps muscles during electrical stimulation. J Biomech 32, 347-357.
- TRESCH, M. C. & KIEHN, O. (1999). Coding of locomotor phase in populations of neurons in rostral and caudal segments of the neonatal rat lumbar spinal cord. J Neurophysiol 82, 3563-3574.
- VANDERHORST, V. G. & HOLSTEGE, G. (1997). Organization of lumbosacral motoneuronal cell groups innervating hindlimb, pelvic floor, and axial muscles in the cat. J Comp Neurol 382, 46-76.
- WHELAN, P. J., HIEBERT, G. W. & PEARSON, K. G. (1995). Stimulation of the group I extensor afferents prolongs the stance phase in walking cats. Exp Brain Res 103, 20-30.
- YAKOVENKO, S., KATINKA, S., MCCREA, D. A. & PROCHAZKA, A. (2001). Spatiotemporal activation of cat lumbosacral motoneurons during MLR-induced fictive locomotion in the cat. In Society for Neuroscience Abstracts, pp. 625.624, San Diego, USA.